POLITICAL INFORMATION CYCLES: WHEN DO VOTERS SANCTION INCUMBENT PARTIES FOR HIGH HOMICIDE RATES?*

John Marshall[†]

April 2017

Do voters sanction incumbent parties for their performance in office? I argue that how governments are held to account depends upon when voters consume information about the relevant incumbent party. If news consumption follows electoral cycles, short-term performance indicators in the news prior to elections may powerfully shape voting behavior. In the context of local homicides and Mexican municipal elections, I test this theory's central implications using three distinct sources of plausibly exogenous variation. First, I show that voters indeed consume more news before local elections, and that homicides before such elections increase the salience of public security and reduce confidence in the mayor. Second, electoral returns confirm that preelection homicide shocks substantially decrease the incumbent party's vote share and re-election probability. However, such sanctioning is limited to mayoral elections, and is barely impacted by longer-term homicide rates. Finally, the punishment of homicide shocks relies on, and increases with, access to local broadcast media stations. These effects are only pronounced among less-informed voters, who principally engage with politics around elections. The findings demonstrate the importance of when voters consume news, and contribute to explaining the mixed electoral accountability often observed outside consolidated democracies and in federal systems.

^{*}I thank Jim Alt, Eric Arias, Abhijit Banerjee, Allyson Benton, David Broockman, Bruce Bueno de Mesquita, Melani Cammett, Francisco Cantú, Ana De La O, Jorge Domínguez, Oeindrila Dube, Raissa Fabregas, Nilesh Fernando, Hernán Flom, Alex Fouirnaies, Jeff Frieden, Thomas Fujiwara, Saad Gulzar, Sebastian Garrido de Sierra, Andy Hall, Torben Iversen, Joy Langston, Horacio Larreguy, Chappell Lawson, Sandra Ley, Rakeen Mabud, Noah Nathan, Brian Phillips, Otis Reed, James Robinson, Rob Schub, Gilles Serra, Jim Snyder, Chiara Superti, Edoardo Teso, Giancarlo Visconti, Julie Weaver, and workshop or conference participants at presentations at APSA, Berkeley, Caltech, Columbia, Harvard, LSE, MIT, MPSA, NYU, NEWEPS, Stanford, Stanford GSB, and Yale for illuminating discussions and useful comments. I thank Horacio Larreguy for kindly sharing municipal election data. Under its previous title, "Information Acquisition, Local Media, and Electoral Accountability: When Do Mexican Voters Punish Incumbents for High Homicide Rates?", the paper was the co-winner of the Kellogg/Notre Dame Award for the Best Paper in Comparative Politics at MPSA 2015.

[†]Department of Political Science, Columbia University. Email: jm4401@columbia.edu.

1 Introduction

Electoral accountability rests upon informed voters electing competent representatives. Given the difficulty of directly observing candidate competence, performance on salient issues like the economy, public security, or corruption represent key signals of an incumbent party's continuing suitability for office (Fearon 1999; Ferejohn 1986). However, particularly in developing contexts, voters are often poorly informed about politics and current affairs (e.g. Keefer 2007; Pande 2011), or do not receive relevant information pertaining to their incumbent party (Snyder and Strömberg 2010).¹ This lack of politically-relevant information may explain why voters often do not hold incumbents to account on available performance metrics, or struggle to assign responsibility across layers of government.² While many scholars have argued that access to media facilitates electoral accountability (see Ashworth 2012; Pande 2011),³ the availability of media coverage does not imply that voters actually *consume* politically-relevant news.

I instead argue that how voters hold incumbent parties accountable for their performance in office reflects the *timing* of their news consumption. In particular, if voters primarily consume news before elections—when they are more likely to actively seek out politically-relevant information (Hamilton 2004; Marshall 2016), and media outlets are more likely to supply such information (Prior 2007)—electoral sanctioning will reflect salient indicators of incumbent performance reported in the media at this time. Consequently, incumbent performance indicators only impact election outcomes if they are covered in the media when voters consume news before elections. I expect the effect of such *political information cycles* on vote choice to be most pronounced among poorly-informed voters, whose weak prior beliefs over the incumbent's continuing suitability for office are most responsive to available performance indicators (Zaller 1992).

This paper provides evidence for the central elements of this theory in the context of examining how Mexican voters hold different levels of government to account for local violent crime.

¹Ill-informed electorates pose similar dilemmas in established democracies (Bartels 2008; Delli Carpini and Keeter 1996), but the risks of electing incompetent politicians are likely greater under weak democratic institutions.

²Evidence that voters, especially in consolidating democracies, punish or reward politicians for economic performance (Remmer 1991; Roberts and Wibbels 1999; Singer and Carlin 2013), levels of public security (Kronick 2014; Vivanco et al. 2015), or malfeasance in office (Arias et al. 2017; Chong et al. 2015; de Figueiredo, Hidalgo and Kasahara 2013; Ferraz and Finan 2008; Larreguy, Marshall and Snyder 2017) is mixed. The developed country literature similarly highlights mixed or conditional effects (see Anderson 2007; Golden 2010; Healy and Malhotra 2013).

³Well-identified studies show that news or advertising content affect voting behavior (e.g. Enikolopov, Petrova and Zhuravskaya 2011; DellaVigna and Kaplan 2007; Gerber et al. 2011; Larreguy, Marshall and Snyder 2016; Snyder and Strömberg 2010), while others emphasize that such information must have a credible source (Alt, Lassen and Marshall 2016; Chiang and Knight 2011; Lenz and Ladd 2009) and may require a minimum level of sophistication to process complex information (Alt, Lassen and Marshall 2016; Gomez and Wilson 2006). Experimental studies primarily focus on providing information just before elections, and rarely mimic typical newscasts.

Following significant democratizing and decentralizing reforms in the 1990s, Mexican elections have become relatively competitive at the national and sub-national levels and responsibility for administering key public services such as policing is shared across municipal, state, and federal governments. Public security is a major concern among Mexican voters, as in many other Latin American nations, and thus local homicides represent a salient incumbent performance metric. However, since voters are largely poorly informed about politics and a significant portion of the electorate lacks strong partisan ties (Chong et al. 2015; Greene 2011; Lawson and McCann 2005; McCann and Lawson 2003), there is scope for pre-election crime reports to influence voting behavior. Although incumbents cannot seek re-election, voters hold the incumbent's party responsible in Mexico's party-centric system (Chong et al. 2015; Larreguy, Marshall and Snyder 2017).

Leveraging plausibly exogenous variation in temporal proximity to local elections, municipallevel homicide shocks just before elections, and access to local broadcast media stations, I test the political information cycles argument at the individual and electoral precinct levels. First, to identify the effect of upcoming local elections on news consumption and voter beliefs, I exploit the irregular timing of survey waves and staggered electoral cycles across Mexican states to isolate variation in the proximity of municipal and state legislative elections (see Eifert, Miguel and Posner 2010). Rather than separate supply and demand explanations for information consumption, I focus on consumption cycles *in equilibrium*.⁴ I first show that voters indeed consume more political news through television and radio just before local elections, and also demonstrate greater political knowledge. While better informed voters increase the intensity of their news consumption, less informed voters effectively only consume political news during elections campaigns.

Second, using electoral returns between 1999 and 2013, I show that pre-election homicide shocks substantially harm the municipal incumbent party's electoral performance. To identify the effects of homicide spikes before elections—when voters consume most news—I exploit idiosyncratic month-to-month volatility in municipal homicide counts. Specifically, I compare "shocked" municipal elections that experienced more homicides in the two months preceding the election to "control" elections from within the same municipality that experienced more homicides in the two months after the election.⁵ Such homicide shocks are consistent with random sample variability and are well balanced over 102 covariates, while I find no evidence that drug trafficking organizations (DTOs) or politicians manipulate homicide rates around elections. Despite being uncorrelated with broader homicide levels and trends, and thus representing weak indicators of long-term performance, pre-election homicide shocks reduce the incumbent party's vote share by

⁴Marshall (2016) seeks to tease apart explanations for increased consumption.

⁵I include only municipalities experiencing at least one homicide over the two months before and after an election.

2.2 percentage points and its probability of winning by 12 percentage points.

Consistent with poorly-informed voters relying on the pre-election signals available when they actually consume news, voters do not hold incumbents accountable for longer-run homicide rates. Using a difference-in-differences design, I find no evidence that homicide rates over the prior year or electoral cycle—arguably more desirable indicators of incumbent performance (Healy and Lenz 2014), but which may not be observed by voters predominantly consuming information just before elections—affect incumbent electoral performance. Furthermore, my analysis of concurrent state and federal elections, and comparisons between municipalities with and without their own police force, indicate that the parties of incumbent mayors, rather than higher levels of government, are held to account for local homicide shocks (see also Vivanco et al. 2015). This finding chimes with research in the U.S. also highlighting that mayoral electoral performance is negatively correlated with local homicides (Arnold and Carnes 2012), but which does not illuminate the role of information cycles.

Third, I link the individual and aggregate-level findings by showing that *local* broadcast media play a crucial role in the transmission of these effects. Local broadcasters report politically-relevant news that voters may not be able to access otherwise, and local crime receives substantial media coverage (e.g. Trelles and Carreras 2012). Using detailed media coverage maps for every radio and television station in the country, I exploit within-neighboring precinct variation in coverage to identify the effects of broadcast media. I find that, on average, each additional local media station based within a precinct's own municipality reduces the incumbent's vote share by 0.22 percentage points, and is driven principally by television. However, the 19% of precincts covered by least local media stations do not significantly punish homicide shocks, while an additional non-local station does not affect voter sanctioning.

Turning to the mechanisms, I find that such behavior reflect voters updating in a Bayesian fashion from a weak signal of incumbent performance. First, indicating that voters internalize news in the media just before elections, pre-election homicides both increase concern about public security and reduce confidence in the municipal incumbent by around 10 percentage points. In contrast with a "recency bias" explanation (see Healy and Lenz 2014; Zaller 1992), homicide shocks occurring before surveys outside of electoral campaigns—when voters consume least news—do not influence voter beliefs in the short term. Second, illustrating voters' capacity to integrate new information with their prior beliefs, sanctioning of homicide shocks is greatest where a different incumbent party at the previous election did not oversee a pre-election homicide shock, or the same incumbent party experiences consecutive shocks. In contrast with Banerjee et al. (2011) and Kayser and Peress (2012), benchmarking is primarily temporal rather than spatial in this context. Third, changes in beliefs and electoral sanctioning are concentrated among less educated voters, who consume less news outside election campaigns. This suggests that voters with weaker prior beliefs about the incumbent party's performance on violent crime update most from pre-election homicide shocks. Fourth, I show that although media coverage of violent crime closely tracks the occurrence of homicides, coverage does not increase before elections. In conjunction with the fact that some voters only consume news before elections, this suggests that demand for politically-relevant news may be the key driver of political information cycles. However, further data is required to dismiss the supply-side explanation that changes in the tone of coverage also explain voter behavior.

My finding that political information cycles induce voters to heavily weight recent performance indicators in ways that can substantially alter electoral outcomes makes several theoretical and empirical contributions. First, it shows that, in addition to voters requiring access to performance indicators,⁶ such indicators may only alter voting behavior when their coverage in the media co-incides with voters actually consuming news. Political information cycles may thus explain the mixed evidence of electoral accountability across different performance metrics and release dates (e.g. Achen and Bartels 2004*b*; Brollo 2009; Chang, Golden and Hill 2010; Roberts and Wibbels 1999). My findings similarly rationalize studies showing that politicians release bad news when voters are distracted by other events (e.g. Durante and Zhuravskaya forthcoming; Eisensee and Strömberg 2007).

Second, in contrast with theories resting on myopic voters (e.g. Downs 1957; Nordhaus 1975), limited memory (e.g. Zaller 1992), or short-lived priming (e.g. Gerber et al. 2011), my argument implies that the common finding that voters respond positively to economic performance just before elections could instead reflect a poorly-informed electorate updating—potentially with overconfidence (Ortoleva and Snowberg 2015)—from the limited information they possess at the time. In this respect, my results micro-found and extend Healy and Lenz's (2014) finding from U.S. survey experiments that voters use recent indicators of economic performance when lacking the longer-term indicators they would prefer. However, the sophisticated updating and differential punishment across levels of government exhibited by Mexican voters suggests that such behavior cannot be accounted for by voters blindly responding to events beyond the incumbent's control (see Achen and Bartels 2004*a*; Healy, Malhotra and Mo 2010).

Third, I contribute evidence that voters attempt to hold governments accountable for public security. Alongside the economy, public security represents the main valence issue in many developing contexts, and particularly in Latin America.⁷ By illustrating how the timing of news

⁶See e.g. Banerjee et al. (2011), Casey (2015), Chang, Golden and Hill (2010), Ferraz and Finan (2008), Healy and Lenz (2014), Larreguy, Marshall and Snyder (2017), and Snyder and Strömberg (2010).

⁷While existing studies have suggested that high levels of violence can affect incumbent electoral performance

consumption affects sanctioning, my results help explain why Mexican scholars have struggled to verify the widely-held belief that voters respond to local crime (e.g. Vivanco et al. 2015).

Fourth, in a complex federal system where shared control of policing reduces "clarity of responsibility" (Powell and Whitten 1993), I show that Mexican voters are able to assign responsibility across layers of government, at least in their own minds, for local homicides. In contrast with previous studies suggesting that voters hold national-level politicians accountable for major policy outcomes (e.g. Rodden and Wibbels 2011), or interpret crime as a threat against voting for parties opposing organized crime (Alesina, Piccolo and Pinotti 2016), I find that voters perceive municipal rather than state or federal incumbents to be responsible for local violent crime.

The paper proceeds as follows. Section 2 analyzes how political information cycles affect electoral accountability. Section 3 describes the Mexican context used to test my argument. The empirical analysis then identifies the three central components of my theory: sections 4-6 respectively examine information consumption and voter beliefs, incumbent electoral performance, and the moderating role of media coverage. Section 7 provides evidence of the mechanisms, before section 8 concludes by discussing the normative and policy implications.

2 Theoretical argument

Electoral accountability is underpinned by retrospective (Ferejohn 1986) or prospective voting (Fearon 1999) models premised on the assumption that voters re-elect incumbent parties based on relevant publicly available information. In light of the mixed evidence that governments are held accountable for their performance in office (see e.g. Anderson 2007; Ashworth 2012; Healy and Malhotra 2013; Pande 2011), scholars have pointed to the importance of voter *access* to politically-relevant information in the media (e.g. Chang, Golden and Hill 2010; Larreguy, Marshall and Snyder 2017; Snyder and Strömberg 2010). However, such studies assume that voters actually *consume* the available information, and by focusing on performance metrics released before elections fail to capture how the timing of politically-relevant news may differentially impact voters.

I instead argue that the impact of news on voting behavior depends upon the timing of voter information consumption. In particular, I propose that news consumption follows a cycle where poorly-informed voters only seriously engage with politically-relevant news just before elections,

⁽e.g. Berrebi and Klor 2006; Cummins 2009; Kronick 2014; Ley 2014), incumbent evaluations (García-Ponce, Wantchekon and Zeitzoff 2014), or increase informed participation among victims (e.g. Bateson 2012; Bellows and Miguel 2009), this study explains why responses to violence primarily relate to recent shocks, rather than standard longer-term measures. Furthermore, by isolating plausibly exogenous variation in homicide shocks, the timing of news consumption, and access to local media, the results advance a literature that has predominantly correlated indicators of violence with electoral outcomes.

and thus heavily weight the incumbent performance indicators in the news at this time when casting their ballot.

2.1 Political information cycles

There are various reasons to believe that consumption of politically-relevant news increases before elections. First, to the extent that politics is a pure consumption good (e.g. Hamilton 2004), electoral campaigns are likely to particularly interest voters. While coverage of political events out of election season is often more focused on specific policy issues, campaigns are specifically designed to appeal to voters. Second, Marshall (2016) shows that voters in social networks that collectively value knowledge about politics may strategically acquire more information to cultivate a reputation as politically sophisticated, especially around elections when political discussion is more common (see also Baker, Ames and Renno 2006). Third, voters may feel a civic duty to become informed around elections (Feddersen and Sandroni 2006). Fourth, voters without instrumental or intrinsic motivations may consume more information around elections simply because more air time is devoted to political news, advertising, and specific election programming. Even for uninterested voters, such information becomes difficult to avoid, especially where there are relatively few channels available (Prior 2007) or such information is recast as entertainment (Baum 2002), and performance indicators are increasingly placed in the context of appraising incumbent politicians and parties (Semetko and Valkenburg 2000).

Regardless of whether increased consumption is driven principally by voter demand for information or political and media supply of information, consumption of politically-relevant news is likely to reflect a political information cycle with a clear spike prior to elections. Such increased consumption is likely to cause many voters—especially in developing contexts where baseline political knowledge is relatively low (Pande 2011)—to only engage with the news before elections. Relative politically engaged voters are likely to to ratchet up their consumption. Since the timing of news consumption is the foundation of my argument, it is essential to first establish the existence of such political information cycles:

H1. Voters start to consume, or consume relatively more, news just before elections.

Akin to voters learning about candidate positions over an electoral campaign (e.g. Alvarez 2001; Druckman 2005; Hirano et al. 2015; Lenz 2009, 2012), increased information consumption is likely to translate into greater voter awareness of current affairs including salient valence issues.

2.2 Implications for electoral accountability

Political information cycles may impact voters' political beliefs by emphasizing politically-relevant information in the news when consumption is greatest. Events in the news may prime the salience of valence issues, or induce voters to learn about the performance of incumbent politicians (Lenz 2012) or their response to or preparation for the event (Ashworth and Bueno de Mesquita 2014*a*). For example, if there are many homicides in the news just before an election, voters may become more concerned about public security and negatively update their beliefs about the incumbent party's ability to address or respond to this issue. This requires that voters believe the signal to be informative about the party's competence. These issue priming and learning mechanisms imply:

H2a. Voters care more about issues that are in the news just before elections.

H2b. Negative information about incumbent performance on salient issues just before elections reduces confidence in the incumbent party's competence for office.

Especially at local elections, voters are poorly informed about incumbent performance in office.⁸ There is thus good reason to believe that voters may substantially update their beliefs in line with the news that they consume before elections. With weak prior beliefs, voters may still update substantially when recent news is a weak signal of performance over the full electoral cycle, especially if they are overconfident in the signal's precision.

For such beliefs to translate into voting behavior, the issues in the news before elections must be sufficiently important to voters (e.g. Krosnick and Kinder 1990).⁹ The model in the Appendix, which derives the hypotheses enumerated in this section, shows that the electoral impact of increasing the salience of an issue can be ambiguous, depending upon posterior beliefs about the relative competence of incumbent and challenger parties on that issue.¹⁰ However, if the news substantially updates voters' posterior beliefs about the suitability of a candidate for office—as the literature examining voter learning about candidate positions suggests (Hirano et al. 2015; Lenz 2009)—the learning effect dominates and implies that:

H3. Negative information about incumbent performance on salient issues just before elections reduces the incumbent party's vote share.

⁸Even in the U.S., Healy and Lenz (2014) provide evidence that voters use recent economic performance over the last year to proxy for cumulative performance over a Presidential term.

⁹Rather than policy issues, voters may rely heavily on partisanship, ethnicity or social ties (e.g. Brader and Tucker 2012; Casey 2015; Chandra 2007).

¹⁰The model shows that bad news about the incumbent on a given issue induces voters to negatively update about the party's suitability for office, and that this negative learning effect is compounded by increasing the issue's salience if the incumbent already scored badly on the issue. If the incumbent scored well, the negative learning effect dominates the opposing priming effect when the increase in salience is small or news departs substantially from a voter's prior.

The voting implications of news consumption also depend upon how voters attribute responsibility for performance across different levels of government. Particularly in federal systems where powers are shared or assigned according to non-transparent rules, disentangling such responsibility can be challenging (Anderson 2006; Powell and Whitten 1993). Although the federal government is often the primary target of sanctions (e.g. Rodden and Wibbels 2011), voters may also rely on visible local indicators of responsibility such as the frequency with which they engage with local as opposed state or federal police forces. Changes in voting behavior induced by recent news are then most pronounced when voters can clearly assign responsibility:

H4. Negative information about incumbent performance on salient issues just before elections only reduces the vote share of incumbent parties deemed responsible for such performance.

Exactly how voters *perceive* responsibility across layers of government on different issues is ultimately an empirical question.

However, the electoral impact of news reports before elections depends upon the availability and content of the news. Voters rely upon media sources such as television and radio to supply credible and *politically-relevant* news—news that pertains to a voter's incumbent party. Locallybased media are more likely to report information about relevant local incumbents (e.g. Snyder and Strömberg 2010). Furthermore, given that media sources may vary in the extent to which they distort or under-report certain types of news (e.g. Besley and Prat 2006; Gentzkow and Shapiro 2006; Mullainathan and Shleifer 2005), voters may have a preference for variety in their media consumption (Larreguy, Marshall and Snyder 2017), and market segmentation may mean that not all stations reach all types of voters (Barabas and Jerit 2009; Prat and Strömberg 2005), the ultimate impact of information about incumbent performance on voting behavior should increase with the number of local media stations as the likelihood that voters consume relevant news increases. I thus hypothesize that:

H5. Negative information about incumbent performance on salient issues just before elections only reduces the incumbent party's vote share if voters can access relevant information via the media, and is increasing in the level of such exposure.

Moreover, events in the news before elections differentially impact different types of voters. Voters with weak priors, who may be consuming politically-relevant information for the first time in months or years, are most susceptible to news because their beliefs are most malleable (e.g. Hirano et al. 2015; Lawson 2004; Zaller 1992). Conversely, voters consuming news throughout the electoral cycle possess stronger prior beliefs over the issues they regard as important and how well different parties address such issues. They are also well-placed to distinguish informative

from uninformative signals. These arguments are demonstrated formally in the Appendix, and receive empirical support from Da Silveira and De Mello (2011) and Larreguy, Marshall and Snyder (2016), who find that relatively uneducated voters in Brazil and Mexico respectively respond most to political advertising. For incumbent performance indicators, I therefore hypothesize that:¹¹

H6. The consumption of negative information about incumbent performance on salient issues just before elections impacts most the beliefs and voting behavior of voters with weak priors.

The empirical analysis will use education as a proxy for the strength of voter priors.

Nevertheless, even if voters primarily consume political information before elections, this does not entail that they entirely lack prior beliefs. Incumbent performance information available around the previous election may represent an important component of priors beliefs that voters can benchmark current indicators of incumbent party performance against. For example, if voters received a positive signal of incumbent performance on a given issue relating to a *previous* incumbent party's competence before the last election, a negative information in the new before the current election may cause voters to negatively update about the current incumbent party. Similarly, consecutive negative or positive signals about the same incumbent party may strengthen posterior beliefs. Conversely, voters should update less when two different parties experience the same signal, or the same party experiences different signals across time; these cases point to common shocks or high signal variance, and thus imply that voters should downweight the signal.

H7. The impact of negative information about incumbent performance on the incumbent party's vote share is greater where voters previously received signals that opposition parties performed well or where such information compounds previous negative information about the incumbent party.

To the extent that voters observe events in neighboring municipalities and believe that their governments would behave similarly in their own municipality, a similar logic may hold for spatial comparisons.

I now test these hypotheses in the context of local violent crime reported in the media—a prevalent performance indicator on a highly salient issue—before Mexican municipal elections.

¹¹The results follow most straight-forwardly from common priors. A sufficiently strong correlation between the position of a voter's prior distribution and education could reverse this result.

3 Violent crime and political accountability in Mexico

Mexico was an archetypal competitive authoritarian regime until the late 1990s, but has since democratized significantly.¹² Three main political parties—the right-wing National Action Party (PAN), left-wing Party of the Democratic Revolution (PRD), and previously-hegemonic Institutional Revolutionary Party (PRI)—have competed for power in recent decades.¹³ Despite growing partisan alignment, often induced by clientelistic exchanges, many Mexicans have yet to develop strong ties to specific political parties (Greene 2011; Lawson and McCann 2005).

Mexico's federal system is divided between three administrative and elected layers of government: approximately 2,500 municipalities, 31 states (excluding the Federal District of Mexico City), and the federal government. Constitutional reforms in the mid-1990s substantially increased mayoral autonomy over the provision of local public services (see Diaz-Cayeros, González and Rojas 2006), inducing municipal spending to rise to around 20% of total government spending.¹⁴ The median municipality contains 13,000 people, although large cities with the exception of the Federal District—which is divided into delegations—represent singular municipal bodies.

Municipal mayors are typically elected every three years to non-renewable terms,¹⁵ and enter office between three and seven months after election day. Municipal elections are almost always held in tandem with state legislative elections, although gubernatorial elections are often held separately. I refer to simultaneous municipal and state legislative elections as "local elections." Such local elections have become increasingly competitive, with fewer than 50% of municipal incumbent parties gaining re-election. Congressional elections to the House and Senate are held every three years, while the President is elected every six years. The majority of local elections do not coincide with federal elections.

Although incumbent mayors could not seek re-election, Mexico's party-centric system ensures that voters are likely to hold incumbent parties accountable for the actions of individual politicians. First, differences in candidate selection mechanisms across parties at the state level ensure that candidate choices differ across party but are highly correlated within parties (Langston 2003). Second,

¹²The PRI had retained a stranglehold on power since 1929 by implementing populist policies, establishing strong clientelistic ties, effectively mobilizing voters, creating barriers to political entry, and manipulating electoral outcomes (e.g. Cornelius 1996; Greene 2007; Magaloni 2006).

¹³After attaining legislative pluralities in the 1990s, the long-time conservative opposition PAN fully broke PRI control when Vicente Fox won the Presidency in 2000. In 2006, Felipe Calderón retained the Presidency for the PAN, narrowly defeating the PRD's Andrés Manuel López Obrador. After regaining legislative majorities, but never relinquishing regional control (Langston 2003), the PRI candidate Enrique Peña Nieto won the Presidency in 2012.

¹⁴This expansion of spending included policing, although most municipalities already had their own police forces. Presidents Fox and Calderón also increased federal transfers to municipalities for policing in the 2000s (Sabet 2010).

¹⁵From 2018, re-election will become possible for legislators, and mayors in most states.

voters are substantially better informed about parties than individual politicians (Chong et al. 2015; Larreguy, Marshall and Snyder 2016); several examples are provided below. Ultimately, various studies show that voters frequently punish parties for an incumbent's performance in office (Arias et al. 2017; Chong et al. 2015; Larreguy, Marshall and Snyder 2017).

3.1 Public security forces

Responsibility for public security is shared across levels of government, and like other federal systems such as the United States, mayors play an important role in fighting crime. Both state and federal laws can be used to prosecute criminals, and uniformed (preventive) and investigative police forces exist at both the federal level and in each of Mexico's 31 states and the Federal District of Mexico City. State and federal police, and increasingly the army, are responsible for investigating major crimes in different jurisdictions. State police investigate state crimes such as homicides, while federal officers focus on organized crime (Reames 2003). However, around three-quarters of municipalities also possess their own police force. Although such forces also support higher-level operations and supply information, their principal role is preventive: they primarily patrol the streets and maintain public order, address administrative issues, and respond first to criminal incidents (Reames 2003; Sabet 2010). Municipal police are by far most numerous, accounting for more than half of Mexica enforcement personnel (Sabet 2010). A large central police force controlled by the mayor of Mexico City covers all delegations within the Federal District.

In municipalities with their own police force, mayors can influence the medium-term incidence of crime. Mayors choose the local police chief and set local policies. PAN mayors have played a key role in supporting Calderón's crackdown on Mexico's DTOs (Dell 2015), while political alignment across neighboring municipalities has reduced rates of violent crime (Durante and Gutierrez 2015). The widely-publicized case of Iguala represents a particularly egregious example of political control of the local police, where the mayor and his wife exploited their links with law enforcement to cover up and possibly instigate the murder of 43 student protesters in 2014. However, as shown below, mayors are not able to manipulate short-term crime rates, at least with respect to homicides.

Given the number of municipal police offers and their "on-the-streets" presence, it is not surprising that municipal police are the foremost police force in the minds of voters. When asked which law enforcement authorities they can identify, the 2010 National Survey About Insecurity (ENSI) finds that while 75% of voters could identify municipal police, only 38% and 48% respectively recognized state and federal forces. Focus groups that I conducted in the field suggest that voters primarily blame local government for crime in their community.



Figure 1: Crime in Mexico

Notes: The data in Figure 1b are from the 2001-2011 Latinobarómetro surveys. Crime includes concerns about crime/public security, drug trafficking, violence, or terrorism/political violence.

3.2 Trends in violence

According to the United Nations Office on Drugs and Crime (UNODC), Mexico suffers one of the world's highest homicide rates. In 2012, 21.5 people per 100,000 were intentionally murdered (UNODC 2013). This represents the 20th highest homicide rate in the world—slightly less than South Africa, Colombia and Brazil, and slightly more than Nigeria, Botswana and Panama.

Using data from the National Institute of Statistics, Geography, and Information (INEGI), Mexico's autonomous statistical agency, Figure 1a plots the number of homicides per month occurring in the average municipality between 1999 and 2013. INEGI defines an intentional homicide as an unnatural death, as determined by a coroner's report.¹⁶ Month and location of death are also based on the coroner's report. The monthly homicide rate has been substantial throughout this period, but increased dramatically after President Calderón entered office in December 2006 and began Mexico's "War on Drugs" (see Dell 2015). Although these figures may understate the true homicide

¹⁶Although there is no official measure of drug-related homicides enshrined in Mexican law, the federal government has sporadically released monthly data. This data suffers from various problems beyond its short time-series (see Heinle, Rodríguez Ferreira and Shirk 2014), while Dell's (2015) estimates show that total homicides measured by INEGI rose substantially more than drug-related homicides following Calderón's war on drugs. Furthermore, voter fears about public security are not only linked to organized crime, but also reflect equally-prevalent non-drug related homicides (of which only a tiny fraction occur within families). Table A11 shows broadly similar results for the small sample (December 2006-October 2011) when homicides were classified as drug-related by a committee composed of representatives from the ministries constituting the National Council of Public Security.

rate if unnatural causes cannot be detected (México Evalúa 2012), it is unlikely that the coroners reports are falsified because aggregated data is only released publicly several years after the event.¹⁷ Encouragingly, 94% of homicides were registered in the same month that they occurred. The homicide clearance rate is only 20% (México Evalúa 2012), and drug-related homicides—which are regionally concentrated—represent 50% of homicides over this period (Heinle, Rodríguez Ferreira and Shirk 2014). However, many municipalities only rarely experience a homicide; in fact, only one homicide occurs in the median municipality each year.

Unsurprisingly, voters are concerned about Mexico's high rates of violent crime. Like many other Latin American countries (Blanco 2013), Figure 1b shows that the number of Mexican Latinobarómetro respondents citing public security as the most important problem facing the country increased broadly in line with homicide rates. For most of the 2000s, public security registered as the most salient issue for voters ahead of the economy.

3.3 Media coverage and voter political knowledge

Like most developing countries, broadcast media is voters' main source of news. According to the 2009 Latinobarometer, 83% of Mexicans receive political information from television, 41% from radio, 30% from newspapers, and 41% from family, friends and colleagues. In 2008, around 10% listened to radio every day, while the average person watched 4 hours of television (Ibope/AGB México 2009). In contrast, only 21% have access to the internet at home, while 3G coverage only started expanding in 2011.¹⁸

The nature of media coverage thus plays a central role in determining whether voters voters receive politically-relevant information. Mexico contains 852 AM radio stations, 1,097 FM stations and 1,255 television stations, which generally cover relatively small geographic areas (see Figure 3 below). Most stations form part of broader regional and national radio or television networks such as Grupo ACIR, Radiorama, Televisa, and TV Azteca—where affiliates share branding or are owned-and-operated subsidiaries (Larreguy, Marshall and Snyder 2017). Among radios, a 2016 survey shows that 83% of radios report news more than once a day, and typically report on municipal rather than national issues. For television stations, identical entertainment content is generally bought from or relayed by network providers. However, while national news is typically centrally provided, affiliates and regional subdivisions emitting from major cities within each state also provide significant local news content. Of the 52 distinct television channels (excluding Mexico City's 24-hour news channels) for which schedules were available in 2015, the average channel broad-

¹⁷Heinle, Rodríguez Ferreira and Shirk (2014) discuss Mexican homicide metrics in depth.

¹⁸In 2010 and 2011, 60% of Latinobarómetro respondents reported that they never use the internet.

casts 3.6 hours of news coverage each weekday (both before and after the June elections). Slightly less than half of this news is devoted to state or city-specific programming. Private radio and television outlets rely on advertising revenues to support themselves, and thus face strong incentives to tailor their programming to local audiences (Larreguy, Marshall and Snyder 2017). The pro-PRI coverage biases that characterized elections before the late 1990s have somewhat dissipated (e.g. Hallin 2000; Lawson 2002).

Homicides are not always reported in great depth, but are regularly covered in print (Ley 2014; Osorio 2015) and "omnipresent" in the local broadcast media (Trelles and Carreras 2012).¹⁹ In a survey of all local radio stations in 2016, 96% report regularly covering local security stories, while Table 9 below demonstrates that newspaper reporting of violent crime indeed tracks the homicide rate. Based on the 2010 ENSI survey, 87% of respondents report learning about public security in the country and in their state from television news programs, while 34%, 29%, and 9% respectively report learning from radio news programs, periodicals or newspapers, and the internet.²⁰ Furthermore, 82% of voters believe that television news has the most important influence on public opinion.

Despite claiming reasonable levels of attention to political news, knowledge of public affairs is limited. I show below that only half of Mexicans can answer basic questions about politics. Although 80% of voters can correctly identify their municipality's incumbent party (Arias et al. 2017), Castañeda Sabido (2011) and Chong et al. (2015) similarly find that voters are generally unaware of mayoral responsibilities and performance in office. This lack of knowledge suggests that many voters may possess weak prior beliefs about their incumbent party's suitability for office, and may thus be particularly likely to internalize information when they actually follow the news.

4 The existence of political information cycles

The starting block of my argument is that voters consume substantially more political information just prior to elections. While some voters only start consuming politically-relevant information during election campaigns, others will consume more than before. This section identifies the effect of upcoming local elections, providing survey-level evidence for the existence of such political information cycles.

¹⁹For example, see this local news report of an gangland-style murder in Monterrey. The same news station also reports on domestic murders, e.g. this case of a woman strangling her partner. News programs may continue to report on arrests and cases that go to trial, especially when the defendants are found guilty. For example, on 17th February 2015, W Radio reported here that two men were convicted of intentional homicide and sentenced to 27 years.

²⁰Although social networks may represent a key source of information on more explicitly political issues (Baker, Ames and Renno 2006), comparatively few learn from work colleagues (5%) or family, friends and neighbors (13%).

4.1 Data

To examine political information cycles, I use four cross-sectional waves of the National Survey of Political Culture and Civil Practices (ENCUP) conducted over several weeks in November 2001, February 2003, December 2005, and August 2012.²¹ Each round draws stratified random samples of around 4,500 Mexican voters for face-to-face interviews from pre-selected electoral precincts within urban and rural strata defined by the electoral register.²² The survey was commissioned by the Interior Ministry and designed to be broadly nationally representative, and focuses on the country's political culture rather than more contentious questions about elections.²³ Each survey wave was conducted in a different month of the year, and the irregularly-spaced waves do not correspond with the federal electoral cycle. The pooled sample thus includes up to 17,213 respondents across 539 municipalities (including delegations within the Federal District).

I measure political news consumption in two ways. First, I use the frequency with which voters watch or listen to the news, programs about politics, or programs about public affairs.²⁴ To understand the margins at which consumption is changing, I examine various consumption intensities: never, at some point, at least monthly, at least weekly, and daily. I also compute a 5-point scale (from 0 to 4). Second, I assess whether politically-relevant news consumption translates into greater political knowledge. I focus on topical political knowledge that voters are more likely to encounter through the media (Barabas et al. 2014). Topical political knowledge is measured as the first (standardized) factor from a set of indicators coding correct responses to simple factual questions regarding recent political news.²⁵ A advantage of this measure is that voters cannot falsely inflate their knowledge; the average respondent answered around half the questions correctly. Because these consumption measures cannot distinguish changes in demand for information among voters from changes in the supply of political news provided by media outlets, this paper emphasizes *equilibrium* political information consumption.

²¹The 2008 wave does not provide a respondent's municipality and asks different questions about media consumption. The ENCUP surveys are preferred to the Latinobarómetro, which covers more years, but does not always ask about media consumption and especially political information, and does not contain municipal identifiers (only states). Nevertheless, the Latinobarómetro returns similar results. My identification strategy is not compatible with the rich Mexican Panel Surveys that have been conducted around the 2000, 2006 and 2012 Presidential elections.

²²In 2012, 5 broad strata were identified, and electoral precincts and then voters were randomly selected from within such strata to match the strata's rural-urban, gender, and age distribution. In 2005 and 2012, 10 voters were surveyed from each precinct according to specific directions (see the 2012 methodological manual here). Although such detailed sampling information is not available for the earlier surveys, the overall design is similar.

²³The study was implemented by INEGI in 2001 and 2003, and private firms in 2005 and 2012. The specific objectives of the study, which does not address elections at all, are enumerated here.

²⁴I focus on radio and television, which are by far the most prevalent sources of political information in Mexico.

²⁵In 2001, 2003, 2005 and 2012 respectively, respondents were asked 6, 3, 3 and 4 questions (see Appendix). Questions regarding basic (national and local) knowledge of political institutions were excluded.

4.2 Identification strategy

Using a similar design to Eifert, Miguel and Posner (2010), I exploit the timing of survey administration with respect to state-specific election cycles to identify the effects of an upcoming election. Operationally, I code an indicator for respondents facing an upcoming municipal election, and typically a simultaneous state legislative election, within five months of the survey. Although there is some state-level discretion over campaigning in such local elections, campaigns generally last around five months.²⁶ As noted above, the ENCUP surveys are administered irregularly—both in terms of the month in which the survey was conducted and the number of years between survey rounds. In conjunction with the fact that states historically follow different electoral cycles, varying both in the month and year of their elections (including over time within states, especially due to a recent constitutional reform),²⁷ whether the surveys were administered just before or after elections or whether an election was held recently at all effectively occurs by chance.

There is no evidence to suggest that these surveys, which do not explicitly address elections, were strategically timed. This is reinforced by the fact that 27 of Mexico's 32 states (including the Federal District) register an election in one of the survey years, of which 14 only hold an election in a survey year but after the survey occurred. Supporting the plausibility of these arguments, Table A2 shows that individuals surveyed prior to local elections are well-balanced over individual and municipal level characteristics: only 2 of 18 tests report a significant difference at the 10% level. Table A5 shows that neither changes in upcoming local elections nor violence predict changes in whether a municipality is included in a given survey round.

To identify the effect of an upcoming local election, I first estimate the following regression for respondents *i* in municipality *m* at survey year *t*:

$$Y_{imt} = \beta U pcoming \ local \ election_{mt} + \mu_t + \varepsilon_{imt}, \tag{1}$$

where Y_{imt} is a measure of politically-relevant news consumption. Survey fixed effects, μ_t , capture common period effects that might arise from concurrent federal elections (in 2003 and 2012), Presidential elections (in 2012), or national trends in political behavior. As a robustness check,

²⁶Political advertising slots are specifically allocated for these purposes five months prior to federal elections (Larreguy, Marshall and Snyder 2016). The results are robust to defining upcoming elections by any number of months between 1 and 12.

²⁷See Table A1 in the Appendix for a full list of municipal elections by month. In Chiapas, Coahuila, Guerrero, Michoacán, Quintana Roo, Veracruz, and Yucatán, the typical 3-year cycle was adjusted over the sample period by switching to a 2 or 4-year term for a single electoral cycle. Moreover, following a constitutional amendment in 2007, states were subsequently mandated to hold local elections on the same day as federal elections when the state cycle coincides (Serra 2014). Consequently, states also changed the month of their elections. To reduce constant electoral competition, some states holding off-cycle elections also homogenized elections after the reform.

I control flexibly for the only—but potentially important—imbalance on respondent education. Throughout, standard errors are clustered by municipality.

Although upcoming local elections are plausibly exogenous to the timing of the survey, I also implement a difference-in-differences design by including state fixed effects, λ_s . This design exploits only variation in the presence of state-level upcoming elections within states across surveys, and thus relies on the parallel trends assumption that states facing and not facing upcoming elections at the time of the survey follow similar trends in political news consumption. While this assumption is especially plausible in a context where all states repeatedly hold elections, this design entails an important efficiency cost. Because half the states in the dataset never experience an election within 5 months in advance of an election, there is no variation in the treatment for half the sample. In addition, the estimates only pertain to the 14 states experiencing at least one upcoming local election over the four survey waves.

4.3 Upcoming elections increase news consumption and political knowledge

The results in panel A-C of Table 1 demonstrate that local elections increases both the likelihood that a voter consumes any political information at all as well as the level of information consumed.²⁸ Column (1) of panel A shows that an upcoming local election significantly increases the probability that a voter listens or watches news at all by five percentage points. Although this shift could still reflect an increase in the supply of political news on television and the radio, only consuming news before elections is likely to represent a conscious acquisition choice by voters that had previously avoided the relatively extensive news coverage available outside election campaigns. Column (1) in panel B also reports a large increase in weekly news and political program consumption. The effect on the five-point scale in column (1) of panel C is also strongly positive. These findings suggest that virtually all types of voters, ranging from the least to most engaged, consume more political information prior to statewide elections.

Panel D shows that the cycle of increased information consumption prior to elections mirrors changes in political knowledge. Contrary to potential concerns about social desirability bias, column (1) demonstrates that voters facing an upcoming local election are more than a quarter of a standard deviation, or nine percentage points, more likely to correctly answer a question testing their political knowledge.²⁹ This increase indicates that voters actively engage by internalizing

²⁸Comparable questions from 2001 were not available.

²⁹Like Hirano et al. (2015), this result shows that voters learn more about politics during electoral campaigns, even around lower-level elections. Unreported results interacting upcoming local elections with baseline covariates for both consumption and knowledge outcomes indicates that the same voters consuming more news also become more knowledgeable about politics.

				Measure of	political news	consumption			
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)
Panel A: Outcome: Watch	h and listen to	news and pol	litical program	ns ever					
Upcoming local election	0.049***	0.033*	0.006						
	(0.014)	(0.017)	(0.018)						
Upcoming local election				0.093***	0.079***	0.044*			
(two-month)				(0.015)	(0.015)	(0.023)			
Months until next election							-0.002***	-0.001*	-0.002***
							(0.001)	(0.001)	(0.001)
Observations	13 030	8 330	13 030	13 030	8 330	13 030	13 030	8 330	13 030
Outcome range	{0.1}	{0.1}	{0,1}	{0,1}	{01}	{0,1}	{0.1}	{01}	{0,1}
Outcome mean	0.87	0.86	0.87	0.87	0.86	0.87	0.87	0.86	0.87
Treatment mean	0.19	0.24	0.19	0.02	0.03	0.02	17.53	18.88	17.53
Panel B: Outcome: Watch	h and listen to	news and pol	itical program	ns weekly					
Upcoming local election	0.082***	0.053**	0.058*	·					
1 0	(0.022)	(0.024)	(0.030)						
Upcoming local election	. ,			0.116***	0.095***	0.028			
(two-month)				(0.038)	(0.030)	(0.045)			
Months until next election							-0.005***	-0.005***	-0.004***
							(0.001)	(0.001)	(0.001)
	12.020	0.220	12.020	12.020	0.220	12.020	12.020	0.220	12.020
Observations	13,030	8,550	13,030	13,030	8,330	13,030	13,030	8,330	13,030
Outcome range	$\{0,1\}$	$\{0,1\}$	$\{0,1\}$	$\{0,1\}$	$\{0,1\}$	$\{0,1\}$	$\{0,1\}$	$\{0,1\}$	$\{0,1\}$
Tractor and mean	0.03	0.03	0.63	0.03	0.03	0.03	0.05	0.03	0.05
	0.19	0.24	0.19	0.02	0.05	0.02	17.55	10.00	17.55
Panel C: Outcome: Watch	h and listen to	o news and pol	litical program	ns scale					
Upcoming local election	0.276***	0.174**	0.158						
	(0.075)	(0.079)	(0.098)						
Upcoming local election				0.416***	0.343***	0.132			
(two-month)				(0.122)	(0.093)	(0.145)	0.01.04.44	0.01.1444	0.010
Months until next election							-0.016***	-0.014***	-0.012***
							(0.002)	(0.003)	(0.003)
Observations	13,030	8,330	13,030	13,030	8,330	13,030	13,030	8,330	13,030
Outcome range	$\{0,1,2,3,4\}$	$\{0,1,2,3,4\}$	$\{0,1,2,3,4\}$	$\{0,1,2,3,4\}$	$\{0,1,2,3,4\}$	{0,1,2,3,4}	$\{0,1,2,3,4\}$	$\{0,1,2,3,4\}$	{0,1,2,3,4}
Outcome mean	2.58	2.57	2.58	2.58	2.57	2.58	2.58	2.57	2.58
Treatment mean	0.19	0.24	0.19	0.02	0.03	0.02	17.53	18.88	17.53
Panel D: Outcome: Topic	al political kn	owledge							
Upcoming local election	0.312***	0.152***	0.119***						
	(0.051)	(0.036)	(0.044)						
Upcoming local election				0.218**	0.188**	0.002			
(two-month)				(0.102)	(0.080)	(0.080)			
Months until next election							-0.006***	-0.003**	-0.002
							(0.002)	(0.001)	(0.001)
Observations	17 213	12 513	17 213	17 213	12 513	17 213	17 213	12 513	17 213
Outcome range	[-2,]1 1 56]	[-2, 11, 1, 56]	[-2, 11, 1, 56]	[-2,]1 1 56]	[-2,11,1,56]	[-2, 11, 1, 56]	[-2,11,1,56]	[-2, 11, 1, 56]	[-2, 11, 1, 56]
Outcome mean	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00
Outcome std_dev	1.00	1.00	1.00	1.00	1.00	1.00	1.00	1.00	1.00
Treatment mean	0.16	0.18	0.16	0.03	0.04	0.03	18.58	19.88	18.58
Education for -1 -fft		/			/			/	
State fixed effects		V	.(~	.(~	./
State fixed cifetts			v			v			v

Table 1: The effect of upcoming local elections on political news consumption and topical political knowledge

Notes: All specifications include survey-year fixed effects, and are estimated using OLS. The outcomes in panels A-C were not collected in the 2001 survey, and education was not collected in 2005. Standard errors clustered by municipality are in parentheses. * denotes p < 0.1, ** denotes p < 0.05, *** denotes p < 0.01.

political information—a prerequisite for political information to influence electoral accountability.

Column (2)-(9) demonstrate the robustness of these findings at both the extensive and intensive margins of political news consumption. First, column (2) shows that the results are broadly similar when including fixed effects for the five educational categories. This reduces the sample because education was not elicited in 2005. Second, at notable efficiency cost, column (3) reports the difference-in-differences results including state fixed effects. Although this approach effectively removes observations from states that do not experience an upcoming election in any survey (more than half the sample), the results are surprisingly stable and remain statistically significant in spite of the reduced precision. Third, the results do not depend upon the particular definition of the upcoming election indicator. Columns (4)-(6) instead define an upcoming local election as a survey within two months before election day. Although the number of instances of upcoming elections decreases, the increased point estimates provide further confidence that voters indeed consume most political information before elections. Adopting a linear approach, columns (7)-(9) similarly confirm that voters exhibit greater news consumption and political knowledge as the number of months until the next local election decreases. In sum, the results provide strong evidence that political news consumption indeed tracks the electoral cycle.

5 Local violence and electoral accountability

Given that voters are significantly better informed just before an election, voters may punish incumbent parties for poor performance at the ballot box—even if such poor performance is a weak signal of performance over the entire election cycle. To test this, I exploit plausibly exogenous spikes in the occurrence of homicides before elections to identify the effect of short-term performance indicators coinciding with voter news consumption cycles on the incumbent party's electoral prospects. I then contrast the effect of pre-election homicide shocks with longer-term homicide rates.

5.1 Data

To examine the electoral effects of violence in the run-up to elections I utilize two main sources of data. First, electoral returns for municipal, state and federal elections covering Mexico's c.67,000 electoral precincts were assembled from the Federal Electoral Institute (IFE),³⁰ state electoral institutes, and freedom of information requests. I focus on municipal elections between 1999 and 2013, for which data is widely available across all states; state and federal electoral returns are used to

³⁰IFE recently became the National Electoral Institute (INE).

parse accountability channels below. Since municipal elections generally occur every three years, the full dataset contains around four elections per municipality. Second, I combine the electoral data with the INEGI's monthly municipal homicide data (described in section 3).

The empirical analysis focuses on two measures of incumbent electoral performance: a municipallevel indicator for whether the incumbent party won the election, and the change in the incumbent party's vote share at the precinct level.³¹ Reflecting their persisting power at the state level (see Langston 2003), 53% of incumbent mayors in the sample are from the PRI; 29% are from the PAN and 12% are from the PRD. As noted above, voters generally sanction parties for the performance of their politicians in Mexico's party-centric system.³² As in Brazil (Klašnja and Titiunik forthcoming), the average incumbent party experiences a 5.7 percentage point decline in their vote share, but still wins 54% of races. Turnout are typically around 60%. Municipalities without their own police forces, including the delegations of the Federal District, are excluded from the analysis.³³

5.2 Identification strategy

While the national homicide rate has changed relatively smoothly over time, this masks considerable inter-month volatility within municipalities. At the height of the War on Drugs in 2010, the monthly homicide count in Ciudad Juárez, Chihuahua, oscillated dramatically, e.g. from 394 in September to 477 in October before falling to 242 in November and again rising to 309 in December. Conversely, the median municipality experiences no homicides in any given month, and only one homicide over a year. More generally, month-to-month fluctuations appear fairly random: since 1999, the average municipal homicide count registering a 0.002 month-on-month increase is dwarfed by its 1.7 homicide standard deviation. Moreover, the count is almost exactly as likely to

³¹Similar results obtain for the incumbent vote share level as a robustness check, but I prefer the precision associated with changes in vote. Given that the analysis weights by the number of registered voters in each precinct and standard errors are clustered by municipality, the precinct-level vote share estimates are almost identical to municipal-level estimates. Because some precincts are missing within municipalities, a precinct-level analysis has slightly more power. The local media analysis exploits precinct-level variation.

 $^{^{32}}$ In the 32% of cases where the incumbent won as part of a coalition formed by several parties, and given such coalitions can vary across elections, if the coalition changes I define the incumbent as the party with the largest vote share at the following election. In general, coalitions are dominated by large parties; the three main parties represent more than 90% of incumbents. Table 2 shows that the results are robust to restricting attention to incumbents containing the three largest parties and single-party incumbents.

³³Using the descriptions of municipal public security forces provided by the municipal governments in the 2000, 2002, 2004, 2011 and 2013 ENGM surveys, and imputing data for missing years, I exclude municipalities without a police force or relying on state, federal, community, private, or other security forces. Given the difficulty of assigning responsibility, the few municipalities with inter-municipal and civil association-run police are excluded. The Appendix describes the imputation procedure.

increase as decrease—both in the run-up to elections and in the middle of a government's tenure.³⁴

To identify the electoral effects of homicides occurring at the times when voters consume most politically-relevant news, I exploit *short-term deviations from trend* in the number of homicides around local elections.³⁵ In particular, I compare otherwise similar elections experiencing a short-term spike relative to the municipal homicide rate before the election to elections experiencing a short-term relative decrease. This strategy is preferred to a design examining changes in pre-election homicides rates across elections because it is hard to purge the correlation between such changes and the longer-run homicide trends that I seek to differentiate pre-election shocks from. Nevertheless, Table A12 shows broadly similar results using a difference-in-differences strategy to examine deviations between pre-election homicide rate and the homicide rate over the electoral cycle.

In particular, I define a municipality as "treated" by a pre-election homicide shock if, conditional on experiencing at least one homicide over the four months spanning the penultimate month before an election and the second month after an election, a municipality *m* experiences more homicides in the two months before an election in month τ than the two months after the election:³⁶

$$Homicide \ shock_{m\tau} \equiv \begin{cases} 1 & \text{if } Homicides_{m\tau-2} + Homicides_{m\tau-1} > Homicides_{m\tau} + Homicides_{m\tau+1} \\ & \text{and} \ \sum_{\tau'=\tau-2}^{\tau+1} Homicides_{m\tau'} > 0, \\ 0 & \text{if } Homicides_{m\tau-2} + Homicides_{m\tau-1} \le Homicides_{m\tau} + Homicides_{m\tau+1} \ (2) \\ & \text{and} \ \sum_{\tau'=\tau-2}^{\tau+1} Homicides_{m\tau'} > 0, \\ . & \text{if} \ \sum_{\tau'=\tau-2}^{\tau+1} Homicides_{m\tau'} = 0. \end{cases}$$

Akin to Ferraz and Finan's (2008) comparison of Brazilian municipalities, where federal audit reports were randomly released just before and just after municipal elections, conditioning on

³⁴Across municipalities since 1999, the monthly homicide change was positive in 11.86% of cases and negative in 11.79% of cases. In election months, these shares are 11.90% and 11.87%. In the month preceding an election, the shares are 11.87% and 11.86%. Most changes are zero because most municipalities do not experience a homicide. This ratio is identical when conditioning on a homicide in either the last month or the current month.

³⁵A benefit of the unavailability of the number of homicides reported by broadcast media stations is that the empirical strategy is not susceptible to differential media reporting biases. Table A6 shows that delayed homicide registration, a potential indicator of strategic registration and missing homicides, is balanced over homicide shocks.

³⁶Elections typically occur on the first Sunday of the month, so I define the treatment using the two months prior to the election month (i.e. $\tau - 2$ and $\tau - 1$). For the 9% of elections held on the 16th or later, I use months $\tau - 1$ and τ . I show below that the results are robust to dropping elections not held during the first half of the month.

the occurrence of a homicide around elections produces the relevant counterfactual by extracting homicide trends.³⁷ I use a two-month window to capture the most final months of the campaign and the post-election period before the winner enters office, while covering a sufficiently short span that month-to-month changes in the homicide count are plausibly random. As Figure 2a below shows, this homicide shock equates to an increase of 2.6 homicides per month in shocked municipalities relative to control municipalities.³⁸ After dropping the 30% of municipalities where no homicide is registered in either the two months before or after the election, Figure A1b in the Appendix highlights the final sample.

I estimate the effects of homicide shocks just prior to the election in precinct p of municipality m in election year t using regressions of the form:

$$Y_{pmt} = \beta Homicide \ shock_{mt} + \eta_m + \mu_t + \varepsilon_{pmt}, \tag{3}$$

where Y_{pmt} is either a municipal-level indicator for the incumbent party winning the election or precinct-level change in the incumbent party's vote share relative to the previous election. Municipality and year fixed effects are respectively denoted by η_m and μ_t . Leveraging only withinmunicipality variation controls for all time-invariant municipal-level features, and also reduces noise generated by differences in homicide histories among municipalities receiving the same treatment. All observations are weighted by the number of registered voters.

5.3 Evidence supporting the identification strategy

The key identifying assumption is that, within municipalities, the *timing* of homicides around elections is effectively random. However, although month-to-month changes in homicide rates are highly idiosyncratic, this does not necessarily imply that short-term changes in homicide shocks across elections in a given municipality occurred by chance. This subsection aims to convince readers of the plausibility of the identification strategy that shocked and non-shocked municipalities do not differ on (observable or unobservable) characteristics.

I start by showing that homicide shocks are uncorrelated with a wide variety of observable pre-treatment covariates and homicide pre-trends. First, Table A6 in the Appendix examines 102 time-varying and time-invariant covariates capturing demographic, socioeconomic, media coverage, and political features of each municipality. Only five differences are statistically significant

³⁷These municipalities differ significantly from even municipalities experiencing only a single homicide over the 4-month window. Municipalities without a homicide are less developed, less politically competitive, and more violent.

³⁸The differences in the total number of homicides over the two-month period before versus after the election between treated and control is 10.59 (standard error of 2.80).

at the 10% level.³⁹ In particular, homicide shocks are equally distributed across political variables including the incumbent's previous vote share, the competitiveness of the previous election, and the incumbent's party identity, but also with respect to the number of media stations, the share that own a television or radio, and educational attainment. Second, Figure 2a examines whether municipalities receiving positive and negative shocks exhibit differential pre-trends in homicide rates or different underlying levels of violence. The figure shows that the difference in monthly homicides between treated and untreated precincts is both relatively constant over the 10 months preceding the period used to define homicide shocks and never statistically significant.⁴⁰ In fact, most of the difference reflects the extremely high homicide rate in Ciudad Juárez in 2010. The balance tests in Table A6 also confirm that the average number of homicides in the 12, 6 and 3 months preceding the period defining the treatment do not significantly vary with homicide shocks. Furthermore, I show below that a placebo shock defined 6 months prior to the election does not affect electoral outcomes, and that the results are robust to controlling for homicide levels. Third, consistent with sampling variability rather than strategic manipulation, Figure 2b shows that the distributions of homicides prior to the period defining a shock, two months before elections, and two months after elections are very similar.

I also find no evidence for the more specific concern that DTOs alter the number of homicides around elections to signal their preferred electoral outcome (e.g. Alesina, Piccolo and Pinotti 2016). Table A6 shows that homicide shocks are not significantly correlated with the number of drug-related homicides in the prior year over the 2006(Dec.)-2011 period when monthly data was made publicly available by the Mexican National Security System. Homicide shocks are also balanced across municipalities registering more than one drug-related homicide in any pre-election year over this period, and no more likely to occur in the 5% of precincts designated by the IFE as high-risk (typically locations with high DTO activity). In addition, I show below that the results are robust to removing municipalities with high levels of drug-related homicides and states with high DTO presence.

Nevertheless, the types of homicide could change without affecting overall levels. Gangland killings are typically concentrated among young and uneducated men, and are often committed using firearms or more gruesome methods—particularly if intended as signals. Using the International Classification of Diseases codes in INEGI's coroner reports, Table A6 also examines the causes of death and victim characteristics of homicides occurring in the two months before an election. In particular, homicide shocks are well balanced with respect to the share of homicides

³⁹Municipality fixed effects are excluded for the time-invariant 2010 Census variables. Table A7 shows analogous tests for precinct-level outcomes.

⁴⁰Fewer homicides in treated municipalities would downwardly bias estimates if voters sanction greater violence.



(a) Difference in the number of homicides between municipalities with a positive and negative homicide shock, by month until the election (95% confidence intervals)

(b) Distribution of homicides by relation to election

Months 5 and 6 before election

Months 1 and 2 before election

Months 3 and 4 before election

Months 1 and 2 after election

Figure 2: Distribution of pre-election homicide shocks

Notes: Each bar in Figure 2a denotes the difference in the number of homicides in treated (positive shock) and control (negative shock) municipalities in the ten months prior to the period defining the homicide shock around the election. Each estimate is from a regression akin to the main estimating equation (3).

caused by a firearm, hanging, or chemical substances. Of relatively frequent types of homicide, only cutting implements register a slight increase. Furthermore, there is no evidence that such homicides disproportionately afflict young, male, unmarried, or uneducated individuals that are most likely to be involved with organized crime.

A different potential concern is that homicide rates change around elections because effective governments can crack down on crime prior to win votes (e.g. Levitt 1997), or because election outcomes themselves impact post-election homicides rates. Although no expert on municipal politics that I interviewed believed that municipal governments crack down on local homicides, I examine this potential concern more systematically. First, proxies for state capacity—including municipality size and budget, police per voter, alignment with Governors or the President, and neighbor homicide shocks—are uncorrelated with homicide shocks, while if anything mayors controlling a municipal police force are more likely to experience a shock. This indicates no differential ability to engage in such crackdowns. Second, using Osorio's (2015) newspaper-based measures of actual DTOs crackdowns, Table A6 demonstrates that homicide shocks are uncorrelated with violent enforcement, drug-related arrests, asset seizures, drug seizures and gun seizures in the two months before the election. Moreover, Table A8 finds no change in the number of security force employees

per voter in election years.⁴¹ Third, although the new mayor does not enter office until more than two months after the election, Table A9 shows that neither pre-election political variables nor election outcome variables predict violence levels in the two months after an election.⁴² The definition of homicide shocks including post-election homicides is thus unlikely to induce bias.

5.4 Pre-election homicide shocks harm municipal incumbent parties

Table 2 shows that pre-election homicide shocks severely hinder the municipal incumbent party's electoral prospects. Column (1) of panel A demonstrates—for this study's main outcome—that such a homicide shock causes an 12 percentage point decline in the incumbent party's probability of being re-elected. This equates to a 22% reduction in the mean mayor's re-election probability. Column (1) of panel B reports that this decline reflects a 2.2 percentage point reduction in the incumbent party's vote share.⁴³ This equates to 0.45 percentage points for each additional homicide per month relative to the baseline levels in the treatment and control group.

These findings are robust to various checks. First, I show that the results are not sensitive to tests of the identifying assumptions. Column (2) of Table 2 first includes 32 time-varying municipallevel covariates to show that the results do not reflect imbalances on observables.⁴⁴ Column (3) shows that the results do not depend upon including municipality and year fixed effects. Given the baseline fixed effect structure, the greatest identification threat reflects time-varying unobservables. To address this, column (4) includes state-year fixed effects to fully control for election-specific state shocks. Column (5) and (6) further include incumbent-specific (for the three largest parties) and municipality-specific linear trends to ensure that the results are respectively not driven by differential trends across party-incumbent performance and municipality. The standard errors for municipality-specific trends increase substantially in this short panel, but the point estimates actually in magnitude. Finally, I conduct a placebo test where a homicide shock is defined six months earlier according to equation (3). The results in column (7) show that such pre-campaign shocks, which could be indicative of pre-trends, do not affect incumbent electoral performance.

⁴¹Although such data are only available from the National Census of Municipal Governments (ENGM) on an annual basis over five waves, I was able to impute 9,655 municipal-years.

⁴²Dell (2015) finds that homicide rates increased where PAN mayors were elected during the Calderón administration, but such mayors do not enter office within two months of the election. More generally, post-election homicides are also uncorrelated with interactions between election outcomes and background covariates.

⁴³Consistent with the model in Arias et al. (2017), unreported results indicate that the change in vote share reflects both an increase in turnout, but also a reduction in the total votes (as a share of registered voters) for the incumbent party.

⁴⁴To preserve sample size, I exclude from the control set variables with significant missingness, namely homicide victim characteristics, threatened precincts (federal elections only), municipal spending, and child mortality.

	Baseline spec.	Controls	No fixed effects	State- year effects	Incumbent party trends	Municipality -specific trends	Placebo 6 months earlier	Violence bin effects	Fewer than 25 homicides per month	Few drug homicides	Non- DTO states	One- month shock	Three- month shock	Four- month shock	First half of month	Main party incumbent	Single- party incumbent
	(1)	(2)	(3)	(4)	(5)	(9)	(£)	(8)	(6)	(10)	(11)	(12)	(13)	(14)	(15)	(16)	(17)
Panel A: Outcome: Inci	mbent party v	win															
Homicide shock Placebo homicide shock Homicide shock (one month) Homicide shock (fruree month) Homicide shock (four month)	-0.119**	-0.086* (0.051)	-0.086** (0.034)	-0.097** (0.040)	-0.108** (0.052)	-0.124 (0.091)	-0.017 (0.071)	-0.126*** (0.046)	-0.107** (0.053)	-0.119** (0.050)	-0.108* (0.065)	-0.047 (0.063)	-0.110** (0.053)	-0.085* (0.050)	-0.116** (0.056)	-0.120** (0.051)	-0.136* (0.080)
Observations Outcome range Outcome mean Homicide measure mean	3,314 $\{0,1\}$ 0.55 0.42	3,279 $\{0,1\}$ 0.55 0.42	3,314 $\{0,1\}$ 0.55 0.42	3,314 $\{0,1\}$ 0.55 0.42	3,042 $\{0,1\}$ 0.55 0.42	3,314 $\{0,1\}$ 0.55 0.42	2,912 $\{0,1\}$ 0.54 0.42	3,314 $\{0,1\}$ 0.55 0.42	3,293 $\{0,1\}$ 0.54 0.43	3,314 $\{0,1\}$ 0.55 0.42	2,261 $\{0,1\}$ 0.53 0.41	2,348 {0,1} 0.55 0.48	3,926 $\{0,1\}$ 0.54 0.43	$\begin{array}{c} 4,390 \\ \{0,1\} \\ 0.54 \\ 0.42 \end{array}$	3,075 $\{0,1\}$ 0.55 0.42	3,042 $\{0,1\}$ 0.55 0.42	2,276 $\{0,1\}$ 0.54 0.40
Panel B: Outcome: Cha	nge in incumb	ent party vote	share														
Homicide shock Placebo homicide shock Homicide shock (one month) Homicide shock (frow month) (four month)	-0.022*** (0.009)	-0.018*** (0.007)	-0.017**	-0.018*** (0.007)	-0.015* (0.008)	-0.030*** (0.011)	-0.004 (0.012)	-0.026*** (0.009)	-0.023** (0.009)	-0.02 *** (0.009)	-0.025** (0.010)	-0.015 (0.011)	-0.024*** (0.009)	-0.027*** (0.009)	-0.019** (0.009)	-0.022*** (0.008)	-0.033*** (0.012)
Observations Outcome range Outcome mean Outcome std. dev. Homicide measure mean	174,148 [-0.96,0.89] -0.05 0.14 0.42	172,956 [-0.96,0.89] -0.05 0.14 0.42	174,148 [-0.96,0.89] -0.05 0.14 0.42	174,148 [-0.96,0.89] -0.05 0.14 0.42	165,467 [-0.96,0.89] -0.05 0.14 0.42	174,148 [-0.96,0.89] -0.05 0.14 0.47	153,861 [-0.96,0.87] -0.05 0.14 0.42	174,148 [-0.96,0.89] -0.05 0.14 0.42	162,299 [-0.96,0.89] -0.05 0.15 0.43	174,148 [-0.96,0.89] -0.05 0.14 0.42	107,964 [-0.96,0.89] -0.06 0.15 0.41	151,826 [-0.96,0.89] -0.05 0.14 0.47	186,276 [-0.96,0.89] -0.05 0.15 0.43	193,610 [-0.96,0.89] -0.06 0.15 0.42	160,216 [-0.96,0.89] -0.05 0.15 0.41	165,467 [-0.96,0.89] -0.05 0.14 0.42	115,342 [-0.96,0.89] -0.05 0.15 0.40
Notes: Columns (2 (5) includes party- over the four-moni drug-related homi and four-month wi) includes the specific incurr h window in b ide a month o idows. Colum	controls in Tat abent trends. C ins of size ten. ver the 12 moi n (15) include	le A6, with th Jolumn (6) inc Column (9) e aths before an s only election	e exception o ludes munici xcludes muni election betw s that take pla	f the variable: pality-specific icipalities aver een 2006 and ace on or befo	s listed in footn : trends. Colun raging more the 12011. Column re the 15th of t	ote 44 with nc m (7) uses a 1 m 25 homicid n (11) exclude the month. Cc	n-trivial miss placebo homia es per month es states with olumn (16) inc	ingness. Colun cide shock defin over the two m high-level of D tudes only elec	nn (3) exclude ned six month onths either be TO activity (s :tions where th	s municipalit s before the e efore or after t ee footnote 4 te PAN, PRD,	y and year fixe lection. Colui the election. C 5). Columns (or PRI are an	d effects. Col mn (8) includd Jolumn (10) e: 12)-(14) respe i incumbent. C	umn (4) inclu ss fixed effect ccludes munic ctively define column (17) ir	des state-year s for the total ipalities that <i>i</i> homicide sho reludes only o	fixed effects. number of hc werage more t ocks over one- bservations w	Column micides han one , three., here the
incumbent is not a denotes $p < 0.05$.	coalition. All	specifications < 0.01.	are estimated	using OLS, ai	nd all observa	tions are weigh	ted by the nu	mber of regist	ered voters in t	he electoral pr	ecinct. Stand	ard errors clus	tered by mun	cipality are in	parentheses.	* denotes $p <$: 0.1, **
uctions $p < v.v.r$	d canonan	< U.UL															

Table 2: The effect of a pre-election homicide shock on municipal incumbent electoral outcomes

Second, the results are not driven by municipalities experiencing particular homicide levels or types of homicide. By including fixed effects for the total number of homicides over the fourmonth window (in 10-homicide bins), column (8) provides further evidence that the results are not driven by differing homicide levels.⁴⁵ Moreover, column (9) excludes municipalities experiencing more than 25 homicides per month to show that the results do not simply reflect a few large municipalities with many homicides. Although homicide shocks are uncorrelated with indicators of DTO presence, to further address the potential concern that the results are driven by strategic political violence in areas with high drug-related criminal activity, I also exclude the elections most vulnerable to strategic political violence by DTOs: column (10) excludes municipalities that average more than one drug-related murder a month over any pre-election year between 2006 and 2011 (when such data were collected), while column (11) excludes nine states with high DTO-related drug crime.⁴⁶ In both cases, the estimates do not substantially change.

Third, the results do not reflect particular definitions of homicide shocks or electoral outcomes. First, columns (12)-(14) show that homicide shocks defined by one-, three-, or four-month windows also reduce the incumbent's vote share by around 2.5 percentage points and the incumbent's probability of winning by 5-11 percentage points. The smaller one-month window estimates could reflect the smaller sample or weaker signals imparted by such shocks. Column (15) shows that the results are robust to removing elections that occurred in the second half of the month, which are more challenging to classify with respect to homicides occurring before the election. Column (16) restricts attention to PAN, PRD, and PRI incumbents, demonstrating that the results are robust to remove from smaller parties. Similarly, column (17) shows that the results are robust to restricting attention to single-party incumbents.

Together, these results robustly show that an inopportune temporary increase in violence can have substantial electoral implications for an incumbent when it coincides with the pre-election period when voters are more likely to acquire political information.

5.5 Negligible effects of longer-run homicide rates

The political information cycles argument implies that voters are particularly likely to respond to homicides that occur just before elections. Longer-run homicide metrics may not affect responsibility attribution if enough voters are insufficiently politically-engaged at other times in the political cycle, even if voters (at least intend to) punish longer-term performance (see Healy and

⁴⁵I obtain similar results using smaller bin sizes, including size 1, at the cost of absorbing large municipalities with such fixed effects.

⁴⁶Baja California, Chihuahua, Durango, Guerrero, Michoacán, Nuevo León, Sinaloa, Sonora, and Tamaulipas.

Lenz 2014; Healy and Malhotra 2013).

To examine whether homicide shocks before elections indeed induce greater incumbent sanctioning than long-run homicide rates, I use a difference-in-differences design to estimate the effects of homicides over the prior year and prior (three-year) electoral cycle on the incumbent party's electoral prospects. Specifically, I compare changes in support for incumbent parties in municipalities that experienced large increases in their longer-run homicide rate between elections to changes in support for incumbent parties in municipalities that did not using the following specification:

$$Y_{pmt} = \beta Average monthly homicide rate_{mt} + \eta_m + \mu_t + \varepsilon_{pmt}.$$
(4)

As a robustness check, I also include municipality-specific time trends to adjust for differential trends in incumbent support across municipalities.

The results in Table 3 indicate that long-run homicide trends have had limited impact on electoral performance. The point estimates in columns (1), (2), (5), and (6) of panel A show that homicides over the year before the election are not significantly correlated with either the incumbent's probability of being re-elected or their vote share, regardless of the inclusion of municipality-specific time trends. In fact, the estimates in columns (5) and (6) suggest that 5 more homicides a month over the year before an election—almost half a standard deviation increase—only translates into a 0.25 percentage point decline in the municipal incumbent party's vote share. This effect is around ten times smaller than the five-homicide average differential between shocked and non-shocked municipalities. Columns (3), (4), (7), and (8) include a quadratic term, but also find little suggestion that any effect is non-linear. Panel B similarly shows that voters do not punish poor performance over the full electoral cycle. This difference mirrors the findings of Healy and Lenz (2014), who suggest that U.S. voters wish to punish performance over the full electoral cycle but rely on more recent information in its absence. The evidence thus suggests that longer-run homicide trends play a substantially smaller role than short-term homicide shocks in informing electoral accountability.

5.6 Voters distinguish responsibility across levels of government

Although mayors play an important role in local public security, state and federal governments are also important players. The punishment of municipal mayors could then reflect broader punishment of the party controlling higher office. Conversely, if voters believe that mayors are primarily responsible for local crime (or that their actions are only weakly correlated with co-partisans at higher levels), they may update less about national parties following local homicide shocks. To

		Incumbent	t party win		Change	in incumbe	ent party vo	te share
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)
Panel A: 12 months before election								
Average monthly homicide rate (12 months before election) Average monthly homicide rate (12 months before election) squared	0.0009 (0.0020)	-0.0031 (0.0028)	$\begin{array}{c} -0.0009 \\ -0.0080 \\ 0.0000 \\ 0.0000 \end{array}$	-0.0007 (0.0114) -0.0000 (0.0000)	-0.0004 (0.0003)	-0.0006) (0.0006)	-0.0014 (0.0012) 0.0000 (0.0000)	-0.0023 (0.0024) 0.0000 (0.0000)
Ohservations	3 404	3 404	3 404	3 404	174 622	174 622	174 622	174 622
Outcome mean	0.47	0.47	0.47	0.47	-0.05	-0.05	-0.05	-0.05
Homicide rate mean	8.13	8.13	8.13	8.13	8.12	8.12	8.12	8.12
Homicide rate standard deviation	21.92	21.92	21.92	21.92	22.01	22.01	22.01	22.01
Municipality-specific time trends		>		>		>		>
Panel B: Since last election								
Average monthly homicide rate (3 years before election) Average monthly homicide rate (3 years before election) squared	0.0027 (0.0024)	-0.0029 (0.0058)	0.0039 (0.0109) -0.0000 (0.0001)	0.0072 (0.0215) -0.0001 (0.0001)	-0.0002 (0.0003)	0.0000) (0.0008)	-0.0004 (0.0014) 0.0000 (0.0000)	0.0001 (0.0033) -0.0000 (0.0000)
Observations	3,314	3,314	3,314	3,314	174,148	174,148	174,148	174,148
Outcome mean	0.47	0.47	0.47	0.47	-0.05	-0.05	-0.05	-0.05
Homicide rate mean	6.97	6.97	6.97	6.97	6.63	6.63	6.63	6.63
Homicide rate standard deviation Municipality-specific time trends	15.07	15.07 ✓	15.07	15.07 ✓	15.09	15.09 ✓	15.09	15.09 ✓
<i>Notes</i> : All specifications are estimated using registered voters for the relevant electoral unit $***$ denotes $p < 0.01$.	OLS, and inc t. Standard er	lude municip rors clusterec	ality and yea I by municipa	r fixed effect dity are in pa	s. All observ rentheses. * (ations are we denotes $p < 0$	ighted by the).1, ** denote	number of s $p < 0.05$,

Table 3: The effect of long-run homicide rates on municipal incumbent electoral outcomes

disentangle these chains of accountability and assignment of responsibility, and thus assess hypothesis H4, I first examine how the effects of homicide shocks vary with the existence of a municipal police force, before comparing the electoral performance of parties at the municipal, state and federal levels using data from simultaneous municipal, state and federal elections, and then finally examining survey-level voter confidence in higher levels of government.

If voters believe that the mayor is responsible for local crime rates because they control the local police, homicide shocks should only be punished in municipalities with their own police force. In contrast with the preceding analysis which excluded municipalities without their own police force, I now include such municipalities (including delegations in the Federal District) to test whether possessing a police force influences electoral sanctioning. Panel A of Table 4 show that, consistent with voters recognizing that elevated homicides rates may be beyond the control of municipal mayors lacking local police forces, homicide shocks only significantly affect the electoral prospects of incumbent parties commanding a local police force. Columns (2) and (4) address the concern that the effect attributed to police forces reflects the lack of sanctioning in smaller and less developed municipalities by controlling for the interaction of a homicide shock with previous incumbent party vote share and the average monthly homicide rate over the last year. At least when they consume politically-relevant news, these results echo previous research suggesting that voters in developing contexts differentiate incumbent performance from external forces (Harding and Stasavage 2014; Kronick 2014).⁴⁷ Combined with the evidence above that voters learn from homicide shocks, this finding further suggests that voters are not indiscriminately punishing events like American football losses (Healy, Malhotra and Mo 2010).

Nevertheless, if the parties of municipal mayors, state deputies, state Governors, or the President are correlated, the substantial electoral penalties found above could reflect punishment of other political actors. To test for such spillovers down layers of government, columns (1)-(3) of panel B in Table 4 respectively examine the effect of a homicide shock on the municipal vote share of the party of the state deputy, state Governor, and the presidency. The results, however, suggest that if anything the Governor and President's parties increase their vote share, and thus provide no support for punishment at higher levels of government spilling over to municipal elections.⁴⁸ Nevertheless, could still reserve punishment for state and federal politicians. However, columns (4) and (5) similarly indicate that a homicide shock reduces neither the vote share of state nor federal deputies in concurrent state and federal elections.

Although I find no evidence that state and federal incumbent parties are held responsible for

⁴⁷Table A14 provides mixed evidence that voters account for homicide shocks affecting neighboring municipalities.

⁴⁸Consistent with Table A15 in the Appendix, the positive effect at the federal level reflects increased support for Calderón's PAN government.

Panel A: Presence of	Incu	imbent	Change in	incumbent			
municipal police force	par	ty win	party vo	te share			
	(1)	(2)	(3)	(4)			
Homicide shock	-0.122**	-0.095**	-0.023***	-0.024***			
	(0.047)	(0.047)	(0.008)	(0.007)			
No municipal police force	0.002	0.023	0.006	-0.006			
	(0.126)	(0.125)	(0.016)	(0.014)			
Homicide shock	0.236	0.234	0.066***	0.051**			
\times No municipal police force	(0.158)	(0.147)	(0.023)	(0.022)			
Observations	3,825	3,788	200,976	200,976			
Outcome range	$\{0,1\}$	$\{0,1\}$	[-0.96,0.89]	[-0.96,0.89]			
Outcome mean	0.48	0.48	-0.04	-0.04			
Homicide shock mean	0.41	0.41	0.42	0.42			
No municipal police force mean	0.09	0.09	0.13	0.13			
Test: Homicide shock +	0.43	0.31	0.04	0.19			
Homicide shock \times No municipal							
police force = $0 (p \text{ value})$							
Interactive controls		\checkmark		\checkmark			
Panel B: Sanctioning across	Change i	n municipal v	ote share of	Change in	Change in	Change in	Change in
levels of government	state	state	President's	state	federal	state vote	federal vote
	district	Governor's	party	vote share	vote share	municipal	municipal
	party	party			incumbent	incumbent	
	(1)	(2)	(3)	(4)	(5)	(6)	(7)
Homicide shock	0.001	0.027*	0.032*	-0.005	-0.005	0.008	-0.017
	(0.009)	(0.014)	(0.018)	(0.009)	(0.015)	(0.016)	(0.017)
Observations	149,341	141,857	174,148	140,692	54,750	133,239	64,821
Outcome mean	0.00	0.03	0.04	-0.04	-0.05	0.45	0.39
Homicide shock mean	0.42	0.42	0.42	0.41	0.41	0.42	0.43

Table 4: Effects of homicide shocks across levels of government with differing responsibilities

Notes: All specifications include municipal and year fixed effects, weight by the number of registered voters, and are estimated using OLS. The interactive controls in panel A are lagged incumbent party vote share and average monthly homicides over the last year (both standardized). Standard errors clustered by municipality are in parentheses. * denotes p < 0.1, ** denotes p < 0.05, *** denotes p < 0.01.

homicide shocks by voters, punishment of the local party could instead spill up to the national level. However, when municipal and higher-level elections are held simultaneously, columns (6) and (7) of panel B show that the state and federal deputy vote shares of the municipal incumbent's party are not affected by a homicide shock. Table A15 also shows that voters do not differentially punish different parties for homicide shocks.

6 The moderating effect of local media

Bringing together the individual- and aggregate-level findings, I argue that the electoral sanctioning of local violence before elections depends upon access to a media environment covering local



Figure 3: Media station commercial quality signal coverage areas

news. To demonstrate that the sanctioning of homicide shocks reflects such media coverage, I exploit precinct-level variation in media signal coverage to identify the effect of an additional local radio or television station when a pre-election homicide shock occurs.

6.1 Data

As part of Mexico's major media reform in 2007 (see Larreguy, Marshall and Snyder 2016; Serra 2012), the IFE required that all radio and television stations in the country submit detailed coverage and technological data. This included the location and power of their antennae and a map detailing their commercial quality coverage—the level of coverage that media stations may base advertising sales on and which the IFE deemed relevant for minimizing cross-state advertising spillovers. The signal inside the commercial quality coverage area is very strong, and should cover virtually all households, while signal quality declines quickly as the distance from the commercial quality coverage boundary increases.⁴⁹

Figure 3 maps the commercial quality coverage of each type of media station. While FM radio, and especially television, stations have limited and primarily urban coverage, AM radio stations can travel considerable distances—particularly when aided by stretches of sea water with high electrical ground conductivity. Virtually all of the population is covered by at least one media station, but the number of outlets—both emitting from within a municipality and without—providing commercial quality signals to any given electoral precinct varies considerably. Furthermore, the extent to which a given electoral precinct is covered by a signal varies substantially: in some cases,

⁴⁹The IFE defines the boundary of the coverage area using a 60 dB μ threshold for signal strength. Accordingly, to the U.S.-based National Communications and Information Administration, this "60 dB μ level is recognized as the area in which a reliable signal can be received using an ordinary radio receiver and antenna." Outside this area, a high-performance antenna is typically required to avoid receiving a weak signal. This is the threshold commonly used to determine a radio station's audience and sell advertising space commercially in the United States.

a signal only covers a tiny fraction of a precinct, while in others the entire precinct is covered. I follow Larreguy, Marshall and Snyder (2017) by defining a precinct as covered by a given media station only if at least 20% of voters fall within the commercial coverage boundary.⁵⁰

6.2 Identification strategy

To identify the electoral effects of media stations covering local homicides, I leverage geographic variation in coverage (see also Larreguy, Marshall and Snyder 2017).⁵¹ I compare neighboring electoral precincts within the same municipality that differ in the total number of *local* AM, FM, and television stations—defined as outlets that provide local content⁵² and whose antennae are located within the electoral precinct's municipality—by which they are covered.⁵³ Since signal quality declines, but without disappearing entirely, across the commercial quality boundary, this strategy identifies the "intent to treat" effect of an increase in the probability of exposure to an additional local media station.⁵⁴ Figure 4 illustrates the identification strategy graphically, using the example of electoral precincts 1571 and 1583 in the municipality of Villa de Tututepec de Melchor Ocampo in Oaxaca. The identifying assumption is that these neighboring precincts differ only because precinct 1583 receives a commercial quality signal from a local television station that does not cover precinct 1571.

Operationally, for each electoral precinct in the country I identify the set of neighboring precincts from within the same municipality that differ in the number of local media stations that they are covered by. Each such grouping n is defined by a "treated" precinct and the set of neighboring precincts receiving different local media exposure. Focusing on municipalities included in the homicide shock sample produced 2,689 neighboring groups, containing an average of 2.3 comparison units per election. The average precinct is covered by 6.3, 8.9, and 3.2 local AM, FM, and television stations respectively, while the total number of local media stations covering a precinct ranges from 0 to 40.⁵⁵

⁵⁰The INEGI provides detailed Census population counts in 2010, for both rural localities and urban blocks. This data is used to identify the proportion of voters affected covered by a commercial quality signal.

⁵¹Ansolabehere, Snowberg and Snyder (2006), Enikolopov, Petrova and Zhuravskaya (2011), Snyder and Strömberg (2010), and Spenkuch and Toniatti (2016) use similar designs.

⁵²This excludes television stations solely retransmitting Televisa and TV Azteca national broadcasts.

⁵³Although data does not exist to adjust for news consumption "non-compliance," any effect would be larger among compliers that only receive news because they were exposed to an additional commercial quality local signal.

⁵⁴Exposure to commercial quality coverage does not constitute a geographic regression discontinuity design because the treatment is not binary (some neighbors differ by more one media station) and where neighbors differ by more than one media station it is not clear how to coherently define the running variable.

⁵⁵Given that data from the Secretariat of Communications and Transportation show that the number of media stations has not changed since 2003, I continue to pool the years 1999 to 2013.



Figure 4: Identification strategy example

Notes: Both precincts are located in the municipality of Villa de Tututepec de Melchor Ocampo in the state of Oaxaca. Precinct 1583 is covered by the local television emitting from within the municipality, but precinct 1571 is not.

There are good reasons to believe that, among neighboring electoral precincts, local media coverage at the commercial quality boundary is effectively random. First, neighboring precincts often differ in coverage because of physical characteristics such as water, geographic contours and large physical objects that aid or impede ground conductivity (in the case of AM radio) and "line of sight" radio waves (in the case of FM radio and television) between an antenna and precincts at the coverage boundary. Second, given that media stations lack the technology to differentiate media markets at fine-grained levels,⁵⁶ and voters that specifically locate according to the availability of local media are unlikely to choose to live close to a coverage boundary (preferring a location guaranteeing coverage), it is unlikely that coverage reflects strategic sorting by either media stations or voters. Finally, I restrict attention to neighboring precincts with an area of less than 2km². This removes larger precincts where media outlets could more plausibly target their signals, and prevents comparisons between urban and suburban, or suburban and rural, precincts that may differ in their electoral behavior. The final sample is shown in Figure A1c.

Combining within-neighbor variation in local media coverage with homicide shocks just before an election,⁵⁷ I estimate the following specification to identify the interaction between homicide

⁵⁶The IFE technical data show that the power of signal transmitters are fairly discrete. The power output in watts for AM, FM and television stations is almost exclusively round thousands that are divisible by five.

⁵⁷Homicide shocks remain well-balanced across pre-treatment variables in this subsample.

shocks and local media coverage:

$$Y_{pmnt} = \beta_1 Homicide \ shock_{mt} + \beta_2 Local \ media_{pt} + \beta_3 \left(Homicide \ shock_{mt} \times Local \ media_{pt} \right) + \zeta_n + \mu_t + \varepsilon_{pmnt}, \tag{5}$$

where ζ_n is a fixed effect for each set of neighboring precincts that ensures I only exploit withinneighbor variation in local media.⁵⁸ To weight each neighboring group equally, precincts are weighted by electorate size divided by the number of control units per neighbor set in order.

To assess the plausibility of exogenous differences in local media coverage between neighboring precincts, I use equation (5) to examine balance across a wide range of demographic, socioeconomic, homicide, and political municipal and precinct characteristics. Table A10 shows that only nine of these 109 variables are significantly correlated with the number of local media stations at the 10% level. In particular, there are no significant differences in key indicators of rural-urban geography (such as precinct area, electorate size, or distance to the municipality head), the number of non-local media stations, homicide indicators, or previous electoral outcomes. The few significant differences are small in magnitude.

6.3 Local media increase punishment of homicide shocks

The estimates in Table 5 show that access to local media stations plays a critical role in supporting the electoral sanctioning of homicide shocks. The key finding in column (1) shows that homicide shocks are only punished when an electoral precinct is covered by sufficient local media stations. In the relatively rare case of precincts with fewer than eight local media stations (19% of the sample), the marginal effect plot in Figure 5 indicates that there is no significant effect of a homicide shock. Implying that local media are necessary for Mexican voters to punish the incumbent party for pre-election violence, the effect of a homicide shock in an electoral precinct with no local media stations is close to zero. However, for more than eight local media stations, the significant interaction coefficient kicks in and homicide shocks substantially reduce the vote share of the municipal incumbent party. Like Ferraz and Finan (2008) and Larreguy, Marshall and Snyder (2017), the significant positive coefficient on the lower-order local media term suggests that voters may reward incumbents that do not experience a homicide shock before the election. In contrast, column (2) shows no significant relationship with homicides over the last year.

Columns (3)-(5) examine which types of local media contribute most to the electoral sanc-

⁵⁸Since I use within-municipality neighbors, neighbor fixed effects incorporate municipality fixed effects.
		Char	nge in incumbe	ent party vote s	share	
			Media n	neasure:		
	Local	Local	Local AM	Local FM	Local	Non-local
	media	media	media	media	television	media
	(1)	(2)	(3)	(4)	(5)	(9)
omicide shock	-0.0195		-0.0045	-0.0620*	-0.0239	-0.0050
	(0.0245)		(0.0326)	(0.0331)	(0.0205)	(0.0182)
ledia measure	0.0017*	-0.0001	0.0025	0.0009	0.0055	0.0000
	(0.0010)	(0.0003)	(0.0035)	(0.0018)	(0.0041)	(0.0006)
omicide shock	-0.0022**		-0.0023	-0.0012	-0.0175^{***}	-0.0006
\times Media measure	(0.0010)		(0.0042)	(0.0026)	(0.0058)	(0.0004)
verage monthly homicide rate		-0.0639***				
(12 months before election)		(0.0165)				
verage monthly homicide rate		0.0004				
× Media measure		(0.0008)				
bservations	30,099	30,099	1,385	16,345	13,422	95,160
utcome range	[-0.63, 0.50]	[-0.63, 0.50]	[-0.50, 0.26]	[-0.63, 0.41]	[-0.62, 0.47]	[-0.75, 0.73]
utcome mean	-0.04	-0.04	-0.02	-0.03	-0.04	-0.05
omicide measure mean	0.40	10.39	0.62	0.33	0.44	0.46
fedia measure mean	18.40	18.40	3.40	10.81	2.92	35.20
fedia measure std. dev.	10.80	10.80	4.20	5.18	1.64	23.21

re of local and non-local media ţ conditional on the Table 5. Electoral effects of homicide rates registered voters divided by the number of comparison units within each neighbor group. Standard errors clustered by municipality are in parentheses. * denotes p < 0.05, *** denotes p < 0.05, *** denotes p < 0.01.



Figure 5: Electoral effect of a homicide shock, conditional on the number of local media stations (95% confidence interval)

Notes: Calculated using the estimates from column (1) of Table 5. The gray density plot above the x axis plots the distribution of the local media variable in the sample.

tioning and rewarding of incumbent parties. Since radio and television coverage is correlated, to identify the contribution of each—albeit in different samples—I exploit within-neighbor differences in the number of local AM, FM, and television stations separately. Given that television is by far the primary media source for Mexicans, it is not surprising to find that local television stations produce the largest effects. As column (5) demonstrates, each additional local television station reduces the vote share of an incumbent facing a homicide shock by 1.8 percentage points. The smaller effects of FM and especially AM radio are relatively similar in magnitude to the average effect reported in column (1), but are not statistically significant in these subsamples.

To assess the importance of local media stations, as opposed to media stations emitting from other municipalities, I use the same strategy to identify the effects of non-local media. Consistent with the findings of Larreguy, Marshall and Snyder (2017), column (6) shows that non-local media stations with weaker incentives to report homicides outside their own municipality do not significantly influence voter sanctioning and produce marginal effects 3.5 times smaller in magnitude. This reiterates the role of receiving politically-relevant news in electoral sanctioning.

7 Mechanisms

The preceding analysis provides clear evidence that voters consume most political information just before elections, and are highly responsive to homicides that occur at this time, but only when covered by local media stations likely to report such news. To understand the theoretical underpinnings of these findings, I now explore the mechanisms driving such political information cycles. In particular, I provide a variety of evidence suggesting that voters are updating their beliefs in a somewhat sophisticated manner, before showing that political information cycles are more likely driven by changes in voter demand for news than changes in media coverage around elections.

7.1 Voters behave as Bayesian with weak prior beliefs

To examine the extent to which the voter behavior underpinning political information cycles, I examine whether belief updating is consistent with voting behavior and how beliefs and behavior vary with the position and precision of priors beliefs.

7.1.1 Voters update their beliefs from homicide shocks

Given the salience of crime in Mexico, greater attention to news in the run-up to an election is likely to increase exposure to violent crime in the local media. However, exposure to information about homicides should only affect voting behavior to the extent that voters update their posterior beliefs about the salience of public security and/or the performance of their municipal institutions in addressing this key valence issue. To test whether these prerequisites for holding municipal incumbents to account hold, this subsection examines whether the content of the news during local election campaigns affects voter beliefs by extending equation (1) to allow the effect of upcoming elections to vary with municipal homicide experiences:

$$Y_{imt} = \beta_1 Upcoming \ local \ election_{mt} + \beta_2 X_{mt} + \beta_3 \left(Upcoming \ local \ election_{mt} \times X_{mt} \right) + \mu_t + \varepsilon_{imt}, \ (6)$$

where X_{mt} is either a pre-survey homicide shock (defined analogously to pre-election homicide shocks),⁵⁹ or the monthly homicide rate over the one or three years prior to the survey.⁶⁰ Given that the former measure is more likely to be reported in the media, the political information cycles argument implies that only recent homicide shocks are likely to influence voter beliefs. The combination of plausibly exogenous variation in upcoming local elections and homicide shocks suggests

⁵⁹Since the day of the survey varies, but homicide data is monthly, a homicide in the month of the survey could occur after the survey was carried out. I thus use an indicator for a homicide occurring in either the month of the survey or the prior month.

⁶⁰Table A2 indicates that neither measure is significantly correlated with upcoming local elections.

a causal interpretation of the interaction effect β_3 .⁶¹ The longer-term measures are less likely to be covered directly in the news, although well-informed voters may have already internalized such information.

I use the ENCUP survey to measure the salience of violence and beliefs about the competence of the incumbent party.⁶² To evaluate whether homicides around elections translate into concern about public security, I define an indicator for respondents citing crime and insecurity, drug trafficking, violence, or vandalism as the most important problem for their community to solve.⁶³ I also create an indicator of low confidence in the municipal mayor, defined by respondents ranking their confidence at 5 or below on a scale from 0 to 10 (for 2012) or expressing no, or almost no, confidence in the given institution (in 2001).⁶⁴

The results in panel A of Table 6 demonstrate that homicides just before an election substantially increase fears regarding public security. The insignificant effects of the lower-order upcoming local election and homicide shock terms, combined with the large positive interactions in columns (1) and (2), show that homicide shocks only increase concerns about security among voters in municipalities experiencing an upcoming local election.⁶⁵ Relative to the sample mean, a homicide shock increases security concerns by almost 50% during an election campaign.

The insignificant lower-order homicide shock term indicates that voters do not update about public security from short-run homicide spikes during periods of lower news consumption. This suggests that it is not the case that voters forget about homicides that happened earlier in the electoral cycle (Zaller 1992). Rather, they never internalized this information in the first place. This finding also suggests that voters are not initially frustrated but subsequently won back by incumbent policies designed to appeal to disgruntled voters (Brollo 2009). Rather, the lack of immediate response indicates that voters do not internalize homicide shocks outside of election campaigns.

It remains possible that voters that primarily consume political news before elections are actually responding to longer-run homicide rates. However, the results in columns (3) and (4) interacting upcoming local elections with longer-term homicide measures reject this possibility. The

⁶¹Reinforcing the identification strategy detailed above, Tables A3 and A4 show that short-run homicide shocks, and their interaction with upcoming local elections, are not systematically correlated with individual-level and municipal covariates.

⁶²The Federal District, where policing is not administered at the delegation-level, is excluded.

⁶³Along with all other political concerns, I include as zeroes respondents that do not cite a problem.

⁶⁴The cutoff on the 0-10 scale was chosen to broadly match the distribution of responses in 2001. However, the results do not depend upon the choice of cutoff. Unfortunately, vote intentions were not elicited in any survey wave.

⁶⁵The positive interactions are robust to simultaneously controlling for the interaction of an upcoming election with pre-treatment municipal characteristics. The positive effect of local elections in columns (3) and (4) reflect the fact that the positive interactions in columns (1) and (2) are being averaged across in these specifications.

	Homicide measure:						
	Hom	icide	Hom	icide	Homicides per month		
	sho	ock	per n	nonth			
			last	year	last 3	years	
	(1)	(2)	(3)	(4)	(5)	(6)	
Panel A: Outcome: Public insecurity the major problem in the community							
Upcoming local election	0.024	-0.033	0.034**	0.004	0.034**	0.004	
	(0.021)	(0.023)	(0.016)	(0.017)	(0.016)	(0.016)	
Homicide measure	-0.011	-0.005	0.009***	0.008***	0.009***	0.009***	
	(0.012)	(0.011)	(0.002)	(0.002)	(0.002)	(0.001)	
Upcoming local election	0.092***	0.085***	0.002	0.001	0.002	0.001	
\times Homicide measure	(0.028)	(0.026)	(0.003)	(0.003)	(0.003)	(0.003)	
Observations	9 764	9 764	12 541	12 541	12 541	12 541	
Outcome range	$\{0,1\}$	{01}	{0,1}	{0,1}	{0,1}	{0,1}	
Outcome mean	0.13	0.13	0.11	0.11	0.11	0.11	
Upcoming local election mean	0.22	0.22	0.20	0.20	0.20	0.20	
Homicide measure mean	0.46	0.46	2.99	2.99	3.06	3.06	
Survey year without data	2012	2012	2012	2012	2012	2012	
Panel B: Outcome: Low confidence in the municipal mayor							
Uncoming local election	-0.019	0.175	0.000	0.091	0.001	0.093	
epeering room ereenen	(0.042)	(0.110)	(0.036)	(0.056)	(0.038)	(0.058)	
Homicide measure	-0.048*	-0.019	0.001	0.001*	0.001***	0.001**	
	(0.026)	(0.021)	(0.001)	(0.001)	(0.000)	(0.000)	
Upcoming local election	0.114**	0.069	-0.001	0.010	-0.002	0.012	
\times Homicide shock	(0.048)	(0.146)	(0.009)	(0.012)	(0.015)	(0.021)	
Observations	5,925	5,925	7,236	7,236	7,236	7,236	
Outcome range	$\{0,1\}$	$\{0,1\}$	$\{0,1\}$	$\{0,1\}$	$\{0,1\}$	$\{0,1\}$	
Outcome mean	0.34	0.34	0.34	0.34	0.34	0.34	
Upcoming local election mean	0.02	0.02	0.03	0.03	0.03	0.03	
Homicide measure mean	0.42	0.42	6.33	6.33	6.98	6.98	
Survey years without data	2003, 2005	2003, 2005	2003, 2005	2003, 2005	2003, 2005	2003, 2005	
State fixed effects		\checkmark		\checkmark		\checkmark	

Table 6: Heterogeneous effects of upcoming local elections on concern for public security and institutional confidence, by short-run and long-run municipal homicide measures

Notes: All specifications include survey year fixed effects, and are estimated using OLS. The number of observations in columns (1) and (2) is lower because voters in municipalities failing to experience any homicide over the two months either side of the survey were dropped. Standard errors clustered by municipality are in parentheses. * denotes p < 0.1, ** denotes p < 0.05, *** denotes p < 0.01.

lower-order terms unsurprisingly show that public security is a greater concern in more violent municipalities. However, the interaction of upcoming elections with the average number of homicides over the last year, or last 3 years, does not significantly affect the salience of public security for voters.

	Low confidence in the President		Low con in state (nfidence Governor
	(1)	(2)	(3)	(4)
Upcoming local election	0.064	0.067**	-0.061	0.052
	(0.039)	(0.033)	(0.040)	(0.090)
Homicide shock	-0.011	-0.007	-0.051**	-0.025
	(0.014)	(0.012)	(0.026)	(0.020)
Upcoming local election	0.046	0.015	-0.005	-0.085
\times Homicide shock	(0.044)	(0.041)	(0.062)	(0.158)
Observations	12,352	12,352	5,912	5,912
Outcome range	$\{0,1\}$	$\{0,1\}$	$\{0,1\}$	{0,1}
Outcome mean	0.26	0.26	0.32	0.32
Upcoming local election mean	0.17	0.17	0.02	0.02
Homicide shock mean	0.44	0.44	0.42	0.42
Survey years without data			2003, 2005	2003, 2005
State fixed effects		\checkmark		\checkmark

 Table 7: Heterogeneous effects of upcoming local elections on institutional confidence at higher

 levels of government, by short-run homicide shocks

Notes: All specifications include survey year fixed effects, and are estimated using OLS. Standard errors clustered by municipality are in parentheses. * denotes p < 0.1, ** denotes p < 0.05, *** denotes p < 0.01.

Turning to beliefs about institutional competence, panel B shows that homicides coinciding with upcoming local elections also cause voters to reduce their confidence in the municipal incumbent. Columns (1) and (2) demonstrate that local elections only increase the likelihood that a respondent expresses low confidence in the incumbent mayor when a homicide shock occurs before a local election. The 11 percentage point reduction in mayoral confidence implied by column (1) represents a 34% drop relative to the sample mean. The coefficients for local elections in columns (1) and (2) suggest a somewhat asymmetric relationship where voters do not significantly update about their incumbents when a non-positive homicide shock occurs. Again, long-run homicide trends do not differentially affect voters around elections.

Finally, reinforcing the precinct-level voting behavior in Table 4, Table 7 shows that homicide shocks around elections do not cause voters to lose confidence in higher levels of government. Columns (1)-(4) show no significant change in confidence in either the President or state Governor. In conjunction with the findings for municipal incumbent parties, these tests indicate that voters indeed assign responsibility for local homicides to municipal incumbent parties perceived to be responsible for local crime.



Figure 6: The effect of a homicide shock on the change in incumbent party vote share, by previous homicide shocks and incumbent identity (95% confidence intervals)

Notes: Same incumbent is an indicator for the current incumbent party winning the previous election. Previous shock is an indicator for a homicide shock occurring before the previous election.

The preceding evidence suggests that homicide both increase the saliency of insecurity (H2a) and induce voters to update about their municipal incumbent (H2b). However, this analysis cannot discern which effect actually induces voters to vote against the incumbent party. I next show that electoral sanctioning at least partially reflects voters updating relative to their prior beliefs.

7.1.2 The position of prior beliefs conditions voter responses to homicide shocks

Voter responses to pre-election homicide shocks may also depend upon the position of their prior beliefs. Where elevated homicide levels are expected, a homicide shock may not cause voters to substantially update their beliefs and behavior. To test this feature of voter learning (H7), I explore whether previous homicide shocks moderate precinct-level electoral sanctioning. Figure 6 shows how the effect of a homicide shock on the change in the incumbent party's vote share varies with the occurrence of a homicide shock before the previous election and whether the current incumbent party differs from the previous incumbent party.⁶⁶

The results indicate that the sanctioning of homicide reflects updating vis-à-vis prior experiences. The first bar shows that sanctioning is greatest among voters in municipalities experiencing a homicide shock before the current election and which did not experience such a shock before the previous election under a different incumbent party. This effect is larger in magnitude than for the second bar where the current incumbent was in office at the previous election but did not

⁶⁶The regression estimates are provided in Table A13.

experience a shock. Similarly, the negative effect of a homicide shock is also large if the same incumbent party was in office for consecutive pre-election homicide shocks. Conversely, voters do not punish current incumbents if a previous incumbent from a different party also presided over a homicide shock. In addition to the likely priming effects of crime in the news, these results exhibit voter learning across time from different local violence reports.

7.1.3 Voters with imprecise prior beliefs respond most to homicide shocks

Another prediction of Bayesian belief updating is that voters with relatively imprecise prior beliefs will update most from incumbent party performance signals (H6). Education is widely believed to increase political knowledge by providing the ability, opportunity and motivation to acquire information (Barabas et al. 2014; Delli Carpini and Keeter 1996).⁶⁷ Using higher education as a proxy for the precision of prior beliefs, I next show—at both the individual and aggregate levels—that the least educated voters are most likely to respond to homicide shocks.

To test whether voters with weaker priors respond more to performance indicators in the news around elections, I first use equation (6) to examine how the effects of short-run homicide shocks on voter beliefs vary by education. Using the ENCUP survey data, I define an education indicator for the 18% of respondents that have completed a university degree ("higher") education. This is positively correlated with news consumption and political knowledge. The marginal effects of an upcoming local election reported in Table 8 suggest that increased concern about public insecurity and reduced confidence in municipal incumbents are most pronounced among less educated voters.⁶⁸ Although the triple interaction is not quite statistically significant in the case of public security concerns, the interaction between upcoming local elections and homicide shocks indicate that—netting across coefficients—a pre-election homicide shock only significantly increases the concerns of voters that did not complete upper secondary schooling.

I link these individual-level findings to precinct-level voting outcomes by examining whether the effect of local media covering a homicide shock also varies with the education. To complement the individual-level analysis, I define an indicator for precincts in the top quartile of the higher education distribution (39.5% or more of voters with higher education in the relatively urban neighbors sample). Consistent with the survey results in columns (1)-(4), the triple interaction in column (5) of Table 8 shows that electoral sanctioning induced by local media is almost entirely driven by precincts lacking highly educated voters. Contrary to the potential confound that the least ed-

⁶⁷Larreguy, Marshall and Snyder (2016) show that education is positively correlated with political knowledge and holding opinions on different political parties in Mexico.

⁶⁸The homicide within the last month measure produced similar but less stark results.

	Public insecurity the major problem in		Low con the munic	Low confidence in the municipal mayor		incumbent ote share
	the con	nmunity				
	(1)	(2)	(3)	(4)	(5)	(6)
Upcoming local election	-0.021	-0.048*	-0.036	0.163		
	(0.022)	(0.025)	(0.049)	(0.118)		
Homicide shock	-0.006	-0.005	-0.043*	-0.013	0.0094	0.0111
	(0.015)	(0.013)	(0.026)	(0.022)	(0.0289)	(0.0313)
Higher education	0.082***	0.064***	-0.001	0.008	0.0367**	0.0198
	(0.021)	(0.021)	(0.029)	(0.024)	(0.0151)	(0.0137)
Upcoming local election	0.107***	0.096***	0.143**	0.095		
\times Homicide shock	(0.037)	(0.034)	(0.063)	(0.149)		
Upcoming local election	0.037	0.045	0.196***	0.182***		
\times Higher education	(0.033)	(0.034)	(0.051)	(0.058)		
Homicide shock \times Higher education	0.000	0.010	-0.035	-0.037	-0.0591**	-0.0494*
C	(0.026)	(0.024)	(0.040)	(0.035)	(0.0283)	(0.0257)
Upcoming local election	-0.057	-0.073	-0.494***	-0.487***	. ,	. ,
\times Homicide shock \times Higher education	(0.054)	(0.053)	(0.070)	(0.075)		
Local media		. ,	. ,	. ,	0.0027**	0.0031***
					(0.0011)	(0.0012)
Homicide shock \times Local media					-0.0036***	-0.0035**
					(0.0013)	(0.0015)
Local media \times Higher education					-0.0019**	-0.0013**
-					(0.0008)	(0.0006)
Homicide shock \times Local media					0.0028*	0.0030**
\times Higher education					(0.0016)	(0.0013)
Observations	6,564	6,564	5,916	5,916	30,099	30,099
Outcome range	$\{0,1\}$	$\{0,1\}$	$\{0,1\}$	$\{0,1\}$	[-0.63,0.50]	[-0.63,0.50]
Outcome mean	0.13	0.13	0.34	0.34	-0.03	-0.03
Homicide measure mean	0.47	0.47	0.42	0.42	0.40	0.40
Higher education measure mean	0.20	0.20	0.16	0.16	0.53	0.53
Upcoming local election mean	0.25	0.25	0.02	0.02		
Local media mean					18.40	18.40
Survey years without data	2003, 2005	2003, 2005	2003, 2005	2003, 2005		
State fixed effects		\checkmark		\checkmark		
Interactive control for development						\checkmark

Table 8: Effect of upcoming local elections on concern about public insecurity and low confidence in incumbent mayors, by homicide shock and completion of higher education

Notes: The specifications in columns (1)-(4) are estimated using the ENCUP data and include survey year fixed effects, and are estimated using OLS; the number of observations differ between columns due to non-responses. The specifications in columns (5) and (6) are estimated using precinct-level electoral returns and includes neighbor group and year fixed effects, and are estimated using OLS; all observations are weighted by the number of registered voters divided by the number of comparison units within each neighbor group. Standard errors clustered by municipality are in parentheses. * denotes p < 0.1, ** denotes p < 0.05, *** denotes p < 0.01.

ucated respond most to homicide shocks in the news because local violence afflicts such voters most heavily (Díaz-Cayeros et al. 2011), the higher education triple interaction becomes stronger in column (6) once a triple interaction with an index of precinct-level poverty is simultaneously

controlled for.69

7.2 Do political information cycles reflect changes in demand for or supply of politically-relevant news?

This article has examined political information cycles as an equilibrium outcome. However, from both a theoretical and policy perspective, it is important to understand whether these cycles reflect changes in voter demand for politically-relevant news or changes in the media's supply of such news around elections. The increased political news consumption at the extensive margin documented in Table 1, together with the fact that the least educated (who are also most likely to increase consumption at the extensive margin) respond most to homicide shocks in the news, suggest that tuning in for the news is a conscious choice, given that radio and television schedules do not change around elections. However, for the many voters increasing consumption along the intensive margin, it is hard to separate increased demand from news providers broadcasting more politically-relevant news. Such voters could be primarily responsible for the large electoral effects.

For news relating to local homicides, a first step in separating these effects is to examine changes in coverage around elections. In the absence of comprehensive broadcast media transcripts, I use data from Osorio's (2015). He uses machine-learning techniques to collate reports relating to drug violence from 105 government agencies and national and local newspapers between 2000 and 2010. Broadcast media coverage is likely to be similar given that these sources serve as an important local information source for broadcast media outlets. Based on a survey of local newspapers and radio stations, 64% report sourcing news from other local media outlets.⁷⁰ To examine the extent to which reporting of violent events increases before elections, I test whether the association between the occurrence and reporting of violent events increases within the five months before local elections using the following difference-in-differences specification:

$$Reports_{mt} = \beta_1 Homicides_{mt} + \beta_2 Upcoming \ local \ election_{mt} + \beta_3 \left(Upcoming \ local \ election_{mt} \times Homicides_{mt} \right) + \eta_m + \mu_t + \varepsilon_{imt},$$
(7)

where $Reports_{mt}$ is a monthly municipal-level count of the number of violent events (including homicides as well as shootings, kidnappings, torture etc.) between DTOs reported for that month,⁷¹

⁶⁹This measure is a standardized scale composed of 2010 Census indicators of socioeconomic development.

⁷⁰Broadcast media reporting on the basis of newspaper reports is common in developing countries (e.g. Keefer and Khemani 2016).

⁷¹The outcome was aggregated up from Osorio's (2015) weekly count. Similar results obtain when using reports

and μ_m and μ_t represent municipality and month-year fixed effects.

	Reports of inter-DTO violence				
	(1)	(2)	(3)	(4)	(5)
Homicides	0.234***	0.224***	0.231***	0.234***	0.224***
	(0.025)	(0.023)	(0.025)	(0.025)	(0.023)
Upcoming local election			-0.015*	-0.036	-0.201***
			(0.009)	(0.049)	(0.072)
Homicides			0.011	0.011	0.014
\times Upcoming local election			(0.015)	(0.015)	(0.012)
Observations	308,360	308,360	308,360	308,360	308,360
Outcome range	[0,102]	[0,102]	[0,102]	[0,102]	[0,102]
Outcome mean	0.15	0.15	0.15	0.15	0.15
Outcome std. dev.	1.43	1.43	1.43	1.43	1.43
Homicides mean	0.44	0.44	0.44	0.44	0.44
Homicides mean	3.33	3.33	3.33	3.33	3.33
Upcoming local election mean	0.15	0.15	0.15	0.15	0.15
Additional fixed effects		State-		Election-	Election-state-
		month-year		month-year	month-year

Table 9: Association between monthly municipal homicide counts and government agency and newspaper reports on violence between gangs, by upcoming local elections

Notes: All specifications include municipality and year-month fixed effects, and are estimated using OLS. Standard errors clustered by municipality are in parentheses. * denotes p < 0.1, ** denotes p < 0.05, *** denotes p < 0.01.

The results in Table 9 indicate that media coverage of homicides is high, but show that such coverage is not significantly greater before elections. Column (1) first demonstrates that for each homicide that occurs in a municipality, the number of violent events reported increases by 0.23. Since not all newspapers are studied and reports only pertain to drug-related violence, this estimate almost certainly substantially understates the propensity of newspapers to report homicides. Nevertheless, the large positive effect clearly suggests a strong association. Turning to differential coverage around elections, column (3) shows that reporting is not significantly greater in the five months preceding local elections.⁷² Both sets of results are precisely estimated and robust to including state-month-year fixed effects (and their interaction with upcoming local elections).

These findings indicate that media coverage of homicides is not sensitive to the electoral cycle, and thus suggest that the sanctioning of pre-election homicides is more likely driven by increased

of violent enforcement, arrests, drug seizures, asset seizures, and gun seizures.

⁷²Unreported estimates show similar effects when focusing on the months immediately before the election.

voter demand for politically-relevant news before elections than changes in the supply of such news. Nevertheless, to the extent that the tone of the news may change around elections, this conclusion should be treated with caution.

7.3 Interpreting voter behavior

The preceding analysis suggests that voters generally respond to homicide shocks in a Bayesian manner, based on weak priors. In contrast, the evidence is inconsistent with recency bias driving the results: instead of discarding or discounting information that they consumed in advance of the election, voters never update from homicide shocks outside of electoral campaigns, even in their immediate aftermath. This does not imply that information received before an election is not quickly forgotten, but does indicate that the timing of information consumption is precondition for recency bias to occur.

Given that voters are updating their beliefs following a pre-election homicide shock, what exactly are they updating about? One possibility is that voters update their beliefs from observing the incumbent party's response, rewarding those revealed as good types and sanctioning those revealed to be bad types (Ashworth and Bueno de Mesquita 2014*b*; Cole, Healy and Werker 2012). However, given that sanctioning is driven by the least educated voters, who are least likely to be able to extract such a sophisticated signal (e.g. Alt, Lassen and Marshall 2016), this seems relatively unlikely. Moreover, contrary to an explanation where the incumbent or challenger parties used their radio and television ads to exploit the issue, unreported results show that electoral sanctioning of homicide shocks does not vary with the proportion of incumbent ads that a precinct receives.

Rather, the most compelling interpretation is that poorly-informed voters regard a homicide shock reported in the news as a signal of incumbent party incompetence. This is consistent with the evidence of voter learning among the least informed and differentiating across levels of government and municipalities with and without their own police force. More sophisticated voters neglect such signals. However, although voters behave as Bayesians with weak prior beliefs along multiple dimensions, the strength of their response appears inconsistent with its weakness as an indicator of longer-run performance. This suggests that the least informed voters may not be fully Bayesian in that they exhibit some degree of overconfidence, such that they systematically overes-timate the precision of the signal (e.g. Ortoleva and Snowberg 2015). Although this is a common phenomenon in behavioral studies, further work is required to precisely disentangle the cognitive source of voter updating.

8 Conclusion

This study demonstrates the importance of the timing of voter news consumption for understanding how voters hold governments to account. Since many voters follow relatively sharp political information cycles, whereby most politically-relevant information is consumed prior to elections, voting behavior can be highly responsive to salient performance indicators, like local homicides, in the news before elections. Leveraging a wealth of fine-grained individual- and precinct-level data, and identification strategies isolating three complementary components of the theory, I show that voters indeed consume more news before elections, and that pre-election homicide shocks substantially reduce support for the municipal incumbent party among poorly informed voters exposed to local media. Conversely, there is little evidence that long-run performance homicide indicators affect electoral outcomes.

While documenting responses to pre-election information, extant studies do not link the effects to the timing of voter news consumption. Nevertheless, they suggest that political information cycles may play a central role in explaining why voters often hold governments accountable for salient short-term indicators of performance, other than violence, in a wide variety of contexts. News revealed about incumbent performance before elections on other valence issues may similarly impact voting behavior. Ferraz and Finan (2008) and Larreguy, Marshall and Snyder (2017) show respectively that Brazilian and Mexican municipal incumbent parties are punished electorally if audits reports reveal malfeasant behavior in office in the run-up to elections, but only in the presence of local media stations. Brollo (2009) finds that such sanctioning in Brazil decreases with months until the election, although she attributes this to greater scope for incumbents to respond to such reports. Furthermore, Italian deputies were only punished for alleged criminal wrongdoing following the "Clean Hands" investigation, which was widely reported in the news before the 1992 and 1994 elections (Chang, Golden and Hill 2010). Economic outcomes paint a similar picture in developed and developing contexts, where voters respond to pre-election economic indicators (e.g. Achen and Bartels 2004*b*; Healy and Lenz 2014; Roberts and Wibbels 1999).

Consequently, understanding the origins of political information cycles demands further work probing the forces underlying equilibrium information consumption. In particular, voter demand for information remains poorly understood, especially in developing contexts where the marginal effects of information on electoral behavior are often substantial. Similarly, little is known empirically about when and how media stations report different types of news (although see Gentzkow and Shapiro 2010; Lawson 2002), especially with respect to the content and tone of media coverage. This paper focused primarily on identifying equilibrium cycles and estimating their electoral

implications, but these fundamental questions require attention.

Voting on the basis of the news before elections may be normatively problematic for democracy. In contrast with models emphasizing the value of information for improving democratic representation (e.g. Besley and Burgess 2002; Besley and Prat 2006) or suggesting that the aggregation of individual information can yield efficient outcomes (Feddersen and Pesendorfer 1997), the case of punishing exogenous homicide shocks prior to elections-which are uncorrelated with longer-run homicide trends-suggests that voters could be committing attribution errors by removing competent incumbents on the basis of noisy short-term performance indicators (see also Achen and Bartels 2004b). Future research should illuminate whether voters indeed elect worse incumbents on the basis of such information, which is itself likely to depend upon how voters view incumbents on average (Ashworth and Bueno de Mesquita 2014b), and whether sanctioning varies with incumbent traits that were previously unobserved by voters. Furthermore, my findings also provide clear incentives for politicians to release bad news when voter engagement is low (e.g. Durante and Zhuravskaya forthcoming; Eisensee and Strömberg 2007). Nevertheless, the political information cycles documented here thus add to an increasing body of work challenging the orthodoxy that an informed electorate necessarily enhances electoral accountability (see Ashworth and Bueno de Mesquita 2014*a*).

Regardless of the welfare implications, it is clear that political parties, interest groups, and NGOs should strategically target incumbent performance information toward the least informed voters around elections—when voters consume political news. While such groups may use information to support their goals, it is essential that purely informative news should relate to horizons beyond proximate security or economic shocks, and thus provide the longer-term indicators that voters profess to value (Healy and Lenz 2014). Although NGO-supported interventions have attempted to achieve this, often by delivering leaflets or newspapers (e.g. Arias et al. 2017; Banerjee et al. 2011), broadcast media appears to be the most effective tool for dissemination (Larreguy, Marshall and Snyder 2017). Consequently, a key challenge is encouraging political actors and media stations to overcome sensationalist and political biases in favor of longer-term performance indicators.

References

- Achen, Christopher H. and Larry M. Bartels. 2004*a*. "Blind retrospection: Electoral responses to drought, flu, and shark attacks." Working paper.
- Achen, Christopher H. and Larry M. Bartels. 2004*b*. "Musical chairs: Pocketbook voting and the limits of democratic accountability." Working paper.
- Alesina, Alberto, Salvatore Piccolo and Paolo Pinotti. 2016. "Organized Crime, Violence, and Politics." Working paper.
- Alt, James E., David D. Lassen and John Marshall. 2016. "Information sources, belief updating, and the politics of economic expectations: Evidence from a Danish survey experiment." *Journal of Politics* 78(2):327–343.
- Alvarez, Michael E. 2001. Information and Elections. Ann Arbor: University of Michigan Press.
- Anderson, Cameron D. 2006. "Economic voting and multilevel governance: a comparative individual-level analysis." *American Journal of Political Science* 50(2):449–463.
- Anderson, Christopher J. 2007. "The end of economic voting? Contingency dilemmas and the limits of democratic accountability." *Annual Review Political Science* 10:271–296.
- Ansolabehere, Stephen, Erik C. Snowberg and Jr. Snyder, James M. 2006. "Television and the incumbency advantage in US elections." *Legislative Studies Quarterly* 31(4):469–490.
- Arias, Eric, Horacio A. Larreguy, John Marshall and Pablo Querubín. 2017. "Priors rule: When do malfeasance revelations help and hurt incumbent parties?" Working paper.
- Arnold, R. Douglas and Nicholas Carnes. 2012. "Holding mayors accountable: New York's executives from Koch to Bloomberg." *American Journal of Political Science* 56(4):949–963.
- Ashworth, Scott. 2012. "Electoral accountability: recent theoretical and empirical work." *Annual Review of Political Science* 15:183–201.
- Ashworth, Scott and Ethan Bueno de Mesquita. 2014*a*. "Is Voter Competence Good for Voters? Information, Rationality, and Democratic Performance." *American Political Science Review* 108(3):565–587.

- Ashworth, Scott and Ethan Bueno de Mesquita. 2014b. "Learning About Voter Rationality." Working paper.
- Baker, Andy, Barry Ames and Lucio R. Renno. 2006. "Social context and campaign volatility in new democracies: networks and neighborhoods in Brazil's 2002 Elections." *American Journal of Political Science* 50(2):382–399.
- Banerjee, Abhijit V., Selvan Kumar, Rohini Pande and Felix Su. 2011. "Do Informed Voters Make Better Choices? Experimental Evidence from Urban India." Working paper.
- Barabas, Jason and Jennifer Jerit. 2009. "Estimating the Causal Effects of Media Coverage on Policy-Specific Knowledge." *American Journal of Political Science* 53(1):73–89.
- Barabas, Jason, Jennifer Jerit, William Pollock and Carlisle Rainey. 2014. "The Question (s) of Political Knowledge." *American Political Science Review* 108(4):840–855.
- Bartels, Larry M. 2008. Unequal democracy: The political economy of the New Gilded Age. Princeton, NJ: Princeton University Press.
- Bateson, Regina. 2012. "Crime victimization and political participation." *American Political Science Review* 106(3):570–587.
- Baum, Matthew A. 2002. "Sex, lies, and war: How soft news brings foreign policy to the inattentive public." *American Political Science Review* 96(1):91–109.
- Bellows, John and Edward Miguel. 2009. "War and local collective action in Sierra Leone." *Journal of Public Economics* 93(11):1144–1157.
- Berrebi, Claude and Esteban F. Klor. 2006. "On terrorism and electoral outcomes theory and evidence from the Israeli-Palestinian conflict." *Journal of Conflict Resolution* 50(6):899–925.
- Besley, Timothy and Andrea Prat. 2006. "Handcuffs for the grabbing hand? Media capture and government accountability." *American Economic Review* 96(3):720–736.
- Besley, Timothy and Robin Burgess. 2002. "The political economy of government responsiveness: Theory and evidence from India." *Quarterly Journal of Economics* 117(4):1415–1451.
- Blanco, Luisa R. 2013. "The impact of crime on trust in institutions in Mexico." *European Journal* of Political Economy 32:38–55.

- Brader, Ted and Joshua Tucker. 2012. "Follow the Leader: Party Cues, Policy Opinion, and the Power of Partisanship in Three Multiparty Systems." *Comparative Politics* 44(4):403–420.
- Brollo, Fernanda. 2009. "Who Is Punishing Corrupt Politicians—Voters or the Central Government? Evidence from the Brazilian Anti-Corruption Program." Working paper.
- Casey, Katherine. 2015. "Crossing Party Lines: The Effects of Information on Redistributive Politics." *American Economic Review* 105(8):2410–2428.
- Castañeda Sabido, Fernando. 2011. "Investigación sobre la percepción ciudadana acerca de la transparencia, rendición de cuentas y fiscalizacin en el uso de los recursos públicos en México." *Reporte para la Auditoría Superior de la Federación*.
- Chandra, Kanchan. 2007. *Why ethnic parties succeed: Patronage and ethnic head counts in India*. Cambridge University Press.
- Chang, Eric C.C., Miriam A. Golden and Seth J. Hill. 2010. "Legislative malfeasance and political accountability." *World Politics* 62(2):177–220.
- Chiang, Chun-Fang and Brian Knight. 2011. "Media bias and influence: Evidence from newspaper endorsements." *Review of Economic Studies* 78(3):795–820.
- Chong, Alberto, Ana De La O, Dean Karlan and Leonard Wantchekon. 2015. "Does Corruption Information Inspire the Fight or Quash the Hope? A Field Experiment in Mexico on Voter Turnout, Choice and Party Identification." *Journal of Politics* 77(1):55–71.
- Cole, Shawn, Andrew Healy and Eric Werker. 2012. "Do voters demand responsive governments? Evidence from Indian disaster relief." *Journal of Development Economics* 97(2):167–181.
- Cornelius, Wayne A. 1996. *Mexican politics in transition: The breakdown of a one-party-dominant regime*. La Jolla: Center for US-Mexican Studies, University of California, San Diego.
- Cummins, Jeff. 2009. "Issue Voting and Crime in Gubernatorial Elections." *Social Science Quarterly* 90(3):632–651.
- Da Silveira, Bernardo S. and João M.P. De Mello. 2011. "Campaign advertising and election outcomes: Quasi-natural experiment evidence from gubernatorial elections in Brazil." *Review of Economic Studies* 78(2):590–612.

- de Figueiredo, Miguel F.P., F. Daniel Hidalgo and Yuri Kasahara. 2013. "When Do Voters Punish Corrupt Politicians? Experimental Evidence from Brazil." Working paper.
- Dell, Melissa. 2015. "Trafficking Networks and the Mexican Drug War." American Economic Review 105(6):1738–1779.
- DellaVigna, Stefano and Ethan Kaplan. 2007. "The Fox News effect: Media bias and voting." *Quarterly Journal of Economics* 122(3):1187–1234.
- Delli Carpini, Michael X. and Scott Keeter. 1996. What Americans Know about Politics and Why It Matters.
- Díaz-Cayeros, Alberto, Beatriz Magaloni, Aila Matanock and Vidal Romero. 2011. "Living in fear: Mapping the social embeddedness of drug gangs and violence in Mexico." Working paper.
- Diaz-Cayeros, Alberto, José Antonio González and Fernando Rojas. 2006. Mexico's Decentralization at a Crossroads. In *Federal and Economic Reform: International Perspectives*, ed. Jessica S. Wallack and T. N. Srinivasan. Cambridge University Press pp. 364—406.
- Downs, Anthony. 1957. An Economic Theory of Democracy. Addison Wesley.
- Druckman, James D. 2005. "Media Matter: How Newspapers and Television News Cover Campaigns and Influence Voters." *Political Communication* 22(4):463–481.
- Durante, Ruben and Ekaterina Zhuravskaya. forthcoming. "Attack When the World Is Not Watching? U.S. News and the Israeli-Palestinian Conflict." *Journal of Political Economy*.
- Durante, Ruben and Emilio Gutierrez. 2015. "Fighting crime with a little help from my friends: Party affiliation, inter-jurisdictional cooperation and crime in Mexico." Working paper.
- Eifert, Benn, Edward Miguel and Daniel N. Posner. 2010. "Political Competition and Ethnic Identification in Africa." *American Journal of Political Science* 54(2):494–510.
- Eisensee, Thomas and David Strömberg. 2007. "News droughts, news floods, and US disaster relief." *Quarterly Journal of Economics* pp. 693–728.
- Enikolopov, Ruben, Maria Petrova and Ekaterina Zhuravskaya. 2011. "Media and political persuasion: Evidence from Russia." *American Economic Review* 101(7):3253–3285.

- Fearon, James D. 1999. Electoral Accountability and the Control of Politicians: Selecting Good Types versus Sanctioning Poor Performance. In *Democracy, Accountability, and Representation*, ed. Adam Przeworski, Susan Stokes and Bernard Manin. Cambridge University Press.
- Feddersen, Timothy and Alvaro Sandroni. 2006. "A Theory of Participation in Elections." *American Economic Review* 96(4):1271–1282.
- Feddersen, Timothy and Wolfgang Pesendorfer. 1997. "Voting Behavior and Information Aggregation in Elections with Private Information." *Econometrica* 65(5):1029–1058.
- Ferejohn, John. 1986. "Incumbent performance and electoral control." Public choice 50(1):5–25.
- Ferraz, Claudio and Frederico Finan. 2008. "Exposing Corrupt Politicians: The Effects of Brazil's Publicly Released Audits on Electoral Outcomes." *Quarterly Journal of Economics* 123(2):703– 745.
- Fox, Jonathan. 1994. "The difficult transition from clientelism to citizenship: Lessons from Mexico." *World Politics* 46(2).
- García-Ponce, Omar, Leonard Wantchekon and Thomas Zeitzoff. 2014. "Are Voters too Afraid to Tackle Corruption? Survey and Experimental Evidence from Mexico." Working paper.
- Gentzkow, Matthew and Jesse M. Shapiro. 2006. "Media bias and reputation." *Journal of Political Economy* 114(2):280–316.
- Gentzkow, Matthew and Jesse M. Shapiro. 2010. "What drives media slant? Evidence from US daily newspapers." *Econometrica* 78(1):35–71.
- Gerber, Alan S., James G. Gimpel, Donald P. Green and Daron R. Shaw. 2011. "How large and long-lasting are the persuasive effects of televised campaign ads? Results from a randomized field experiment." *American Political Science Review* 105(1):135–150.
- Golden, Miriam. 2010. Puzzles of Political Corruption in Modern Advanced Democracies. In Democracy and Accountability: Globalized Political Responsibility, ed. Hideko Magara. Toyko: Fukosha pp. 184–199.
- Gomez, Brad T. and J. Matthew Wilson. 2006. "Cognitive heterogeneity and economic voting: A comparative analysis of four democratic electorates." *American Journal of Political Science* 50(1):127–145.

- Greene, Kenneth F. 2007. Why Dominant Parties Lose: Mexico's Democratization in Comparative Perspective. Cambridge University Press.
- Greene, Kenneth F. 2011. "Campaign Persuasion and Nascent Partisanship in Mexico's New Democracy." *American Journal of Political Science* 55(2):398–416.
- Hallin, Daniel. 2000. Media, political power, and democratization in Mexico. In *De-Westernizing Media Studies*, ed. James Currant and Myung-Jin Park. London: Routledge pp. 97–110.
- Hamilton, James. 2004. All the news that's fit to sell: How the market transforms information into news. Princeton University Press.
- Harding, Robin and David Stasavage. 2014. "What democracy does (and doesnt do) for basic services: School fees, school inputs, and African elections." *Journal of Politics* 76(1):229–245.
- Healy, Andrew and Gabriel S. Lenz. 2014. "Substituting the End for the Whole: Why Voters Respond Primarily to the Election-Year Economy." *American Journal of Political Science* 58(1):31–47.
- Healy, Andrew J., Neil Malhotra and Cecilia Hyunjung Mo. 2010. "Irrelevant events affect voters' evaluations of government performance." *Proceedings of the National Academy of Sciences* 107(29):12804–12809.
- Healy, Andrew and Neil Malhotra. 2013. "Retrospective voting reconsidered." *Annual Review of Political Science* 16:285–306.
- Heinle, Kimberly, Octavio Rodríguez Ferreira and David A. Shirk. 2014. "Drug Violence in Mexico: Data and Analysis Through 2013." Justice in Mexico Project Special Report.
- Hirano, Shigeo, Gabriel S. Lenz, Maksim Pinkovskiy and Jr. Snyder, James M. 2015. "Voter Learning in State Primary Elections." *American Journal of Political Science* 59(1):91–108.
- Ibope/AGB México. 2009. "Audiencias y Medios en México.".
- Kayser, Mark Andreas and Michael Peress. 2012. "Benchmarking across borders: electoral accountability and the necessity of comparison." *American Political Science Review* 106(3):661– 684.
- Keefer, Philip and Stuti Khemani. 2016. "The Government Response to Informed Citizens: New Evidence on Media Access and the Distribution of Public Health Benefits in Africa." *World Bank Economic Review* 30(2):233–267.

- Keefer, Phillip. 2007. Seeing and Believing: Political Obstacles to Better Service Delivery. In *The Politics of Service Delivery in Democracies: Better Access for the Poor*, ed. Shantayanan Devarajan and Ingrid Widlund. Stockholm: Expert Group on Development Issues pp. 42–55.
- Klašnja, Marko and Rocío Titiunik. forthcoming. "The Incumbency Curse: Weak Parties, Term Limits, and Unfulfilled Accountability." *American Political Science Review*.
- Kronick, Dorothy. 2014. "Crime and Electoral Punishment." Working paper.
- Krosnick, Jon A. and Donald R. Kinder. 1990. "Altering the foundations of support for the president through priming." *American Political Science Review* 84(2):497–512.
- Langston, Joy. 2003. "Rising from the ashes? Reorganizing and unifying the PRI's state party organizations after electoral defeat." *Comparative Political Studies* 36(3):293–318.
- Larreguy, Horacio A., John Marshall and Jr. Snyder, James M. 2016. "Leveling the playing field: How equalizing access to campaign advertising helps locally non-dominant parties in consolidating democracies." Working paper.
- Larreguy, Horacio A., John Marshall and Jr. Snyder, James M. 2017. "Publicizing malfeasance: How local media facilitates electoral sanctioning of Mayors in Mexico." NBER Working Paper 20697.
- Lawson, Chappell H. 2002. Building the fourth estate: Democratization and the rise of a free press in Mexico. University of California Press.
- Lawson, Chappell H. 2004. Television Coverage, Vote Choice, and the Campaign. In *Mexico's Pivotal Democratic Election: Candidates, Voters, and the Presidenial Campaign of 2000*, ed. Jorge I. Domínguez and Chappell H. Lawson. Johns Hopkins University Press pp. 187–211.
- Lawson, Chappell and James A. McCann. 2005. "Television News, Mexico's 2000 Elections and Media Effects in Emerging Democracies." *British Journal of Political Science* pp. 1–30.
- Lenz, Gabriel S. 2009. "Learning and opinion change, not priming: Reconsidering the priming hypothesis." *American Journal of Political Science* 53(4):821–837.
- Lenz, Gabriel S. 2012. Follow the leader? How voters respond to politicians' policies and performance. Chicago: University of Chicago Press.

- Lenz, Gabriel S. and Jonathan M. Ladd. 2009. "Exploiting a Rare Communication Shift to Document the Persuasive Power of the News Media." *American Journal of Political Science* 53(2):394–410.
- Levitt, Steven D. 1997. "Using Electoral Cycles in Police Hiring to Estimate the Effect of Police on Crime." *American Economic Review* 87(3):270–290.
- Ley, Sandra. 2014. "Citizens in Fear: Participation and Voting Behavior in the Midst of Violence." Dissertation.
- Magaloni, Beatriz. 2006. Voting for Autocracy: Hegemonic Party Survival and its Demise in Mexico. New York: Cambridge University Press.
- Marshall, John. 2016. "Signaling sophistication: How social expectations can increase political information acquisition." Working paper.
- McCann, James A. and Chappell H. Lawson. 2003. "An Electorate Adrift? Public Opinion and the Quality of Democracy in Mexico." *Latin American Research Review* 38(3):60–81.
- México Evalúa. 2012. "Seguridad y Justicia Penal en los estados." Report.
- Mullainathan, Sendhil and Andrei Shleifer. 2005. "The Market for News." *American Economic Review* 95(4):1031–1053.
- Nordhaus, William D. 1975. "The Political Business Cycle." *Review of Economic Studies* 42(2):169–190.
- Ortoleva, Pietro and Erik Snowberg. 2015. "Overconfidence in Political Behavior." *American Economic Review* 105(2):504–535.
- Osorio, Javier. 2015. "The Contagion of Drug Violence: Spatiotemporal Dynamics of the Mexican War on Drugs." *Journal of Conflict Resolution* 59(8):1403–1432.
- Pande, Rohini. 2011. "Can informed voters enforce better governance? Experiments in lowincome democracies." *Annual Review of Economics* 3(1):215–237.
- Powell, G. Bingham, Jr. and Guy D. Whitten. 1993. "A Cross-National Analysis of Economic Voting: Taking Account of the Political Context." *American Journal of Political Science* 37(2):391– 414.

- Prat, Andrea and David Strömberg. 2005. "Commercial Television and Voter Information.". CEPR Discussion Paper No. 4989.
- Prior, Markus. 2007. Post-broadcast democracy: How media choice increases inequality in political involvement and polarizes elections. Cambridge University Press.
- Reames, Benjamin. 2003. "Police forces in Mexico: A profile." Center for US-Mexican Studies.
- Remmer, Karen L. 1991. "The political impact of economic crisis in Latin America in the 1980s." *American Political Science Review* 85(3):777–800.
- Roberts, Kenneth M. and Erik Wibbels. 1999. "Party systems and electoral volatility in Latin America: a test of economic, institutional, and structural explanations." *American Political Science Review* 93(3):575–590.
- Rodden, Jonathan and Erik Wibbels. 2011. "Dual accountability and the nationalization of party competition: Evidence from four federations." *Party Politics* 17(5):629–653.
- Sabet, Daniel. 2010. Police Reform in Mexico: Advances and Persistent Obstacles. In Shared Responsibility: U.S.-Mexico Policy Options for Confronting Organized Crime, ed. Eric L. Olson, David A. Shirk and Andrew Selee. Woodrow Wilson International Center for Scholars Mexico Institute.
- Semetko, Holli A. and Patti M. Valkenburg. 2000. "Framing European politics: A content analysis of press and television news." *Journal of Communication* 50(2):93–109.
- Serra, Gilles. 2012. "The Risk of Partyarchy and Democratic Backsliding in Mexico." *Taiwan Journal of Democracy* 8(1):93–118.
- Serra, Gilles. 2014. "The 2012 elections in Mexico: Return of the dominant party." *Electoral Studies* (34):349–353.
- Singer, Matthew M. and Ryan E. Carlin. 2013. "Context counts: The election cycle, development, and the nature of economic voting." *Journal of Politics* 75(3):730–742.
- Snyder, James M., Jr. and David Strömberg. 2010. "Press Coverage and Political Accountability." *Journal of Political Economy* 118(2):355–408.
- Spenkuch, Jorg L. and David Toniatti. 2016. "Political Advertising and Election Outcomes." Working Paper.

- Trelles, Alejandro and Miguel Carreras. 2012. "Bullets and votes: Violence and electoral participation in Mexico." *Journal of Politics in Latin America* 4(2):89–123.
- UNODC. 2013. "Global Study on Homicide." Report, available here.
- Vivanco, Edgar F., Jorge Olarte, Alberto Díaz-Cayeros and Beatriz Magaloni. 2015. Drugs, Bullets, and Ballots: The Impact of Violence on the 2012 Presidential Election. In *Mexico's Evolving Democracy: A Comparative Study of the 2012 Elections*, ed. Jorge I. Domínguez, Kenneth F. Greene, Chappell H. Lawson and Alejandro Moreno. Johns Hopkins University Press pp. 153–181.

Zaller, John R. 1992. The Nature and Origins of Mass Opinion. Cambridge University Press.

A Appendix

Contents

A.1	Formal	model illustrating the voter updating process
A.2	Month	s and years of municipal elections
A.3	Data de	escription
	A.3.1	ENCUP survey data
	A.3.2	Precinct and municipal homicide and electoral data
	A.3.3	Precinct local media coverage data
A.4	Map of	f municipalities included in different samples
A.5	Additio	onal analyses
	A.5.1	Assessing the identification assumptions
	A.5.2	Drug-related homicides
	A.5.3	Difference-in-differences approach to identifying the effects of pre-election
		homicides
	A.5.4	Differential updating across elections
	A.5.5	Learning from the experiences of neighboring municipalities
	A.5.6	Differential impacts of homicide shocks across parties

A.1 Formal model illustrating the voter updating process

This section develops a simple decision-theoretic selection model (alla Fearon 1999) to clarify formally the theoretical implications discussed in the main paper. In particular, the model incorporates salience and learning considerations to examine how information about incumbent performance affects the voting behavior of forward-looking voters seeking to choose the best candidate.

By consuming information, forward-looking voters learn about the contemporary political world. Specifically, let a voter receive *n* signals about incumbent performance in office, and thus the suitability of the incumbent party for continuing in office (or "competence").⁷³ For simplicity, each signal *s* of incumbent performance is an independent draw from a Normal distribution $N(\mu, \sigma^2)$ with known variance $\sigma^2 > 0$. The mean of the distribution μ represents the unknown true level of incumbent competence, while a larger variance reflects the possibility that different media sources distort or under-report certain types of news (e.g. Besley and Prat 2006; Gentzkow and Shapiro 2006; Mullainathan and Shleifer 2005), fail to serve all segments of the market (Prat and Strömberg 2005), or are dwarfed by news that does not pertain to the voter's incumbent party (Larreguy, Marshall and Snyder 2017). The voter also possesses a prior belief $N(\delta, \tau^2)$ over incumbent competence. For simplicity the challenger's level of competence is normalized to 0, and I assume that information about the incumbent does not cause voters to also update about the challenger.⁷⁴ The voter updates their belief about the incumbent's competence downwards (upwards) when $\delta > (<)\bar{s}$, where $\bar{s} = \sum_{t=1}^{n} s_t / n$ is the average signal received by a voter.

The voter also receives a shock $b \in \mathbb{R}$ toward the incumbent. This shock b may be positive or negative, and could derive from partisanship, clientelistic ties, or other valence factors that helped the incumbent party win the last election. Ultimately, individuals vote according to both their bias and expectations of politician competence in office. To capture issue salience, the voter weights the importance of competence by the weighting function w(n) > 0, which is increasing in n.⁷⁵ For simplicity, I assume that the voter is risk-averse and simply maximizes the sum of (expected) competence and bias.⁷⁶

⁷³Politicians are assumed to perform according to their underlying competence level μ or 0 when in office.

⁷⁴The results could be easily extended to allow for voters to have a distribution of beliefs over the challenger. If I allowed beliefs about the incumbent and challenger to be correlated, then the results would be unaffected for a sufficiently small correlation.

⁷⁵Similar results would hold if the weighting also depended on the content of the news, i.e. $w(n, \bar{s})$, where most plausibly $w_{\bar{s}} < 0$ and $w_{n\bar{s}} < 0$. Changes in the weight attached to beliefs about incumbent competence could also reflect a voter's certainty about the information; in such a setting, more signals would then cause voters to increasingly voter according with the signal because it relatively increases the precision of their belief about the incumbent's competence (relative to other factors affecting vote choice).

⁷⁶Abstracting from risk-aversion presents the results particularly clear. If the voter were risk-averse, information would help the incumbent by reducing uncertainty about the utility voters would expect to receive from electing

Upon receiving n signals, a voter learns from the news about the competence of the incumbent party. Updating their prior belief distribution, the voter's posterior belief distribution is given by the following standard Normal learning result:

$$N\left(\frac{\frac{\delta}{\tau^2} + \frac{n\bar{s}}{\sigma^2}}{\frac{1}{\tau^2} + \frac{n}{\sigma^2}}, \left(\frac{1}{\tau^2} + \frac{n}{\sigma^2}\right)^{-1}\right)$$
(A1)

where \bar{s} is the mean signal received by the voter. The extent to which signals about incumbent performance, particularly when noisy or relatively uninformative (i.e. high σ^2), affect voter posterior beliefs depends upon the strength of their prior beliefs. A voter with weak priors (i.e. large τ^2) are mostly likely to update their beliefs. While a noisy signal should have limited effect on well-informed voters who recognize that a short-term shock is not a good reflection of incumbent competence, a voter with a weak prior does not know whether a short-term shock reflects long-term performance. To the extent that uninformed voters believe that such information may be informative about incumbent competence, their posterior beliefs—both the mean and variance about incumbent competence change.

The expected utility associated with the incumbent politician is then given by integrating over this posterior distribution. Combining this with the bias b toward the incumbent, a voter with bias b votes for the incumbent when:

$$B \equiv b + w(n) \frac{\delta \sigma^2 + n\bar{s}\tau^2}{\sigma^2 + n\tau^2} \ge 0$$
(A2)

Intuitively, a voter is more likely to re-elect the incumbent when the bias *b* toward the incumbent is high and the expected competence of the incumbent is high. The relative importance attached to these factors is reflected by w(n). The individual-level implications could be aggregated into party-level vote shares by integrating over the distribution of voter priors and biases.

To understand how information impacts the voters propensity of voting for the incumbent, I differentiate B with respect to the number of signals, n:

$$\frac{\partial B}{\partial n} = \underbrace{w(n) \frac{\sigma^2 \tau^2(\bar{s} - \delta)}{(\sigma^2 + n\tau^2)^2}}_{Learning \ effect} + \underbrace{w'(n) \frac{\delta \sigma^2 + n\bar{s}\tau^2}{\sigma^2 + n\tau^2}}_{Salience \ effect}.$$
(A3)

The first term represents the learning effect: how the impact of new information, by changing

them. This could easily be achieved by adopting a risk-averse utility function (for tractability a constant absolute risk-aversion functional form could easily map Normal posterior beliefs into a utility function that separates the utility associated with the mean and variance).

the voter's posterior belief, affects their likelihood of supporting the incumbent. The following proposition analyzes the comparative static implications of the learning effect:

Proposition 1. If $\bar{s} < \delta$, such that information does not help the incumbent, then the learning effect impacts voting behavior as follows:

- 1. The likelihood that the voter votes for the incumbent decreases with the number of signals received (n).
- 2. When $\frac{1}{\tau^2} < \frac{n}{\sigma^2}$, the learning increases in the precision of the signal $(\frac{1}{\sigma^2})$, decreases in the precision of the prior $(\frac{1}{\tau^2})$, and (if $\sigma^2 n\tau^2 < \sqrt{2n\tau^2\sigma^2}$) increases in their interaction.

The reverse results hold when $\overline{s} > \delta$.

Proof: Focusing on the learning effect sets the salience effect to zero. For part 1, given w(n) > 0 and all variances are positive, it is easy to see that the learning effect, $L \equiv w(n) \frac{\sigma^2 \tau^2(\bar{s} - \delta)}{(\sigma^2 + n\tau^2)^2}$, is positive when $\bar{s} < \delta$. Part 2 is established by taking subsequent derivatives. The cross-partial (i.e. the partial of the learning effect) with respect to σ^2 is given by:

$$\frac{\partial L}{\partial \sigma^2} = \frac{w(n)(\overline{s} - \delta)\tau^2[n\tau^2 - \sigma^2]}{(\sigma^2 + n\tau^2)^3} > 0, \tag{A4}$$

which, given $\overline{s} < \delta$, is negative when $n\tau^2 < \sigma^2$. Similarly,

$$\frac{\partial L}{\partial \tau^2} = -\frac{w(n)(\bar{s} - \delta)\sigma^2[n\tau^2 - \sigma^2]}{(\sigma^2 + n\tau^2)^3} < 0, \tag{A5}$$

by the same conditions. Finally,

$$\frac{\partial^2 L}{\partial \tau^2 \partial \sigma^2} = -\frac{w(n)(\bar{s} - \delta)\sigma^2 [2n\tau^2 \sigma^2 - (\sigma^2 - (n\tau^2)^2)]}{(\sigma^2 + n\tau^2)^3} > 0, \tag{A6}$$

provided that $\sigma^2 - n\tau^2 < \sqrt{2n\tau^2\sigma^2}$. The multiplicative nature of the comparative statics ensures that the reverse results hold when $\bar{s} > \delta$.

When $\bar{s} < (>)\delta$, the voter updates negatively (positively) about the competence of their incumbent. Consider the case where $\bar{s} < \delta$. The average signal of incumbent performance that they receive from the news suggests that the incumbent's true level of competence is below their prior δ . Consequently, part 1 unsurprisingly states that when the news suggests that incumbent performance is low, a voter is less likely to re-elect the incumbent. This reflects the decrease in the posterior belief over the competence of the incumbent, while the posterior belief about the challenger is unaffected. Part 2 further shows that the punishment of the incumbent is increasing in the precision of the signals and decreasing in the precision of the prior—provided that the prior is sufficiently weak ($\frac{1}{\tau^2} < \frac{n}{\sigma^2}$). Furthermore, under an additional condition, the impact of a precise signal is strongest when voter priors are weakest. Intuitively, when news coverage is particularly credible or informative relative to their prior beliefs, voters are less likely to believe that the incumbent has sufficient competence to merit re-election.

The second term in equation (A3) represents the salience effect: the impact of information increasing the relative importance of incumbent performance to the voter. This increases support for the incumbent if the posterior belief is positive (and thus exceeds the normalized belief about the challenger's competence). This opens the possibility that even if voters receive a negative signal, such that $\bar{s} < \delta$, the benefits of priming an issue on which the incumbent initially scored well could outweigh the negative learning effect. The following result clarifies this insight, and describes several key comparative static predictions in terms of vote choice:

Proposition 2. If $\bar{s} < \delta$, such that information does not help the incumbent, then the salience effect impacts voting behavior as follows:

- 1. The likelihood that the voter votes for the incumbent increases (decreases) with the number of signals received (n) when the posterior belief about incumbent competence, $\frac{\delta\sigma^2 + n\bar{s}\tau^2}{\sigma^2 + n\tau^2}$, is positive (negative).
- 2. A negative salience effect increases in the precision of the signal $(\frac{1}{\sigma^2})$, decreases in the precision of the prior $(\frac{1}{\tau^2})$, and $(if \frac{1}{\tau^2} < \frac{n}{\sigma^2})$ decreases in their interaction.

Proof: Focusing on the salience effect sets the learning effect to zero. For part 1, given w'(n) > 0 and all variances are positive, it is easy to see that the salience effect, $S \equiv w'(n) \frac{\delta \sigma^2 + n\bar{s}\tau^2}{\sigma^2 + n\tau^2}$, is positive when the posterior belief $\frac{\delta \sigma^2 + n\bar{s}\tau^2}{\sigma^2 + n\tau^2} > 0$ (or when $\delta \sigma^2 + n\bar{s}\tau^2 > 0$, given the denominator is positive). Part 2 is again established by taking subsequent derivatives. The cross-partial (i.e. the partial of the salience effect) with respect to σ^2 is given by:

$$\frac{\partial S}{\partial \sigma^2} = \frac{w'(n)n\tau^2[\delta - \bar{s}]}{(\sigma^2 + n\tau^2)^2} > 0.$$
(A7)

Similarly,

$$\frac{\partial S}{\partial \tau^2} = -\frac{w'(n)n\sigma^2[\delta-\bar{s}]}{(\sigma^2 + n\tau^2)^2} < 0.$$
(A8)

Finally,

$$\frac{\partial^2 S}{\partial \tau^2 \partial \sigma^2} = \frac{w'(n)n[\delta - \bar{s}](\sigma^2 - n\tau^2)}{(\sigma^2 + n\tau^2)^3} < 0, \tag{A9}$$

provided that $\sigma^2 - n\tau^2 < 0$. The multiplicative nature of the comparative statics again ensures that the reverse results hold when $\bar{s} > \delta$.

The first part of the proposition reiterates that the direction of the salience effect depends upon the voter's posterior belief—namely if it is above that of the challenger, i.e. $\frac{\delta\sigma^2 + n\bar{s}\tau^2}{\sigma^2 + n\tau^2} > 0$. The magnitude of the negative impact of the salience effect is increasing in the precision of the signal $(1/\sigma^2)$ and increasing in the weakness of the voter's prior $(1/\tau^2)$. The interaction shows that such results trade-off.

Finally, I consider when the salience effect overpowers the learning effect. In particular,

Proposition 3. If $\bar{s} < \delta$, the salience effect compounds the negative learning effect when the posterior belief about incumbent competence, $\frac{\delta\sigma^2 + n\bar{s}\tau^2}{\sigma^2 + n\tau^2}$, is positive (negative). The learning effect dominates a confounding salience effect when $\frac{w'(n)}{w(n)} < \frac{\sigma^2\tau^2(\delta-\bar{s})}{(\delta\sigma^2 + n\bar{s}\tau^2)(\sigma^2 + n\tau^2)}$

Proof: The results follow simply from part 1 of Proposition 2, and setting L < S.

This result unsurprisingly shows that when the salience effect is negative, it compounds the negative learning effect—and that this again requires that the posterior belief about the incumbent is below that of the challenger. Even if this does not hold, the second element of the proposition notes new information may still have a negative effect if the learning effect dominates the salience effect, which requires that the relative increase in salience is sufficiently small or the signal significantly departs from the prior.

A.2 Months and years of municipal elections

Table A1 reports the municipal elections potentially covered by the survey and aggregate elections in the main analysis.

A.3 Data description

A.3.1 ENCUP survey data

Upcoming local election. Indicator coded 1 for respondents living in a state/municipality with an upcoming local election occurring within the year of the survey. States/municipalities where an election has already occurred within the year of the survey are coded 0.

State	Election dates
Aguascalientes	August 2001, August 2004, August 2007, July 2010, July 2013.
Baja California	June 2001, August 2004, August 2007, July 2010, July 2013.
Baja California Sur	February 1999, February 2002, February 2005, February 2008, February 2011.
Campeche	July 2000, July 2003, July 2006, July 2009, July 2012.
Chiapas	October 2001, October 2004, October 2007, July 2010, July 2012.
Chihuahua	July 2001, July 2004, July 2007, July 2010, July 2013.
Coahuila	September 1999, September 2002, September 2005, October 2008, July 2010, July 2013.
Colima	July 2000, July 2003, July 2006, July 2009, July 2012.
Distrito Federal	July 2000, July 2003, July 2006, July 2009, July 2012.
Durango	July 2001, July 2004, July 2007, July 2010, July 2013.
Guanajuato	July 2000, July 2003, July 2006, July 2009, July 2012.
Guerrero	October 1999, October 2002, October 2005, October 2008, July 2012.
Hidalgo	November 1999, November 2002, November 2005, November 2008, July 2011.
Jalisco	November 2000, July 2003, July 2006, July 2009, July 2012.
Estado de México	July 2000, March 2003, March 2006, July 2009, July 2012.
Michoacán	November 2001, November 2004, October 2007, October 2011.
Morelos	July 2000, July 2003 , July 2006, July 2009, July 2012.
Nayarit	July 1999, July 2002, July 2005, July 2008, July 2011.
Nuevo León	July 2000, July 2003, July 2006, July 2009, July 2012.
Oaxaca	August 2001, August 2004, August 2007, July 2010, July 2013.
Puebla	November 2001, November 2004, November 2007, July 2010, July 2013.
Querétaro	July 2000, July 2003 , July 2006, July 2009, July 2012.
Quintana Roo	<i>February 1999</i> , February 2002 , February 2005, February 2008, July 2010, July 2013.
San Luis Potosí	July 2000, July 2003 , July 2006, July 2009, July 2012.
Sinaloa	November 2001, November 2004, October 2007, July 2010, July 2013.
Sonora	July 2000, July 2003 , July 2006, July 2009, July 2012.
Tabasco	October 2000, October 2003, October 2006, October 2009, July 2012.
Tamaulipas	October 2001, November 2004, November 2007, July 2010, July 2013.
Tlaxcala	November 2001, November 2004, October 2007, July 2010, July 2013.
Veracruz	September 2000, September 2004, September 2007, July 2010, July 2013.
Yucatán	May 2001, May 2004, May 2007, May 2010, July 2012.
Zacatecas	July 2001, July 2004, July 2007, July 2010, July 2013.

Table A1: Municipal elections, 1999-2013, by state

Notes: Emboldened election are counted as upcoming local elections in the survey analysis. State-level elections were held in Hidalgo in February 2002 without concurrent municipal elections, and are counted as upcoming local elections. Italicized elections are not included in the sample for the homicide shocks analysis due to data unavailability (or exclusion in the case of the Federal District). Except in the cases of Baja California 2001 and 2004 and Oaxaca 2013, missingness reflects the fact that data from the preceding election required to define the change in vote share was not available. For Baja California 2001 and 2004, the precinct numbering changed across elections and thus cannot be matched. For Oaxaca 2013, precinct level data was unavailable.

Watch and listen to news and political programs ever/monthly/weekly/daily. Indicator coded 1 for a respondent that answers that they watch political programs or listen to news at least ever/once a month/at least once a week/daily. ("¿Qué tan seguido escucha noticias o ve programas sobre política?")

Watch and listen to news and political programs scale. 5-point scale from 0 to 4, with values corresponding to levels of watching and listening to new and political programs (in ascending order).

Topical political knowledge. First factor from a factor analysis containing the following questions: What is the name of the youth movement that recently started in Mexico? (2012) Where was the plan to build an airport that was subsequently abandoned due to local pressure? (2003, 2005) Which political party intends to charge VAT on medicines, food, and tuition? (2001) Which party holds your state governorship? (2001, 2003, 2005, 2012) What is the name of your state Governor? (2001)

Education. 5-point variable, where 0 is less than completed primary education, 1 is a maximum education level of completing high school, 2 is a maximum level of completing lower secondary education, 3 is a maximum level of complete secondary education (*preparatoria*), and 4 is at least a university degree.

Homicide shock. This indicator is coded 1 if either the number of homicides in the two months prior to the month of the survey (including the survey month) or the two months prior to the survey month exceed those in the two months immediately after the month of the survey, based on the INEGI monthly homicide statistics for the occurrence of homicides among a municipality's residents. In 2005, the indicator is coded using the current month if the day of the month is greater than the 16th. (Note that homicides figures are subject to reclassification across time.)

Homicides last year. Average number of residents suffering a homicide per month within the municipality in the preceding 12 months (excluding the current month). (Note that homicides figures are subject to reclassification across time.)

Homicides last 3 years. Average number of residents suffering a homicide per month within the municipality in the preceding 36 months (excluding the current month). (Note that homicides figures are subject to reclassification across time.)

Public insecurity the major problem in the community. Indicator coded 1 if, in an open response, a respondent lists violence, crime or public security as the main problem facing their community (including as 0s respondents that listed no problem).

Low confidence in mayor. Indicator coded 1 for respondents answering that their confidence in the municipal mayor is 5 or below on a scale from 0 to 10 (for the 2003, 2005, and 2012 surveys) or

expressing no or almost no confidence in the municipal mayor (in 2001), in response to a question asking about the level of confidence that respondents have in the listed institutions.

Number of organizations. The number of organizations that a respondent reports being a member of, or previously being a member of. The number of possible organizations slightly varies across survey.

Organizations talk about politics. Indicator coded 1 for respondents that answer that politics is discussed at the organizations they are a member of.

Number of group meetings. The number of political organizations at which an individual has attended a meeting during the last year.

Discuss community problems. A scale measuring the regularity with which respondents discuss problems in the community with friends and neighbors, ranging through never (coded 0), occasionally (coded 1) and frequently (coded 2).

Incumbent win margin (lag). The difference in vote share between the incumbent and secondplaced finisher at the previous municipal mayoral election (or an election held later in the year of the survey). In *Usos y Custombres* in Oaxaca, the incumbent win margin is set to the maximum of 1.

ENPV(lag). The effective number of political parties (by vote share) at the previous municipal mayoral election (or an election held later in the year of the survey). In Usos y Custombres in Oaxaca, ENPV is set to the maximum of 1.

Incumbent won (lag). Indicator coded 1 for municipalities where the incumbent party was re-elected at the most recent election (or an election held later in the year of the survey).

Incumbent vote share (lag). The municipal vote share of the incumbent party at the most recent election (or an election held later in the year of the survey).

Police per voter (lag). Total number of municipal security employees in the previous year (in thousands), divided by the total number of registered voters.

A.3.2 Precinct and municipal homicide and electoral data

Change in incumbent party vote share. Change in the precinct-level share of all votes cast for the incumbent between the current municipal election and the prior municipal election (3 or 4 years earlier). When multiple parties form an incumbent coalition, the incumbent vote share is determined by the vote share of the largest party/coalition containing an incumbent party at the next election in terms of vote share.

Incumbent party win. Indicator coded 1 if the incumbent party wins the municipal election. In the case of coalitions, is defined similarly to the above.

Change in turnout. Change in the precinct-level turnout rate between the current municipal election and the prior municipal election (3 or 4 years earlier).

Change in incumbent vote share (registered). Change in the precinct-level share of votes, as a share of all registered voters, cast for the incumbent between the current municipal election and the prior municipal election (3 or 4 years earlier).

Homicide shock. Defined in equation (2) of the main paper, using INEGI homicide statistics for intentional homicides that occurred in each month to residents of a given municipality. Oneand three-month versions are similarly defined. (Note that homicides figures are subject to reclassification across time.)

Placebo homicide shock (6 months earlier). Defined as in equation (2) of the main paper, with the exception that all months are shifted 6 months forward in time. (Note that homicides figures are subject to reclassification across time.)

Average monthly homicide rate (12 months/3 years before election). Average number of residents suffering a homicide per month within the municipality in the 12 months/3 years prior to the municipal election, again based on INEGI homicide data. (Note that homicides figures are subject to reclassification across time.)

Post-2006. Indicator coded 1 for elections held since December 2006.

No municipal police force. Indicator coded 1 for municipalities without a municipal police force under its direct control. This category includes municipalities that work solely with state police or federal police, work with the community, run security using a private or other service, or have no service at all. Municipalities that share police forces or use civil associations were excluded because channels of accountability are hard to discern. These categorizations were homogenized across the 2000, 2002, 2004, 2011 and 2013 ENGM surveys. Missing years were imputed according to the following rules: I first used the last available data, and if no previous coding was available took the nearest year in the future.

Calderón Presidency. Indicator coded one for elections in the years 2007-2012.

PAN/PRD. Indicator coded 1 for PAN/PRD municipal incumbents.

A.3.3 Precinct local media coverage data

Local media. Number of AM radio, FM radio or television stations, with an emitter based in the precinct's municipality, covering at least 20% of the precinct population (as defined by detailed population data—block-level population in urban areas, and rural locality locations).

Local AM radio/FM radio/television. Number of AM radio/FM radio/television stations, with an emitter based in the precinct's municipal, covering at least 20% of the precinct population (as defined by detailed population data—block-level population in urban areas, and rural locality locations).

Non-local media. Number of AM radio, FM radio or television stations, with an emitter based outside the precinct's municipality, covering at least 20% of the precinct population (as defined by detailed population data—block-level population in urban areas, and rural locality locations).

High higher education. Indicator coded 1 for electoral precincts where more than 40% of residents had experienced higher education in 2010, according to the 2010 Census.

A.4 Map of municipalities included in different samples

In separate figures, Figure A1 shades in red the municipalities that appear at least once in each of the main empirical analyses—the survey sample, the homicide shock sample, and the local media sample.

A.5 Additional analyses

The following subsections present the results of additional analyses cited in the main paper.

A.5.1 Assessing the identification assumptions

Tables A2-A10 show the results of balance tests for the three main sets of empirical findings in the paper: Table A2 shows that upcoming local elections are well balanced across individual and municipal characteristics in the ENCUP survey data; Tables A6 and A7 shows that homicide shocks are well balanced across a wide variety of covariates, where municipality fixed effects are excluded for time-invariant Census and geographic variables; and Table A10 shows that the number of local media stations is well-balanced across these same covariates.

In addition, Tables A3 and A4 show that homicides and their interaction with upcoming local elections are well-balanced in the ENCUP surveys. Moreover, Table A5 demonstrates that neither changes in upcoming local elections nor changes in measures of local violence predict whether a municipality is included in any given year. In each panel the outcome is an indicator for whether a municipality is included in that survey wave, conditional on a municipality appearing in at least one of the four ENCUP surveys.

I also provide additional tests to support the exogeneity of homicide shocks. First, using a simple specification containing municipality and year fixed effects, Table A8 shows that the number of police per voter does not systematically vary across election and non-election years. Second, Table A9 shows that election outcomes are uncorrelated with homicide counts in the two months



(a) Municipalities in the ENCUP survey samples



(b) Municipalities in the homicide shock sample



(c) Municipalities in the local media neighbor sample

Figure A1: Municipalities included in each empirical analysis (shaded in red)
after elections. In particular, columns (3) and (4) find no correlation between the identity of the winning party and post-election homicides either throughout the sample or during the Calderón administration. This does not conflict with Dell (2015), who focuses on drug-related homicides over the *subsequent* mayor's term, or as many months as possible of that mayor's term.

A.5.2 Drug-related homicides

As noted in the main text, the Calderón government released monthly municipal data on the number of drug-related deaths between 2007 and 2011. However, there are data issues with such data. First, these numbers are contentious (see Heinle, Rodríguez Ferreira and Shirk 2014), and do not necessarily follow the homogeneous coroner's report criteria used by INEGI. Second, the limited availability of this data, combined with the definition of a homicide shock that requires at least one drug-related death over the four-month window around elections, substantially reduces the sample size by around 75%. More generally, it is not clear theoretically whether voters should respond more or less to drug-related homicides, as opposed to other homicides. In fact, voters might think that these are less relevant to them than more arbitrary murders which constitute around 50% of totals homicides even during the drug war (only a tiny fraction of such homicides as domestic). Nevertheless, I use the drug-related homicide data as a robustness check.

The results in Table A11 report the effect of drug-related homicide shocks just before an election, defined according to equation (2) but instead using drug-related homicides. Given that only five years of data exist, and thus many municipalities only appear once, I use state fixed effects instead of municipality fixed effects. Although the large drop in sample size unsurprisingly reduces precision substantially, the point estimates are relatively similar—negative, and large—to those reported in Table 2. These findings thus further reinforce the claim in the main text that voters respond similarly to different types of homicide.

A.5.3 Difference-in-differences approach to identifying the effects of pre-election homicides

The results in the main text focus on homicide shocks coded as a binary variable. A key advantage of this approach is that it is highly short-term, and thus accurately maps political information cycles. By exploiting idiosyncratic shifts in monthly homicide counts, I am able to generate plausibly exogenous variation. However, I also now consider an alternative approach to capturing the effects of short-run shocks around local elections.

In particular, I utilize a difference-in-differences design to identify how proportional changes in the number of homicides around elections affects vote choice. Since the homicide *level*, as opposed to the short-term *change* exploited in the main paper, just before an election is highly correlated

	Female	Catholic	Age	Education	Employed last year	Own econ. in month	Number of	Org.s talk about	Number of group
	(1)	(2)	(3)	(4)	(5)	(9)	org.s (7)	politics (8)	meetings (9)
Upcoming local election	-0.007 (0.011)	-0.011 (0.020)	-0.100 (0.394)	0.202*** (0.075)	-0.015 (0.016)	-0.029 (0.021)	-0.066 (0.077)	-0.013 (0.021)	-0.084 (0.103)
Observations Outcome mean Local election mean	15,976 0.55 0.16	$11,983 \\ 0.81 \\ 0.20$	15,976 40.76 0.16	11,756 1.70 0.18	$11,756 \\ 0.47 \\ 0.18 \\ 0.18$	12,322 1.57 0.20	15,976 1.06 0.16	12,576 0.26 0.20	15,976 1.52 0.16
	Discuss community problems (10)	Homicide shock (11)	Homicide last year (12)	Homicide last 3 years (13)	Incumbent win margin (lag) (14)	ENPV (lag) (15)	Incumbent won (lag) (16)	Incumbent vote share (13) (17)	Police per voter (last year) (18)
Upcoming local election	0.005 (0.029)	-0.022 (0.081)	1.607 (1.080)	1.607 (1.090)	0.022 (0.016)	0.011 (0.066)	0.094 (0.064)	0.012 (0.011)	-0.063 (0.178)
Observations Outcome mean Local election mean	12,576 0.70 0.20	12,664 0.44 0.17	15,941 4.50 0.16	15,941 4.82 0.16	15,976 0.15 0.16	15,976 2.57 0.16	15,976 0.61 0.16	15,778 0.48 0.17	13,666 2.30 0.16

Table A2: Balance of upcoming local elections in the ENCUP surveys over 18 individual and municipal level variables

A14

*** denotes p < 0.01.

	Female	Catholic	Age	Education	Employed last year	Own econ. in month	Number of	Org.s talk about	Number of group
	(1)	(2)	(3)	(4)	(5)	(9)	org.s (7)	politics (8)	meetings (9)
Homicide shock	-0.002	-0.008	-0.267	-0.010	-0.006	0.024	0.026	-0.010	0.029
	(0.007)	(0.012)	(0.273)	(0.053)	(0.012)	(0.016)	(0.064)	(0.014)	(0.062)
Observations	12,664	9,522	12,664	9,464	9,464	9,591	12,664	9,764	12,664
Outcome mean	0.55	0.81	40.58	1.79	0.47	1.57	1.05	0.26	1.45
Homicide shock mean	0.44	0.44	0.44	0.45	0.45	0.46	0.44	0.46	0.44
	Discuss community problems (10)	Upcoming local election (11)	Homicide last year (12)	Homicide last 3 years (13)	Incumbent win margin (lag) (14)	ENPV (lag) (15)	Incumbent won (lag) (16)	Incumbent vote share (lag) (17)	Police per voter (last year) (18)
Homicide shock	-0.015	-0.010	-2.357*	-3.143*	0.002	-0.032	-0.000	0.004	-0.139
	(0.020)	(0.036)	(1.267)	(1.860)	(0.010)	(0.042)	(0.041)	(0.006)	(0.106)
Observations	9,764	12,664	12,664	12,664	12,664	12,664	12,521	12,606	$10,584 \\ 2.17 \\ 0.45$
Outcome mean	0.69	0.17	5.62	6.03	0.15	2.56	0.54	0.48	
Homicide shock mean	0.46	0.44	0.44	0.44	0.44	0.44	0.44	0.44	

Table A3: Balance of homicide shock in the ENCUP surveys over 18 individual and municipal level variables

*** denotes p < 0.01.

Table A4: Interactive balance of upcoming local election and homicide shock in the ENCUP surveys over 17 individual and municipal level variables

			þ		Last year	in month	of	Urg.s talk about	Number of group
	(1)	(2)	(3)	(4)	(5)	(9)	(7)	politics (8)	(6)
Upcoming local election -(0.007	-0.028	-0.458	0.156	-0.007	-0.012	-0.076	-0.039	-0.092
(C	0.018)	(0.031)	(0.595)	(0.116)	(0.025)	(0.034)	(0.108)	(0.028)	(0.123)
Homicide shock -(0.005	-0.017	-0.307	-0.015	-0.003	0.033*	0.060	-0.008	0.068
(0	0.008)	(0.013)	(0.319)	(0.061)	(0.014)	(0.020)	(0.068)	(0.015)	(0.060)
Homicide shock C	0.022	0.042	0.218	0.023	-0.014	-0.043	-0.215*	-0.014	-0.249
× Upcoming local election (0	0.023)	(0.032)	(0.859)	(0.160)	(0.032)	(0.047)	(0.128)	(0.032)	(0.188)
Observations 12	2.664	9.522	12.664	9.464	9.464	9.591	12.664	9.764	12.664
Outcome mean	0.55	0.81	40.58	1.79	0.47	1.57	1.05	0.26	1.45
Upcoming local election mean Homicide shock mean									
D	liscuss	Upcoming	Homicide	Homicide	Incumbent	ENPV	Incumbent	Incumbent	Police
com	nmunity	local	last year	last 3 years	win margin	(lag)	won	vote share	per voter
brc	oblems	election			(lag)		(lag)	(lag)	(last year)
-	(10)	(11)	(12)	(13)	(14)	(15)	(16)	(17)	
Upcoming local election -(0.031	-0.429	-1.097	0.056^{**}	0.071	0.159*	0.008	-0.157	
(0	0.047)	(1.504)	(1.748)	(0.023)	(0.095)	(0.091)	(0.017)	(0.257)	
Homicide shock -(0.009	-3.110^{*}	-4.139*	0.00	-0.018	0.020	0.003	-0.176	
(0	0.025)	(1.585)	(2.267)	(0.012)	(0.053)	(0.048)	(0.008)	(0.123)	
Homicide shock -(0.031	4.610^{**}	6.067^{**}	-0.036	-0.085	-0.110	0.004	0.210	
× Upcoming local election (0	0.056)	(2.256)	(2.816)	(0.035)	(0.126)	(0.139)	(0.025)	(0.332)	
Observations 9	9,764	12,664	12,664	12,664	12,664	12,521	12,606	10,584	
Outcome mean (0.69	5.62	6.03	0.15	2.56	0.54	0.48	2.17	
Upcoming local election mean									
Homicide shock mean									

*** denotes p < 0.01.

Panel A: Linear predictors	5	Surveyed n	nunicipali	ty indicato	or
	(1)	(2)	(3)	(4)	(5)
Local election	0.016				
Homicide within last month	(0.057)	-0.020			
Homicide shock		(0.039)	0.036		
Homicides last year			(0.039)	-0.001	
Homicides last 3 years				(0.001)	-0.000 (0.001)
Observations Outcome mean	2,092 0.46	2,088 0.46	1,351 0.52	2,088 0.46	2,088 0.46
Panel B. Flection-homicide interactions	Surve	ved munic	inality ind	licator	
Tanci D. Election-nonnetice interactions	(1)	(2)	(3)	(4)	
Local election	0.032	0.050	0.021	0.020	
	(0.055)	(0.075)	(0.042)	(0.042)	
Homicide within last month	-0.017				
Local election \times Homicide within last month	(0.040) -0.029 (0.070)				
Homicide shock	· /	0.052			
		(0.043)			
Local election \times Homicide shock		-0.124			
Homicides last year		(0.102)	-0.001		
Tomendes fast year			(0.001)		
Local election \times Homicides last year			-0.003		
			(0.004)		
Homicides last 3 years				-0.000	
Local election \times Homicides last 3 years				(0.001)	
				(0.004)	
Observations	2,088	1,351	2,088	2,088	
Outcome mean	0.46	0.52	0.46	0.46	

Table A5: Predictors of municipalities included in each survey wave

Notes: All specifications include municipality and survey-year fixed effects, and are estimated using OLS. Standard errors clustered by municipality are in parentheses. * denotes p < 0.1, ** denotes p < 0.05, *** denotes p < 0.01.

Table A6: Municipal-level balance on 102 variables over pre-election homicide shock	ks
---	----

Outcome	Effect of	of homicide sho	ock	Outcome	Effect o	f homicide sh	ock
	Coef.	SE	Obs.		Coef.	SE	Obs.
Electorate	5138.2785	(6070.0779)	3,314	Single victims before election	0.0223	(4.7211)	2,169
Electorate density	-0.0002	(0.0005)	3,314	Job-related homicides before election	0.9489	(1.4437)	1,631
Incumbent vote share (lag)	-0.0003	(0.0094)	3,279	Organs examined after homicide before election	2.3932	(9.4195)	2,001
Incumbent win (lag)	0.0389	(0.0568)	2,814	Family-incident homicides before election	0.0489	(0.0839)	2,217
Municipality win margin (lag)	0.0035	(0.0142)	3,279	Urban homicide victims before election	1.3901	(10.838)	1,870
Municipality ENPV (lag)	-0.0225	(0.0472)	3,314	Non-Mexican homicide victims before election	-0.0303	(0.1554)	2,264
Turnout (lag)	-0.0172*	(0.0099)	3,121	Neighbor average homicide shock	0.0049	(0.0348)	3,172
PRI incumbent	-0.0722	(0.0509)	3,314	Electorate-weighted neigh. ave. hom. shock	-0.0062	(0.0352)	3,307
PAN incumbent	0.0498	(0.0516)	3,314	Average non-homicide deaths (prior 3 years)	-2.7841	(2.6206)	3,314
PRD incumbent	0.0244	(0.0384)	3,314	Average non-homicide deaths (prior year)	-4.1514	(2.9176)	3,314
Mayor's party aligned with President	0.0668	(0.0476)	3,314	Average child mortalities (prior 3 years)	-0.0511	(0.3519)	2,615
Mayor's party aligned with Governor	-0.0790	(0.0527)	3,314	Average child mortalities (prior year)	0.1771	(0.3819)	2,615
Mayor's party aligned with President and Governor	0.0076	(0.0298)	3,314	Total municipal spending	24.6691	(87.6639)	2,063
Homicides 12 months before election	-2.0812	(2.9771)	3,314	Police per voter	0.0555	(0.0905)	3,314
Homicides 11 months before election	-1.5810	(3.6363)	3,314	Area (km ²)	419.0371*	(219.2123)	3,314
Homicides 10 months before election	-2.2831	(3.4014)	3,314	Local media	-1.0697	(1.2086)	3,309
Homicides 9 months before election	-2.1495	(3.2739)	3,314	Non-local media	-0.3218	(1.3307)	3,309
Homicides 8 months before election	-2.9404	(2.6120)	3,314	Occupants per dwelling	-0.0198	(0.0153)	3,283
Homicides 7 months before election	-2.0478	(2.8407)	3,314	Occupants per room	0.0030	(0.0127)	3,283
Homicides 6 months before election	-1.9702	(3.0893)	3,314	Share with 2 bedrooms	-0.0088	(0.0076)	3,283
Homicides 5 months before election	-1.5124	(2.2152)	3,314	Share 3+ bedrooms	-0.0074	(0.0081)	3,283
Homicides 4 months before election	-0.7409	(2.8034)	3,314	Share female	-0.0003	(0.0008)	3,283
Homicides 3 months before election	-1.0332	(2.9280)	3,314	Share working age	0.0006	(0.0019)	3,283
Average homicides (prior 3 years)	-0.7327	(1.8893)	3,314	Children per woman	-0.0036	(0.0188)	3,283
Average homicides (prior year)	-1.8340	(2.9060)	3,314	Share born out of state	-0.0038	(0.0142)	3,283
Top homicide quartile (prior year)	0.0080	(0.0427)	3,314	Share Catholic	-0.0009	(0.0053)	3,283
Top homicide decile (prior year)	0.0063	(0.0268)	3,314	Share indigenous speakers	0.0025	(0.0036)	3,283
Presence of DTO	0.0053	(0.0427)	2,508	Years of schooling	-0.0203	(0.0865)	3,283
Average drug-related homicides (prior year)	0.0064	(0.0141)	3,314	Female years of schooling	0.0119	(0.0867)	3,283
Reports of inter-DTO violence before election	-0.4655	(1.1353)	2,420	Male years of schooling	0.0003	(0.0919)	3,283
Reports of violent enforcement before election	0.0298	(0.1511)	2,420	Share illiterate	0.0005	(0.0028)	3,283
Reports of arrests before election	0.0979	(1.0800)	2,420	Share with no schooling	-0.0002	(0.0026)	3,283
Reports of drug seizures before election	0.5847	(1.5005)	2,420	Share incomplete primary school	0.0002	(0.0026)	3,283
Reports of asset seizures before election	-0.3351	(0.3663)	2,420	Share complete primary school	-0.0017	(0.0057)	3,283
Reports of gun seizures before election	0.4696	(0.6290)	2,420	Share incomplete secondary school	0.0010	(0.0073)	3,283
Gun-related homicides before election	-0.1996	(6.9310)	3,314	Share complete secondary school	0.0018	(0.0077)	3,283
Chemical substance-related homicides before election	-0.0279	(0.0221)	3,314	Share higher education	0.0028	(0.0082)	3,283
Hanging-related homicides before election	0.1386	(0.2559)	3,314	Share economically active	-0.0027	(0.0032)	3,283
Drowning-related homicides before election	0.1104**	(0.0500)	3,314	Share without health care	-0.0014	(0.0061)	3,283
Explosives-related homicides before election	0.0049	(0.0063)	3,314	Share state workers health care	0.0026	(0.0022)	3,283
Smoke/fire-related homicides before election	0.0328	(0.0407)	3,314	Share running water	-0.0059	(0.0067)	3,283
Cutting object-related homicides before election	0.6499**	(0.2857)	3,314	Share drainage	-0.0038	(0.0053)	3,283
Blunt object-related homicides before election	0.1540	(0.1037)	3,314	Share washing machine	-0.0095	(0.0103)	3,283
Delayed homicide registrations before election	1.8169	(7.4294)	3,314	Share landline telephone	-0.0118	(0.0147)	3,283
Male homicide victims before election	1.7658	(8.6080)	2,262	Share radio	-0.0043	(0.0065)	3,283
Age of homicide victims before election	1.5388*	(0.8059)	2,225	Share fridge	-0.0056	(0.0078)	3,283
Victims with no schooling before election	0.3240	(0.3977)	2,118	Share cell phone	-0.0023	(0.0104)	3,283
Victims with incomplete primary schooling before	1.4145	(9.7034)	2,118	Share television	-0.0021	(0.0032)	3,283
Victims with complete primary schooling before election	0.9600	(9.1598)	2,118	Share car or truck	-0.0044	(0.0100)	3,283
Victims with incomplete secondary schooling before election	1.3192	(5.6828)	2,118	Share computer	-0.0052	(0.0108)	3,283
Victims with complete secondary schooling before election	0.9925	(5.5046)	2,118	Share internet	-0.0043	(0.0097)	3,283

Notes: Specifications include municipality and year fixed effects, and are estimated using OLS. The time-invariant variables—area, media coverage, and the 2010 Census variables (occupants per dwelling-share internet)—exclude municipality fixed effects. Standard errors clustered by municipality in parentheses. * denotes p < 0.1, ** denotes p < 0.05, *** denotes p < 0.01.

|--|

Outcome Outcome <t< th=""><th>Outcome</th><th>Effect</th><th>of homioido sk</th><th>oalr</th><th>Outcome</th><th>Effect</th><th>of homioida</th><th>shoak</th></t<>	Outcome	Effect	of homioido sk	oalr	Outcome	Effect	of homioida	shoak
Locat June June <t< td=""><td>Outcome</td><td>Coef</td><td>SE SE</td><td>Obc</td><td>Outcome</td><td>Coef</td><td>SE</td><td>Obs</td></t<>	Outcome	Coef	SE SE	Obc	Outcome	Coef	SE	Obs
Betchnate 80.810 (53.490 P4,148 Victures with incomplete secondary schooling before election 12.815 (4.108) (4.508) Beconrate density 163.57 (41.348) P1.156 Single victures schooling vectorie decision 4.021 (4.108) 14.550 Beconrate density (10.011) 163.571 (41.348) P1.156 Single victures before election 4.021 (41.348) 17.345 Win rangin (lag) 0.0012 0.00001 17.341 Harm Hymicole before election 0.048 (00.056) 17.448 Win rangin (lag) 0.0017 0.00161 17.441 Nom-Mexican homicide before election 0.028 (11.410) 18.957 Municipality ENV (lag) 0.0179 0.0015 17.441 Nom-Mexican homicide before election 0.028 (17.451) Turnort (lag) 0.0179 0.0016 17.441 Nom-Mexican homicide dealts (prior 3 years) 2.275 (2.018) 17.444 PN incombent 0.0078 17.446 Nearge onthitomicide dealts (prior 3 years) 2.2464 66.5060 17.451			5L	003.			51	003.
Munic pulpic velocinate 5220 750 (4737 4467) 73.861 Victurs with complete secondary schedung before election 0.9457 (4.1018) 14.39.85 Incomment vote share (lag) 0.0005 0.0072 174.485 Job reliated homacides before election 0.0051 1.018.88 0.0051 1.018.88 0.0051 1.018.85 0.0051 1.018.85 0.0051 1.018.85 0.0051 1.018.85 0.0051 1.018.85 0.0051 1.018.95 0.0051 1.019.95 0.011.95 0.0011 0.0013 0.0118 1.74.148 Nemicipative schematices before election 0.023 0.0118 1.74.148 Nemicipative schematices before election 0.023 0.0114 1.94.95 Nemicipative schematices before election 0.0060 0.0023 0.0141 1.94.95 Nemicipative schematices before election 0.0060 0.0021 1.74.48 Nemicipative schematices before election 0.0060 0.0021 1.74.48 Nemicipative schematices before election 0.0066 0.0021 1.74.48 Nemicipative schematices before election 0.006 1.74.148 Nemicipative schematinge schematices before election 0	Electorate	80.8510	(53.4489)	174,148	Victims with incomplete secondary schooling before election	1.2815	(4.1507)	145,965
Lectoring density 18.2.70 (41.948) 11.567 Single victums before election -0.025 (1.9488) 14.958 Wins margin (lag) 0.0005 (0.007) 17.4.18 Family indices here election 0.0455 (0.066) 147.058 Wins margin (lag) 0.0012 (0.010) 17.3.41 Family indices here election 0.0455 (0.066) 147.058 Nunicipality yoin margin (lag) -0.0233 (0.011) 17.3.41 Family indices here election -0.0328 (0.114) 12.9.45 Nunicipality Lumoot (lag) -0.0217 (0.007) 16.4.05 Neweigneon-homicide denfits piors in years) -2.040 0.0040 (0.0223) 17.4.148 PM incombent -0.0179* (0.007) 14.148 Average chil monicide denfits pior years) -4.1968 (2.2467) 17.4.148 PM incombent -0.078* (0.007) 17.4.148 Average chil monicide denfits pior years) -2.466 (6.306) 14.5151 PM incombent -0.078* (0.072) 17.4.148 Average chil monicide denfits pior years) -2.4664 (6.306)	Municipality electorate	5220.7501	(4737.3467)	173,863	Victims with complete secondary schooling before election	0.9487	(4.0218)	145,965
Incumbent vice share (lag) 0.0012 14,448 Job-reliated normicable before election 0.9524 (1.006) 12.468 Win margin (lag) 0.0141 0.0141 14.345 Corpans continued larb homicable before election 0.0234 (1.006) 12.448 Win margin (lag) 0.0010 17.341 Firstly hundrated larb homicable before election 0.0238 (1.106) 12.448 Numicapit (lag) -0.0038 (0.1018) 17.441 Nom-Mexicab book 0.0046 0.0238 (1.106) 12.458 Numicapity ENPV (lag) -0.0179* (0.0074) 16.455 Normace cambo homicable datas phase hook. 0.0046 (0.0228) 17.448 PAN incombert 0.0479 (0.0074) 16.455 Average onhomicable datas prior 3 yeans) -0.4168 (0.0280) 17.448 PAN incombert 0.0229 17.448 Average child mortallites (prior yeans) -0.4168 (0.0280) 17.448 PAN incombert 0.0229 17.444 Average child mortallites (prior yeans) -0.4168 (0.0158) 17.4548 PAN incombert 0.02	Electorate density	18.5276	(41.3484)	171,567	Single victims before election	-0.0251	(3.4588)	147,395
Incumbent vm (lag) 0.0414 14.5.43 Organs examined after homicales before election 2.4184 (0.847) 14.7480 Municipally vin margin (lag) 0.0012 (0.000) 174,148 Funly-incident intervices before election 1.2798 (7.5541 127.254 Municipally run margin (lag) 0.0179** (0.0010) 174,148 Neighborts intervices before election 0.00040 (0.0212) 174,048 Municipally run (lag) 0.0179** (0.077) 164,157 Electronice developed neighborts were place horns shock 4.0006 (0.027) 174,158 PRI incumbent 0.0179** (0.029) 174,148 Average cubi nontalities (prior year) 4.1908 (2.2467) 174,148 PRD incumbent 0.0299 174,148 Average cubi nontalities (prior year) 0.1809 (0.2308) 184,503 Mayor's part aligned with President 0.0258 (0.2317) 174,148 Average cubi nontalities (prior year) 0.1809 (0.2308) 184,51 Mayor's part aligned with President all Governor 0.0058 (0.0458) 174,148 Norte cubi nontalities (prior year) <td< td=""><td>Incumbent vote share (lag)</td><td>0.0005</td><td>(0.0072)</td><td>174,148</td><td>Job-related homicides before election</td><td>0.9524</td><td>(1.0067)</td><td>129,680</td></td<>	Incumbent vote share (lag)	0.0005	(0.0072)	174,148	Job-related homicides before election	0.9524	(1.0067)	129,680
Wn margin (lag) 0.0012 (0.0000) 14,448 Family-nedent homacids tection 0.0488 (0.0020) 14/148 BNM incipality FDPV (lag) -0.0233 (0.0318) 174,455 Non-Maccian homacids victims before election -0.0328 (0.1140) 149,857 Dimoipality FDPV (lag) -0.0217 (0.0003) 174,445 Non-Maccian homacids victims before election -0.0328 (0.0127) 174,145 PM incipality FDPV (lag) -0.0179** (0.0000) 174,148 Rectine-weighted night, non-shock -0.0189* (0.0272) 174,148 PN incimbert 0.009 174,148 Average cinh homachilise (prin y year) -1.098* 2.4064 (0.0272) 174,148 PN incimbert 0.0299 174,148 Average cinh montalise (prin year) -0.0582 (0.0681) 174,543 PN incimbert 0.0208 174,148 Average cinh montalise (prin year) -0.0582 (0.0681) 173,553 Mayor's pary aligned with Greened -0.0788 (0.0329) 174,148 Vaeta (m ²) -0.0581 174,158 Mayor's pary aligned with Gr	Incumbent win (lag)	0.0410	(0.0414)	145,432	Organs examined after homicide before election	2.3184	(6.8473)	142,448
Municipality vin margin (lag) 0.003 0.00110 12,24 Urban Margin (lag) 12,25 Municipality ENPV (lag) 4.0021 0.0033 174,145 New Advican homicide vicinus before election 4.0023 0.01318 174,145 New Advican homicide vicinus before election 4.0023 0.0235 174,145 Municipality Immont (lag) 4.00714 0.00714 14,145 New Advican homicide deaths (prior 3 years) 4.198* 2.2578 (2.216) 174,148 PRI incumbent 0.0719 10.0350 174,148 Average child mortalities (prior 3 years) 4.098* 2.2467 174,148 PRD incumbent 0.0269 174,148 Average child mortalities (prior 3 years) 0.0460 0.2508 183,503 Mayor's party aligned with Posisient and Governor 0.0656 174,148 Total mortaling spending 4.0674 2.2368 173,563 Mayor's party aligned with Posisient and Governor 0.0788* (0.0429) 174,148 Area (mri) 4.0514* 12.020 173,453 Menicisal 12 months before election -2.1576 (2.3434) 174,148 <t< td=""><td>Win margin (lag)</td><td>0.0012</td><td>(0.0090)</td><td>174,148</td><td>Family-incident homicides before election</td><td>0.0485</td><td>(0.0605)</td><td>147,050</td></t<>	Win margin (lag)	0.0012	(0.0090)	174,148	Family-incident homicides before election	0.0485	(0.0605)	147,050
EAPY (lag)0.023 (0.038) 174,145 Non-Maxxan homicale varge homisicle shock - 0.0048 (0.027) 174,085 Minicipally runout (lag) - 0.0179* (0.037) 144,148 Neighbor varge homicale shock - 0.0040 (0.027) 174,085 Minicipally runout (lag) - 0.0179* (0.037) 144,148 Neighbor varge homicale daths (prior 3 year) - 0.2877 (0.238) 174,148 Normal participal specific states with membrat - 0.0499 (0.097) 174,148 Neighbor varge child merulines (prior 3 year) - 0.0499 (0.2377) 135,503 Mayor's party aligned with Dresident - 0.0668* (0.026) 174,148 Varenge child merulines (prior year) - 0.0499 (0.2377) 135,503 Mayor's party aligned with Dresident - 0.0678* (0.0450) 174,148 Varenge child merulines (prior year) - 0.0499 (0.0590) 174,148 Varenge child merulines (prior year) - 0.0592 (0.0494) 173,553 Mayor's party aligned with Dresident - 0.0678* (0.0450) 174,148 Varenge child merulines (prior year) - 0.0592 (0.0494) 173,563 Mayor's party aligned with Dresident - 0.0788* (0.0455) 174,148 Varenge child merulines (prior year) - 0.0592 (0.0494) 173,563 Mayor's party aligned with Dresident - 2.2558 (2.2496) 174,148 Varenge child merulines (prior year) - 0.0592 (0.0159) 173,513 Homicides 11 months before election - 2.2578 (2.2496) 174,148 Varenge child merulines (prior year) - 0.0580 (0.0079) 173,563 Homicks 51 months before election - 2.2493 (2.2496) 174,148 Varenge Availance - 0.0053 (0.0079) 173,563 Homicks 51 months before election - 2.0493 (2.2149 174,148 Varenge Availance - 0.0053 (0.0079) 173,682 Homicks 51 months before election - 1.0475 (1.2287) 174,148 Varenge Mayoff and (1.448 Varenge Mayoff	Municipality win margin (lag)	0.0035	(0.0110)	173,241	Urban homicide victims before election	1.2799	(7.5544)	127,256
	ENPV (lag)	-0.0203	(0.0318)	174,145	Non-Mexican homicide victims before election	-0.0328	(0.1140)	149,867
Limoti (lag) -0.0179** (00.074) 164.05 Jectorate-weighted neghts weighted neghts weights weights weighted neghts weights weights weights weighted n	Municipality ENPV (lag)	-0.0217	(0.0363)	174,148	Neighbor average homicide shock	0.0040	(0.0268)	170,451
Municipally lumout (lig.)-0.01-0.0111.15Nearing to a homical deaths (prior 3 years)-2.82.8(2.2467)17.4.18PKI incumbent0.004900.0039017.4.148Verage on-homical deaths (prior year)-0.004690.2280717.4.18PKI incumbent0.0029017.4.148Verage of hild montalities (prior year)0.014600.22870185.503Mayor's party aligned with Desident and Governor-0.00660.0066710.4.148Total municipal spendities (prior year)0.40600.2380817.8.653Mayor's party aligned with Desident and Governor-0.00550.0029717.4.148Vara (knf)4.675442.13850171.567Homicides 11 months before election-1.5767(2.3343)174.148Non-local media-0.0316(1.3461)17.2072Homicides 12 months before election-2.1625(2.26496)174.148Non-local media-0.0016(0.0018)17.3.83Homicides 7 months before election-2.1626(2.3420)174.148Nare yab bedrooms-0.0007(0.0074)17.3.682Homicides 7 months before election-1.9516(2.2000)174.148Nare yab bedrooms-0.0058(0.0079)17.3.682Homicides 5 months before election-1.9516(2.2000)174.148Nare per wama-0.0058(0.0079)17.3.682Homicides 6 months before election-1.9516(2.2000)174.148Nare of hermal school-0.0058(0.0079)17.3.682Homicides 5 months before election-1.9516<	Turnout (lag)	-0.01/9**	(0.0074)	164,673	Electorate-weighted neigh. ave. hom. shock	-0.0066	(0.0272)	174,095
PK1 incumbent -0.071* 00.390 174,148 Average child moralities (prior year) -4.90* (2.2467) 174,148 PKD incumbent 0.0229 00.2260 172,148 Average child moralities (prior year) 0.1800 0.2808 173,850 PKD incumbent 0.0078* 0.0078* 0.00405 174,148 Average child moralities (prior year) 0.1800 0.2808 173,850 Mayor's party aligned with President 0.0078* 0.0078* 0.00405 174,148 Police per veter 0.0059 0.00694 173,850 Mayor's party aligned with President and Governor -2.0558 0.2129 174,148 Neucomation -0.0156 0.0129 172,072 173,183 100,001 100,0074 173,813	Municipality turnout (lag)	-0.01/9**	(0.0074)	164,155	Average non-homicide deaths (prior 3 years)	-2.8278	(2.0218)	174,148
PN incumbent 0.0499 (0.1492) (174, 148 Average child mortalities (prior 3 years) -0.0499 (0.257) (185, 03) Mayor's party aligned with Governor 0.00668 (0.0066) 174, 148 Total municipal spending 24, 464 (65, 5036) 114, 551 Mayor's party aligned with Governor 0.0005 (0.0229) 174, 148 Area (m ²) 4, 67, 54* 2, 43860 171, 567 Mayor's party aligned with Develocit -1, 5767 (2, 3334) 174, 148 Area (m ²) -0, 0168 (0.0118) 173, 163 Homicides 11 months before election -2, 1262 (2, 5544) 174, 148 Non-local media -0, 0016 (0.0127) 173, 813 Homicides 40 months before election -2, 1620 (2, 5544) 174, 148 Nace with 2 bedrooms -0, 0007 (0.0071) 173, 853 Homicides 40 months before election -1, 927 (2, 4206) 174, 148 Nace working age -0, 0007 (0.0079) 173, 863 Homicides 40 months before election -1, 927 (1, 74, 148 Share working age -0, 0007 (0, 0019)	PRI incumbent	-0.0/11*	(0.0390)	1/4,148	Average non-homicide deaths (prior year)	-4.1908*	(2.2467)	1/4,148
PKD incumbent 0.0229 (174,184 Average Child motifiates (prior year) 0.1800 0.2808 138,503 Mayor's party aligned with President 0.0626 (0.0056) (174,184 Fold municipal spending 24.664 (66.5036) (14.551 Mayor's party aligned with President and Governor 0.0058 (0.0029) 174,148 Vac (m ²) 4.0754 (2.4864) (17.507 Homicides 12 months before election -2.0558 (2.312) 174,148 Local media -0.0166 (1.3461) (2.017) (17.2072) (17.2072) (17.417 Noncilas 10 months before election -2.162 (2.3544) (17.4148 Noncilas 10 months before election -0.0186 (0.0178) (17.307) Homicides 8 months before election -2.0490 (2.2440) 174,148 Nare female -0.0007 (0.0007) (17.237) Homicides 6 months before election -1.9516 (2.2400) 174,148 Share female -0.0018 (0.0078) (17.833) Homicides 6 months before election -1.9516 (2.2400) 174,148 Share female -0.0018 <td>PAN incumbent</td> <td>0.0499</td> <td>(0.0392)</td> <td>174,148</td> <td>Average child mortalities (prior 3 years)</td> <td>-0.0469</td> <td>(0.2577)</td> <td>138,503</td>	PAN incumbent	0.0499	(0.0392)	174,148	Average child mortalities (prior 3 years)	-0.0469	(0.2577)	138,503
Mayor's party aligned with President 0.0668° (0.0496) $174, 148$ Foldar Hunchraghending 24.864 (66.0306) 142.51 Mayor's party aligned with President and Governor 0.0078° (0.0429) $174, 148$ Neice previer 0.0052 (0.0494) 173.863 Mayor's party aligned with President and Governor -2.058° (2.3129) $174, 148$ Non-local media -0.016 (1.0472) 172.072 Homicides 11 months before election -2.172° (2.6496) $174, 148$ Non-local media -0.016 (0.0158) 173.813 Homicides 9 months before election -2.2162° (2.5544) $174, 148$ Nare with 2 bedrooms -0.0070 (0.0070) 173.682 Homicides 7 months before election -2.0593 (2.2104) $174, 148$ Nare female -0.0040 0.0008 173.693 Homicides 7 months before election -1.975° (2.1824) $174, 148$ Nare female -0.0016 0.0018 173.693 Homicides 4 months before election -1.475° (2.1824) $174, 148$ Nare female -0.0016 0.0018 173.693 Homicides 5 months before election -0.174° (1.4901) $174, 148$ Nare indigenous speakers 0.0002 0.0024 0.0018 173.693 Homicides 6 prior gar) -1.2626 (2.376) $174, 148$ Nare indigenous speakers 0.0004 0.0024 0.0018 173.693 Average homicides (prior year) -1.2628 (2.2786) 174	PRD incumbent	0.0229	(0.0296)	174,148	Average child mortalities (prior year)	0.1800	(0.2808)	138,503
Mayor sparty aligned with Governor -0.078^{80} (0.0490) $174,148$ Folice per voler 0.00974 $173,280$ Homicides 12 months before election -2.0558 (2.3129) $174,148$ Local media -1.0871 (1.2072) $172,072$ Homicides 10 months before election -2.2578 (2.3129) $174,148$ Local media -0.0166 (1.3461) $172,072$ Homicides 10 months before election -2.2772 (2.6496) $174,148$ Nocupants per room 0.0025 (0.0127) $173,813$ Homicides 5 months before election -2.2499 (2.0342) $174,148$ Share with 2 bedrooms -0.0070 (0.0074) $173,682$ Homicides 7 months before election -1.9516 (2.4000) $174,148$ Share formale -0.0080 (0.0009) $173,693$ Homicides 6 months before election -1.9516 (2.4000) $174,148$ Share formale -0.0040 (0.0008) $173,693$ Homicides 6 months before election -1.0726 (1.827) $174,148$ Share formale -0.0040 (0.0019) $173,693$ Homicides 6 months before election -1.0726 (1.827) $174,148$ Share formale -0.0040 (0.0020) $173,093$ Average homicides (prior year) -0.1734 (1.4491) $174,148$ Share indegenous speakers -0.0040 (0.0024) $173,692$ Top homicide decite (prior year) 0.0067 (0.0300) $174,148$ Share indegenous speakers -0.0047 (0.0026) $173,852$ <td>Mayor's party aligned with President</td> <td>0.0668*</td> <td>(0.0366)</td> <td>174,148</td> <td>Total municipal spending</td> <td>24.4664</td> <td>(66.5036)</td> <td>114,551</td>	Mayor's party aligned with President	0.0668*	(0.0366)	174,148	Total municipal spending	24.4664	(66.5036)	114,551
Mayors party algned with President and Governor0.0045 (0.0229) $(1.4,148)$ Area (km ⁻)4.6/34"(2.4380) (1.150) Homicides 11 months before election-1.5767(2.8334) $(17.4,148)$ Non-local media-0.3016 (1.3072) (1.2072) (1.418) Share forma -0.0058 (0.0079) (1.723) Homicides for nonths before election -1.677 (1.4272) (1.414) Share forma -0.0016 (0.0008) (1.369) Average homicides (prior year) -1.687 (1.272) (1.414) Share catholic -0.0004 (0.0012) (1.363) Average homicides (prior year) -1.687 (1.272) (1.414) Share catholic -0.0004 (0.0014) (1.363) Average homicides (prior year) -1.687 (0.0074) (1.414) Share indigenous speakers 0.0024 (0.0048) (1.363) Average homi	Mayor's party aligned with Governor	-0.0788*	(0.0405)	1/4,148	Police per voter	0.0592	(0.0694)	1/3,863
Homicades 12 months before election -2.0528 (2.3129) $(7.4, 148)$ Local media $-1.08/1$ $(1.20/2)$ $172, 072$ Homicides 10 months before election -2.2772 (2.6496) $174, 148$ Occupants per room 0.0016 (0.0125) $173, 813$ Homicides 5 months before election -2.2693 (2.1042) $174, 148$ Share 3+ bedrooms -0.00070 (0.0071) $173, 813$ Homicides 6 months before election -2.9393 (2.2104) $174, 148$ Share 3+ bedrooms -0.0008 (0.0079) $173, 803$ Homicides 6 months before election -1.9516 (2.4000) $174, 148$ Share 3+ bedrooms -0.0018 (0.0182) $173, 693$ Homicides 6 months before election -1.0166 (2.278) $174, 148$ Share 2 ontof -0.0018 (0.0182) $173, 693$ Homicides (prior year) -0.0167 (0.3206) $174, 148$ Share Catholic -0.0004 (0.0048) $173, 693$ Homicide quartile (prior year) 0.0067 (0.0300) $174, 148$ Share Catho	Mayor's party aligned with President and Governor	0.0065	(0.0229)	174,148	Area (km ²)	4.6754*	(2.4386)	171,567
Homicadis I months before election-1.5/16/(2.83:4) $174,148$ Non-local media-0.016 (1.3461) $172,012$ Homicids I months before election-2.1626(2.5544) $174,148$ Occupants per dwelling0.0018 (0.0157) $173,813$ Homicids I months before election-2.0990(2.0124) $174,148$ Share strup the Decross 0.00070 (0.0079) $173,682$ Homicides I months before election-1.9516(2.0000) $174,148$ Share strup the Decross 0.00070 (0.0019) $173,693$ Homicides I months before election-1.9716(2.1000) $174,148$ Share working age 0.0007 (0.0019) $173,693$ Homicides I months before election-0.0166(2.287) $174,148$ Share working age 0.0007 (0.0018) $173,693$ Homicides Grior 3 years)-0.7374 (1.4691) $174,148$ Share thorn out state -0.0006 (0.0048) $173,693$ Average homicides (prior year)-1.8268 (2.284) $174,148$ Share Caholic -0.0006 (0.0048) $173,693$ Average homicides (prior year)0.0067 (0.030) $174,148$ Share Caholic -0.0006 (0.0048) $173,693$ Top homicide decile (prior year)0.0067 (0.030) $174,148$ Share Caholic -0.0052 (0.028) $173,813$ Average homicides (prior year)0.0067 (0.030) $174,148$ Share stoholing -0.0017 (0.0028) $173,852$ Reports of arks before election<	Homicides 12 months before election	-2.0558	(2.3129)	174,148	Local media	-1.08/1	(1.2072)	172,072
Homicades I0 months before election -2.2772 (2.6490) $174,148$ Occupants per weiling -0.0180 (0.0185) $173,813$ Homicides Months before election -2.9409 (2.0342) $174,148$ Share with 2 bedrooms -0.0070 (0.0074) $173,863$ Homicides Months before election -1.9516 (2.340) $174,148$ Share working age -0.0004 (0.0008) $173,693$ Homicides Months before election -1.9756 (2.1827) $174,148$ Share Fanale -0.0040 (0.0008) $173,693$ Homicides Months before election -1.0166 (2.278) $174,148$ Share rowana -0.0016 (0.0182) $173,693$ Average homicides (prior year) -0.0167 (2.1827) $174,148$ Share Catholic -0.0040 (0.0048) $173,693$ Average homicides (prior year) -0.0067 (0.020) $174,148$ Share Catholic -0.0046 (0.0034) $173,693$ Average homicides (prior year) 0.0067 (0.020) $174,148$ Share Catholic -0.0047 (0.0034) $173,813$ Top homicide quartile (prior year) 0.0067 (0.020) $174,148$ Share Catholic -0.0047 (0.023) $173,813$ Average fung-related homicides (prior year) 0.0067 (0.020) $174,148$ Share interper of schooling -0.0037 (0.023) $173,813$ Average drug-related homicides (prior year) 0.0067 (0.020) $135,005$ Male years of schooling -0.0037 (0.023)	Homicides 11 months before election	-1.5/6/	(2.8334)	174,148	Non-local media	-0.3016	(1.3461)	172,072
	Homicides 10 months before election	-2.2772	(2.6496)	174,148	Occupants per dwelling	-0.0186	(0.0158)	1/3,813
Homicide's months before election -2.9490 (2.0342) $(74,148$ Share with 2 bedrooms -0.0070 $(0.007)4$ $(73,682)$ Homicide's months before election -1.9516 (2.4000) $(74,148$ Share Female -0.0004 (0.0008) $(73,693)$ Homicide's months before election -1.9756 (2.1827) $(74,148$ Share working age 0.0007 (0.0019) $(73,693)$ Homicide's months before election -1.0166 (2.2786) $(74,148)$ Share working age -0.0016 (0.0182) $(73,693)$ Average homicides (prior y gens) -0.7374 (1.491) $(74,148)$ Share Catholic -0.0006 (0.0048) $(73,693)$ Average homicides (prior y gens) -0.7374 (1.491) $(74,148)$ Share Catholic -0.0016 (0.0026) $(173,693)$ Average homicides (prior y gens) 0.0774 (0.303) $(74,148)$ Share indigenous speakers 0.0027 (0.0039) $(133,693)$ Presence of DTO 0.0067 (0.0206) $(174,148)$ Share illiterate 0.0057 (0.0028) $(133,692)$ Reports of inter-DTO violence before election 0.4325 (0.8225) (0.8225) (0.8225) (0.8225) (0.8226) (0.826) (0.0026) $(133,682)$ Reports of ing exizures before election 0.1254 (0.7871) $(130,792)$ Share incomplete primary school 0.0006 (0.0026) $(133,682)$ Reports of an exizures before election 0.4278 (1.0101) $(130,792)$ S	Homicides 9 months before election	-2.1626	(2.5544)	1/4,148	Occupants per room	0.0025	(0.0127)	1/3,813
Homicides / months before election -2.0593 (2.2104) $(74.148$ Share 3-bedrooms -0.0088 (0.0098) (73.693) Homicides for onths before election -1.9516 (2.2000) (74.148) Share working age 0.0007 (0.0019) (73.693) Homicides for nonths before election -0.7296 (2.1287) (74.148) Share born out of state -0.0048 (0.0129) (73.693) Average homicides (prior 3 years) -0.7374 (1.4691) (74.148) Share Catholic -0.0024 (0.0129) (73.693) Average homicides (prior year) -1.8208 (2.2634) (74.148) Share Catholic -0.0016 (0.0048) (73.693) Top homicide decile (prior year) 0.0067 (0.0206) (74.148) Share indigenous speakers 0.0024 (0.0203) (73.813) Presence of DTO 0.0067 (0.0206) (74.148) Share indigenous speakers 0.0087 (0.023) (73.682) Reports of inter-DTO violence before election 0.0057 (0.0109) (74.148) Share with no schooling -0.0004 (0.0026) (73.682) Reports of inter-DTO violence before election 0.0345 (0.872) (1.971) (30.792) Share complete primary school 0.0004 (0.0026) (73.682) Reports of arrests before election 0.254 (0.7871) $(1.30.792)$ Share conduct school (0.0021) (73.682) Reports of arrests before election 0.6278 (1.0910) (30.792) Share c	Homicides 8 months before election	-2.9409	(2.0342)	174,148	Share with 2 bedrooms	-0.0070	(0.0074)	173,682
Homicides 6 months before election -1.9516 (2.4000) $(74,148$ Share tende -0.0004 (0.0018) $(73,693)$ Homicides 7 months before election -1.4975 (1.7223) $71,4148$ Share working age 0.0007 0.0018 (0.0129) $173,693$ Homicides 1 months before election -1.0726 (2.1827) $174,148$ Share vorking age -0.0049 (0.0129) $173,693$ Average homicides (prior 3 years) -0.7374 (1.4691) $174,148$ Share catholic -0.0006 (0.0084) $173,693$ Average homicides (prior year) -0.0757 (0.0303) $174,148$ Share catholic -0.0016 (0.0084) $173,813$ Top homicide decile (prior year) 0.0067 (0.0206) $174,148$ Share indigenous speakers 0.0021 (0.0087) $173,813$ Average drug-related homicides (prior year) 0.0067 (0.0206) $174,148$ Share indigenous schooling -0.0025 (0.0028) $173,813$ Average drug-related homicides (prior year) 0.0067 (0.0190) $174,148$ Share indigenous schooling -0.0026 (0.0026) $173,813$ Average drug-related homicides (prior year) 0.0052 (0.0309) $130,792$ Share with o schooling -0.0014 (0.0026) $173,682$ Reports of inter-DTO violence before election 0.0145 (0.191) $130,792$ Share complete primary school 0.0004 (0.0026) $173,682$ Reports of gas sizures before election 0.1254 (0.771)	Homicides / months before election	-2.0593	(2.2104)	174,148	Share 3+ bedrooms	-0.0058	(0.0079)	173,682
Homicides S months before election $1.49/5$ (1.723) $174,148$ Share working age 0.0001 (0.0009) $173,893$ Homicides M nonths before election -0.7296 (2.1827) $174,148$ Share Catholic -0.0049 (0.0129) $173,893$ Average homicides (prior year) -1.0166 (2.2786) $174,148$ Share Catholic -0.0006 (0.0048) $173,693$ Average homicides (prior year) -1.8268 (2.2634) $174,148$ Share Catholic -0.0017 (0.0894) $173,693$ Top homicide decile (prior year) 0.0067 (0.0206) $174,148$ Share indigenous speakers 0.0024 (0.0034) $173,813$ Top homicide decile (prior year) 0.0067 (0.0206) $174,148$ Female years of schooling -0.0032 (0.0095) $173,813$ Average drug-related homicides (prior year) 0.0067 (0.0109) $174,148$ Share ilterate 0.0005 (0.0028) $173,862$ Reports of inter-DTO violence before election -0.4525 (0.8225) Share incomplete primary school -0.0014 (0.0026) $173,862$ Reports of inter-DTO violence before election -0.4525 (0.8225) Share incomplete primary school -0.0014 (0.0026) $173,862$ Reports of arg seizures before election -0.2454 (0.774) $130,792$ Share incomplete primary school -0.0017 (0.0077) $173,682$ Reports of arg seizures before election -0.2454 (0.2641) $130,792$ Share incomplete prim	Homicides 6 months before election	-1.9516	(2.4000)	174,148	Share female	-0.0004	(0.0008)	173,693
Homicides 4 months before election -0.296 $(2,1827)$ $174,148$ Share Children per woman -0.0018 (0.0182) $173,813$ Average homicides (prior 3 years) -0.7374 (1.4691) $174,148$ Share chono ut of state -0.0006 (0.0048) $173,693$ Average homicides (prior year) -1.8268 (2.2634) $174,148$ Share cholic -0.0067 (0.0034) $173,693$ Top homicide decile (prior year) 0.0067 (0.0206) $174,148$ Share sof schooling 0.0087 (0.0033) $173,813$ Presence of DTO 0.0067 (0.0109) $174,148$ Share illierate 0.0005 (0.0028) $173,862$ Average drug-related homicides (prior year) 0.0067 (0.0109) $174,148$ Share illierate 0.0005 (0.0028) $173,862$ Reports of inter-DTO violence before election -0.4525 (0.8225) $130,792$ Share incomplete primary school 0.0004 (0.0026) $173,862$ Reports of ing seizures before election 0.1254 (0.7811) $130,792$ Share complete secondary school 0.0001 (0.0027) $173,682$ Reports of gan seizures before election -0.2454 (0.7811) $130,792$ Share incomplete secondary school 0.0001 (0.0026) $173,682$ Reports of gan seizures before election -0.2454 (0.2411) $130,792$ Share complet secondary school 0.0011 (0.0077) $173,682$ Reports of gan seizures before election -0.2424 (0.2411) 130	Homicides 5 months before election	-1.49/5	(1.7223)	174,148	Share working age	0.0007	(0.0019)	173,693
Homicides 3 months before election-1.0166 $(2.2.786)$ $174,148$ Share born out of state-0.0049 (0.0129) $173,695$ Average homicides (prior year)-1.8268 (2.2634) $174,148$ Share Cabholic-0.0006 (0.0044) $173,692$ Top homicide quartile (prior year)0.0067 (0.0330) $174,148$ Share Cabholic-0.0167 (0.0084) $173,682$ Top homicide decile (prior year)0.0067 (0.0206) $174,148$ Fraale years of schooling0.0087 (0.0093) $173,813$ Average drug-related homicides (prior year)0.0067 (0.0109) $135,005$ Male years of schooling-0.0012 (0.0032) $173,813$ Average drug-related homicides (prior year)0.0067 (0.0109) $174,148$ Share incomplete primary school0.0004 (0.0026) $173,682$ Reports of inter-DTO violence before election0.01254 (0.7811) $130,792$ Share incomplete primary school -0.0014 (0.0026) $173,682$ Reports of arg seizures before election0.0257 (0.2641) $130,792$ Share incomplete scondary school -0.0014 (0.0026) $173,682$ Reports of arg seizures before election 0.254 (0.2641) $130,792$ Share incomplete scondary school 0.0006 0.0004 $173,682$ Reports of gun seizures before election 0.2245 (0.2641) $130,792$ Share complete scondary school 0.0011 0.0027 $173,682$ Reports of gun seizures before election 0.2784 $(0.$	Homicides 4 months before election	-0.7296	(2.1827)	1/4,148	Children per woman	-0.0018	(0.0182)	1/3,813
Average homicides (prior 3 years)-0.734(1.4991)174,148Share Catholic-0.0006(0.0048)173,682Average homicide quartile (prior year)0.0067(0.0330)174,148Years of schooling-0.0167(0.0034)173,813Top homicide quartile (prior year)0.0067(0.0206)174,148Years of schooling-0.0057(0.0330)174,813Presence of DTO0.0052(0.0206)174,148Share vars of schooling-0.0052(0.0035)173,813Average drug-related homicides (prior year)0.0067(0.0109)174,148Share illierate0.0005(0.0028)173,682Reports of inter-DTO violence before election-0.4525(0.8225)130,792Share complete primary school0.0004(0.0026)173,682Reports of violent enforcement before election0.1254(0.7871)130,792Share complete primary school0.0004(0.0026)173,682Reports of gun seizures before election0.6278(1.0910)130,792Share complete primary school0.0011(0.0073)173,682Reports of gun seizures before election0.4325(0.2641)130,792Share complete primary school0.0011(0.0021)173,682Reports of gun seizures before election0.4322(0.4582)130,792Share complete secondary school0.0011(0.0021)173,682Gun-related homicides before election0.4932(0.4582)130,792Share arcmplete arcmate secondary school0.0011(0.0021)173,682<	Homicides 3 months before election	-1.0166	(2.2786)	174,148	Share born out of state	-0.0049	(0.0129)	173,693
Average nomicodes (prior year)-1.8268 (2.2054) (1.4) (1.4) (1.5) (0.024) (0.0026) $(1.73,813)$ Average drug-related homicides (prior year) 0.0067 (0.0125) (0.8225) $(1.30,792)$ Share with no schooling -0.0014 (0.0026) $(1.73,682)$ Reports of violent enforcement before election 0.0254 (0.017) $(1.30,792)$ Share incomplete primary school 0.0011 (0.0077) $(1.73,682)$ Reports of un seizures before election 0.0224 (0.274) (0.274) (0.274) (0.0177) $(1.73,682)$ Reports of un seizures before election 0.0432 (0.2824) $(1.30,792)$ Share incomplete secondary school 0.0011 (0.0077) $(1.73,682)$ <td>Average homicides (prior 3 years)</td> <td>-0./3/4</td> <td>(1.4691)</td> <td>174,148</td> <td>Share Catholic</td> <td>-0.0006</td> <td>(0.0048)</td> <td>173,693</td>	Average homicides (prior 3 years)	-0./3/4	(1.4691)	174,148	Share Catholic	-0.0006	(0.0048)	173,693
Iop homicide quartite (prior year) 0.0067 (0.0330) $174,148$ Years of schooling -0.0167 (0.0089) $173,813$ Presence of DTO 0.0057 (0.0020) $174,148$ Fernale years of schooling 0.0032 (0.0965) $173,813$ Average drug-related homicides (prior year) 0.0067 (0.0109) $174,148$ Share viln on schooling -0.0032 (0.0026) $173,862$ Reports of inter-DTO violence before election 0.0455 (0.8225) $130,792$ Share viln on schooling -0.0004 (0.0026) $173,682$ Reports of violent enforcement before election 0.1254 (0.7871) $130,792$ Share incomplete primary school -0.0017 (0.0057) $173,682$ Reports of arests before election 0.6278 (1.0910) $130,792$ Share complete primary school -0.0017 (0.0077) $173,682$ Reports of arest surges before election 0.6278 (1.0910) $130,792$ Share complete scondary school 0.0006 (0.0077) $173,682$ Reports of surge sizures before election 0.4932 (0.4582) $130,792$ Share complete scondary school 0.0010 (0.0027) $173,682$ Reports of gun seizures before election 0.4932 (0.4582) $130,792$ Share normplete scondary school 0.0011 (0.0077) $173,682$ Gun-related homicides before election 0.4932 (0.4582) $130,792$ Share tomplet scondary school 0.0022 (0.0032) $173,693$ Gun-related homicides before elec	Average homicides (prior year)	-1.8268	(2.2634)	174,148	Share indigenous speakers	0.0024	(0.0034)	173,682
Iop nomicide decile (prior year)0.0067(0.0206)173, 14Freemace years or schooling0.007(0.0905)173, 813Average drug-related homicides (prior year)0.0067(0.0109)174, 148Share sither soft schooling-0.0023(0.0026)173, 813Average drug-related homicides (prior year)0.0067(0.0109)174, 148Share with no schooling-0.0004(0.0026)173, 682Reports of violent enforcement before election0.0345(0.1019)130, 792Share incomplete primary school0.0004(0.0026)173, 682Reports of violent enforcement before election0.1254(0.7871)130, 792Share incomplete secondary school0.0017(0.0077)173, 682Reports of arrests before election0.6278(1.0910)130, 792Share incomplete secondary school0.0011(0.0077)173, 682Reports of gun seizures before election0.4332(0.4582)130, 792Share conomically active-0.0020(0.0084)173, 682Gun-related homicides before election0.4932(0.4582)130, 792Share conomically active-0.0020(0.0031)173, 682Gun-related homicides before election0.019(0.0034)(5.4080)174, 148Share without health care-0.0022(0.0022)173, 693Gun-related homicides before election0.1278(0.0162)174, 148Share runing water-0.0055(0.0025)173, 682Drowning-related homicides before election0.1278(0.032)(0.034)17	Top nomicide quartile (prior year)	0.0067	(0.0330)	174,148	Years of schooling	-0.016/	(0.0894)	172,012
Presence of D1O 0.0052 (0.0509) $153,005$ Male years or schooling -0.0022 (0.0022) $173,682$ Average drug-related homicides (prior year) 0.0067 (0.0109) $174,148$ Share illicrate 0.0004 (0.0026) $173,682$ Reports of inter-DTO violence before election 0.0345 (0.1091) $130,792$ Share incomplete primary school 0.0004 (0.0026) $173,682$ Reports of arcg seizures before election 0.6278 (1.0910) $130,792$ Share incomplete primary school 0.0001 (0.0073) $173,682$ Reports of arset seizures before election 0.6278 (1.0910) $130,792$ Share complete secondary school 0.0001 (0.0073) $173,682$ Reports of gun seizures before election -0.3245 (0.2641) $130,792$ Share complete secondary school 0.0011 (0.0077) $173,682$ Reports of gun seizures before election -0.3245 (0.2641) $130,792$ Share complete secondary school 0.0011 (0.0073) $173,693$ Gun-related homicides before election -0.1984 (5.4080) $174,148$ Share sither control and a control	Top nomicide decile (prior year)	0.0067	(0.0206)	1/4,148	Female years of schooling	0.0087	(0.0903)	172,813
Average ang-related homicides (prior year) 0.0067 (0.0109) $174,148$ Share initerate 0.0005 (0.0028) $173,682$ Reports of inter-DTO violence before election 0.4525 (0.8225) $130,792$ Share with no schooling -0.0004 (0.0026) $173,682$ Reports of arrests before election 0.1254 (0.7871) $130,792$ Share complete primary school -0.0017 (0.0007) $173,682$ Reports of arrests before election 0.6278 (1.0910) $130,792$ Share complete secondary school 0.0001 (0.0077) $173,682$ Reports of asset seizures before election -0.3245 (0.2641) $130,792$ Share complete secondary school 0.0011 (0.0077) $173,682$ Reports of gun seizures before election -0.3245 (0.2641) $130,792$ Share complete secondary school 0.0011 (0.0077) $173,682$ Gun-related homicides before election -0.1984 (0.172) $174,148$ Share without health care -0.0020 (0.0025) $173,693$ Gun-related homicides before election -0.1984 (0.0172) $174,148$ Share running water -0.0025 (0.0065) $173,682$ Prowning-related homicides before election 0.1094^{**} (0.172) $174,148$ Share running water -0.0025 (0.0005) $173,682$ Smoke/fire-related homicides before election 0.1096^{***} (0.217) $174,148$ Share running water -0.0055 (0.0065) $173,682$ Smoke/fire-related homic	Presence of DTO	0.0052	(0.0309)	135,005	Male years of schooling	-0.0032	(0.0965)	173,813
Reports of inter-D1O Violence before election -0.325 $(0.8225$ $130,792$ Share with no schooling -0.0004 (0.0026) $173,682$ Reports of violent enforcement before election 0.1254 (0.7871) $130,792$ Share incomplete primary school 0.0004 (0.0026) $173,682$ Reports of arrests before election 0.6278 (1.091) $130,792$ Share incomplete secondary school 0.0006 (0.0073) $173,682$ Reports of gun seizures before election 0.6278 (0.2641) $130,792$ Share incomplete secondary school 0.0011 (0.0077) $173,682$ Reports of gun seizures before election 0.4325 (0.4582) $130,792$ Share complete secondary school 0.0011 (0.0077) $173,682$ High-risk electoral precinct 0.0019 (0.034) $52,403$ Share economically active -0.0030 (0.0022) $173,693$ Chemical substance-related homicides before election $-0.1284*$ (0.172) $174,148$ Share workers health care -0.0022 (0.0025) $173,682$ Drowning-related homicides before election 0.1278 (0.172) $174,148$ Share running water -0.00202 (0.0025) $173,682$ Cutting object-related homicides before election $0.1284*$ (0.172) $174,148$ Share running water -0.0046 (0.0021) $173,682$ Explosives-related homicides before election $0.1278*$ (0.076) $174,148$ Share running machine -0.0046 (0.0010) $173,682$ <td< td=""><td>Average drug-related nomicides (prior year)</td><td>0.0067</td><td>(0.0109)</td><td>1/4,148</td><td>Share initerate</td><td>0.0005</td><td>(0.0028)</td><td>173,082</td></td<>	Average drug-related nomicides (prior year)	0.0067	(0.0109)	1/4,148	Share initerate	0.0005	(0.0028)	173,082
Reports of volent enforcement before election 0.0343 (0.07871) $130,792$ Share incomplete primary school 0.0006 (0.0027) $173,682$ Reports of arrests before election 0.6278 (1.0910) $130,792$ Share incomplete secondary school 0.0006 (0.0073) $173,682$ Reports of asset seizures before election 0.6278 (1.0910) $130,792$ Share complete secondary school 0.0011 (0.0077) $173,682$ Reports of gun seizures before election 0.4932 (0.4822) $130,792$ Share complete secondary school 0.0011 (0.0077) $173,682$ Gun-related homicides before election 0.019 (0.0034) $52,403$ Share enomically active -0.0020 (0.0082) $173,693$ Chemical substance-related homicides before election -0.1284 (0.0172) $174,148$ Share state workers health care 0.0027 (0.0025) $173,682$ Drowning-related homicides before election 0.1278 (0.1085) $174,148$ Share training water -0.0055 (0.0065) $173,682$ Drowning-related homicides before election 0.1278 (0.217) $174,148$ Share adming machine -0.0024 (0.0005) $173,682$ Smoke/fire-related homicides before election 0.1278 (0.217) $174,148$ Share radio -0.0045 (0.0064) $173,682$ Smoke/fire-related homicides before election 0.0320 (0.0314) $174,148$ Share radio -0.0056 (0.0064) $173,682$ Blunt object-r	Reports of inter-DTO violence before election	-0.4323	(0.8223)	120,792	Share with no schooling	-0.0004	(0.0026)	172,682
Reports of artests before election 0.1254 (0.7817) $150,92$ Share complete scondary school -0.0017 (0.0073) $173,682$ Reports of drug seizures before election 0.6278 (1.0910) $130,792$ Share complete secondary school 0.0011 (0.0073) $173,682$ Reports of gun seizures before election 0.4932 (0.4582) $130,792$ Share complete secondary school 0.0010 (0.0073) $173,682$ High-risk electoral precinct 0.0019 (0.0034) $52,403$ Share complete secondary school 0.0020 (0.0084) $173,693$ Chemical substance-related homicides before election $-0.0284*$ (0.0172) $174,148$ Share virthout health care -0.0022 (0.0059) $173,693$ Hanging-related homicides before election 0.1278 (0.1965) $174,148$ Share running water -0.0055 (0.0065) $173,682$ Drowning-related homicides before election 0.0320 (0.3385) $174,148$ Share drainage -0.0046 (0.0012) $173,682$ Explosives-related homicides before election 0.0320 (0.334) $174,148$ Share running water -0.0089 (0.0100) $173,682$ Cutting object-related homicides before election 0.6487^{***} (0.2217) $174,148$ Share radio -0.0046 (0.0075) $173,682$ Delayed homicides before election 0.6487^{***} (0.2217) $174,148$ Share cardine telephone -0.0022 (0.0010) $173,682$ Delayed homicide registra	Reports of violent enforcement before election	0.0345	(0.1091)	130,792	Share incomplete primary school	0.0004	(0.0020)	172,682
Reports of arug seizures before election 0.0278 (1.910) $130,792$ Share incomplete secondary school 0.0006 (0.0073) $173,682$ Reports of gun seizures before election 0.2451 (0.2641) $130,792$ Share complete secondary school 0.0011 (0.0077) $173,682$ High-risk electoral precinct 0.0019 (0.034) $52,403$ Share conomically active -0.0030 (0.0022) $173,693$ Gun-related homicides before election -0.1984 (5.4080) $174,148$ Share state workers health care -0.0022 (0.0027) $173,693$ Chemical substance-related homicides before election $-0.0284*$ (0.1072) $174,148$ Share state workers health care -0.0022 (0.0022) $173,693$ Hanging-related homicides before election 0.1278 (0.1965) $174,148$ Share running water -0.0055 (0.00051) $173,682$ Drowning-related homicides before election 0.1096^{***} (0.0385) $174,148$ Share running water -0.0055 (0.00051) $173,682$ Smoke/fire-related homicides before election 0.0320 (0.0314) $174,148$ Share arainage -0.0046 (0.0051) $173,682$ Blunt object-related homicides before election $0.6487***$ (0.2217) $174,148$ Share radio -0.0045 (0.0064) $173,682$ Blunt object-related homicides before election $0.552*$ (0.0796) $174,148$ Share call phone -0.0022 (0.0104) $173,682$ Male homicide vi	Reports of arrests before election	0.1234	(0.7871)	120,792	Share complete primary school	-0.001/	(0.0057)	172,682
Reports of asset selzures before election -0.2243 (0.2641) $130,792$ Share complete secondary school 0.0011 (0.0077) $173,682$ Reports of gun seizures before election 0.4932 (0.4582) $130,792$ Share higher education 0.0020 (0.0084) $173,682$ High-risk electoral precinct 0.0019 (0.0034) $52,403$ Share economically active -0.0030 (0.0022) $173,693$ Gun-related homicides before election -0.1984 (5.4080) $174,148$ Share state workers health care -0.0022 (0.0021) $173,693$ Hanging-related homicides before election 0.1278 (0.0172) $174,148$ Share running water -0.0055 (0.0065) $173,682$ Drowning-related homicides before election 0.1096^{***} (0.0314) $174,148$ Share drainage -0.0046 (0.0051) $173,682$ Explosives-related homicides before election 0.0050 (0.0049) $174,148$ Share running water -0.0055 (0.0065) $173,682$ Smoke/fire-related homicides before election 0.0320 (0.0314) $174,148$ Share radio -0.0046 (0.0051) $173,682$ Blunt object-related homicides before election 0.5487^{***} (0.2217) $174,148$ Share radio -0.0045 (0.0064) $173,682$ Blunt object-related homicides before election $0.552*$ (0.2217) $174,148$ Share cell phone -0.0020 (0.0016) $173,682$ Male homicide victims before election 1.68	Reports of drug seizures before election	0.0278	(1.0910)	120,792	Share meetinglete secondary school	0.0000	(0.0073)	172,602
Reports of gun selzines before election 0.4952 (0.4852) $153, 192$ Share nginer education 0.0020 (0.0034) $173, 682$ Gun-related homicides before election -0.1984 (5.4080) $174, 148$ Share economically active -0.0022 (0.0032) (1.0032) Gun-related homicides before election -0.1984 (5.4080) $174, 148$ Share economically active -0.0022 (0.0022) $173, 693$ Hanging-related homicides before election 0.1278 (0.172) $174, 148$ Share state workers health care 0.0027 (0.0022) $173, 693$ Drowning-related homicides before election 0.1096^{***} (0.0385) $174, 148$ Share running water -0.0046 (0.0065) $173, 682$ Drowning-related homicides before election 0.1096^{***} (0.385) $174, 148$ Share washing machine -0.0046 (0.0051) $173, 682$ Smoke/fire-related homicides before election 0.0320 (0.0314) $174, 148$ Share radio -0.0046 (0.0064) $173, 682$ Cutting object-related homicides before election 0.6487^{***} (0.2217) $174, 148$ Share radio -0.0045 (0.0007) $173, 682$ Blun object-related homicides before election 0.552^{**} (0.5707) $174, 148$ Share cell phone -0.0022 (0.0017) $173, 682$ Male homicide registrations before election 1.8014 (5.7970) $174, 148$ Share cell phone -0.0023 (0.0100) $173, 682$ Male homicide	Reports of asset seizures before election	-0.3243	(0.2041)	120,792	Share complete secondary school	0.0011	(0.0077)	172,682
Inights decloral predict0.00190.0034 $32,403$ Share economically active-0.0030(0.0032)173,693Gun-related homicides before election-0.1984(5.4080)174,148Share without health care-0.0022(0.0052)173,693Chemical substance-related homicides before election0.1278(0.1965)174,148Share without health care-0.0055(0.0065)173,682Drowning-related homicides before election0.1096***(0.1905)174,148Share running water-0.0046(0.0051)173,682Explosives-related homicides before election0.0050(0.0049)174,148Share washing machine-0.0046(0.0051)173,682Smoke/fire-related homicides before election0.0320(0.0314)174,148Share vashing machine-0.0045(0.0064)173,682Cutting object-related homicides before election0.6487***(0.2217)174,148Share radio-0.0045(0.0064)173,682Delayed homicide registrations before election0.1552*(0.0796)174,148Share care fridge-0.0023(0.0100)173,682Male homicide victims before election1.6872(6.3566)149,857Share car or truck-0.0039(0.0100)173,682Male homicide victims before election0.3211(0.2870)145,965Share computer-0.0035(0.009)173,682Victims with no schooling before election0.3211(0.2870)145,965Share computer-0.0047(0.0098)173,682<	High risk alastaral provinct	0.4932	(0.4382)	52 402	Share accommissily active	0.0020	(0.0084) (0.0032)	172,602
Chamicates before election -0.1964 (0.003) $174,148$ Share windou neurine and care -0.0022 (0.0039) $173,093$ Chemical substance-related homicides before election -0.0284^* (0.0172) $174,148$ Share state workers health care 0.0027 (0.0022) $173,693$ Imaging-related homicides before election 0.1278 (0.1965) $174,148$ Share state workers health care -0.0027 (0.0022) $173,693$ Drowning-related homicides before election 0.1278 (0.1965) $174,148$ Share running water -0.0046 (0.0051) $173,682$ Explosives-related homicides before election 0.0020 (0.0049) $174,148$ Share avashing machine -0.0046 (0.0010) $173,682$ Smoke/fire-related homicides before election $0.6487***$ (0.2217) $174,148$ Share radio -0.0045 (0.0064) $173,682$ Blunt object-related homicides before election $0.6487***$ (0.2217) $174,148$ Share radio -0.0045 (0.0064) $173,682$ Delayed homicide registrations before election 1.6872 (6.3566) $149,857$ Share carl phone -0.0023 (0.0100) $173,682$ Male homicide victims before election 1.6872 (6.3566) $149,857$ Share carl or truck -0.0039 (0.0039) $173,682$ Victims with no schooling before election 0.3211 (0.2870) $148,965$ Share car or truck -0.0047 0.0049 $173,682$ Victims with ncomplete primary school	Gun related homigides before election	0.0019	(0.0034)	32,405	Share economically active	-0.0030	(0.0052)	172,693
Chemical substance-related homicides before election -0.024^{+} -0.0172^{-} $174,148^{+}$ Share state state workers health care -0.0057^{-} $(0.0022)^{-}$ $173,682^{-}$ Hanging-related homicides before election 0.1278^{+} $(0.0965)^{-}$ $174,148^{+}$ Share running water -0.0055^{-} $(0.0065)^{-}$ $173,682^{-}$ Drowning-related homicides before election 0.0050^{-} $(0.0385)^{-}$ $174,148^{-}$ Share vashing machine -0.0046^{-} $(0.0065)^{-}$ $173,682^{-}$ Smoke/fire-related homicides before election 0.0320^{-} $(0.0314)^{-}$ $174,148^{-}$ Share landline telephone -0.0122^{-} $(0.0143)^{-}$ $173,682^{-}$ Cutting object-related homicides before election 0.6487^{***} $(0.0796)^{-}$ $174,148^{-}$ Share radio -0.0045^{-} $(0.0064)^{-}$ $173,682^{-}$ Blunt object-related homicides before election 0.552^{*} $(0.0796)^{-}$ $174,148^{-}$ Share radio -0.0045^{-} $(0.0064)^{-}$ $173,682^{-}$ Delayed homicide victims before election 1.552^{*} $(0.0796)^{-}$ $174,148^{-}$ Share call phone -0.0023^{-} $(0.0100)^{-}$ $173,682^{-}$ Male homicide victims before election 1.6872^{-} $(6.3566)^{-}$ $149,857^{-}$ Share car or truck -0.0023^{-} $(0.0031)^{-}$ $173,682^{-}$ Victims with no schooling before election 0.3211^{-} $(0.2870)^{-}$ $145,965^{-}$ Share car or truck -0.0047^{-} 0.0098^{-} $173,682^{-}$ Victims with ncomplete pri	Chamical substance related homigides before election	-0.1964	(0.0172)	174,140	Share state workers health care	-0.0022	(0.0039)	172,602
Hanging-related homicides before election 0.1278 (0.965) $174,148$ Share running water -0.0053 (0.0053) $173,682$ Drowning-related homicides before election 0.1096^{***} (0.0385) $174,148$ Share drainage -0.0046 (0.0051) $173,682$ Explosives-related homicides before election 0.0050 (0.0049) $174,148$ Share washing machine -0.0089 (0.0100) $173,682$ Smoke/fire-related homicides before election 0.6487^{***} (0.2217) $174,148$ Share radio -0.0045 (0.0064) $173,682$ Cutting object-related homicides before election 0.6487^{***} (0.2217) $174,148$ Share radio -0.0045 (0.0064) $173,682$ Blunt object-related homicides before election 0.5152^{**} (0.2217) $174,148$ Share radio -0.0045 (0.0064) $173,682$ Delayed homicide registrations before election 0.1552^{**} (0.2770) $174,148$ Share cell phone -0.0023 (0.0100) $173,682$ Male homicide victims before election 1.6872 (6.3566) $149,857$ Share cell phone -0.0023 (0.0031) $173,682$ Age of homicide victims before election 0.3211 (0.2870) $148,965$ Share computer -0.0035 (0.0099) $173,682$ Victims with no schooling before election 0.3211 (0.2870) $145,965$ Share computer -0.0047 0.0098 $173,682$ Victims with nomplete primary schooling before election 0.88	Una sing related herminidae hefere election	-0.0284	(0.0172)	174,140	Share state workers heatin care	0.0027	(0.0022)	172,693
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	Drowning related homicides before election	0.1278	(0.1903)	174,148	Share running water	-0.0033	(0.0003)	172 682
Explosives-related homicides before election 0.0030 (0.0049) $174,148$ Share washing machine -0.0045 (0.0160) $173,082$ Smoke/fire-related homicides before election 0.0320 (0.0314) $174,148$ Share landline telephone -0.0122 (0.0143) $173,682$ Cutting object-related homicides before election $0.6487***$ (0.2217) $174,148$ Share radio -0.0045 (0.0064) $173,682$ Blunt object-related homicides before election $0.1552*$ (0.0796) $174,148$ Share radio -0.0050 (0.0075) $173,682$ Delayed homicide registrations before election 1.8014 (5.7970) $174,148$ Share cell phone -0.0023 (0.0100) $173,682$ Male homicide victims before election 1.6872 (6.3566) $149,857$ Share car or truck -0.0021 (0.0031) $173,682$ Victims with no schooling before election $1.5006***$ (0.2870) $145,965$ Share car or truck -0.0057 (0.0099) $173,682$ Victims with no schooling before election 1.3313 (7.0974) $145,965$ Share internet -0.0047 (0.0098) $173,682$ Victims with nomplete primary schooling before election 0.8803 (6.6975) $145,965$ Share internet -0.0047 0.0098 $173,682$	Explosives related homicides before election	0.1090	(0.0383)	174,140	Share washing mashing	-0.0040	(0.0031)	172 682
Sinderine-related homicides before election 0.6487^{***} (0.2217) $174,148$ Share radio -0.0122 (0.0043) $173,682$ Cutting object-related homicides before election 0.6487^{***} (0.2217) $174,148$ Share radio -0.0045 (0.0044) $173,682$ Blunt object-related homicides before election 0.1552^{**} (0.0796) $174,148$ Share radio -0.0045 (0.0044) $173,682$ Delayed homicide registrations before election 1.8014 (5.7970) $174,148$ Share cell phone -0.0023 (0.0100) $173,682$ Male homicide victims before election 1.6872 (6.3566) $149,857$ Share cell phone -0.0021 (0.0031) $173,682$ Age of homicide victims before election 1.5006^{***} (0.5803) $148,789$ Share car or truck -0.0039 (0.0099) $173,682$ Victims with no schooling before election 1.3313 (7.0974) $145,965$ Share on puter -0.0047 0.0098 $173,682$ Victims with nomplete primary schooling before election 0.8803 (6.6975) $145,965$ Share internet -0.0047 0.0098 $173,682$	Smales/fire related homicides before election	0.0030	(0.0049) (0.0214)	174,140	Share lendling talanhone	-0.0089	(0.0100)	172 682
Cutting object-related homicides before election 0.067^{-1} 0.067^{-1} 0.067^{-1} 0.0067^{-1} $173,082^{-1}$ Blunt object-related homicides before election 0.1552^{*} (0.0796) $174,148^{-1}$ Share fridge -0.0050^{-1} (0.0075) $173,682^{-1}$ Delayed homicide registrations before election 1.8014 (5.7970) $174,148^{-1}$ Share cell phone -0.0023^{-1} (0.0099) $173,682^{-1}$ Male homicide victims before election 1.6872 (6.3566) $149,857^{-1}$ Share cell phone -0.0021^{-1} (0.0099) $173,682^{-1}$ Age of homicide victims before election 1.5006^{***} (0.5803) $148,789^{-1}$ Share car or truck -0.0039^{-1} $(0.0099)^{-1}$ $173,682^{-1}$ Victims with no schooling before election 0.3211^{-1} $(0.2870)^{-1}$ $145,965^{-1}$ Share internet -0.0047^{-1} $(0.0098)^{-1}$ $173,682^{-1}$ Victims with ncomplete primary schooling before election 0.8803^{-1} $(6.6975^{-1})^{-1}$ $145,965^{-1}$ Share internet -0.0047^{-1} 0.0098^{-1} $173,682^{-1}$ Victims with complete primary schooling before election 0.8803^{-1} $(6.6975^{-1})^{-1}$ $145,965^{-1}$ -0.0047^{-1} 0.0098^{-1} $173,682^{-1}$	Cutting object related homicides before election	0.0320	(0.0314) (0.2217)	174,140	Share radio	0.0045	(0.0143)	173,082
Brain tolect-retaid infinition of the formation of the fo	Plunt abject related homigides before election	0.1552*	(0.2217)	174,140	Share fridge	-0.0045	(0.0004)	172 692
Declayed infinite registrations before election 1.6014 (5.7570) 174,146 Shate cert priore -0.0025 (0.0100) 173,082 Male homicide victims before election 1.6872 (6.3566) 149,857 Share television -0.0021 (0.0031) 173,682 Age of homicide victims before election 1.5006*** (0.5803) 148,789 Share car or truck -0.0039 (0.0099) 173,682 Victims with no schooling before election 0.3211 (0.2870) 145,965 Share computer -0.0055 (0.0109) 173,682 Victims with incomplete primary schooling before election 1.3313 (7.0974) 145,965 Share internet -0.0047 (0.0098) 173,682 Victims with complete primary schooling before election 0.8803 (6.6975) 145,965 Share internet -0.0047 (0.0098) 173,682	Data object-related nonnerices before election	1.8014	(0.0790)	174,148	Share call phone	-0.0030	(0.0073)	173,082
Age of homicale victims before election 1.0072 (0.300) 149,67 Share television -0.0021 (0.0031) 173,682 Age of homicale victims before election 1.5066*** (0.8803) 148,789 Share car or truck -0.0039 (0.0099) 173,682 Victims with no schooling before election 0.3211 (0.2870) 145,965 Share computer -0.0055 (0.0109) 173,682 Victims with incomplete primary schooling before election 1.3313 (7.0974) 145,965 Share internet -0.0047 (0.0098) 173,682 Victims with complete primary schooling before election 0.8803 (6.6975) 145,965 Share internet -0.0047 (0.0098) 173,682	Male homicide victims before election	1.6014	(5.1910)	1/4,148	Share television	-0.0023	(0.0100) (0.0021)	173,082
Age of holine we within school ing before election 1.0000** (10.305) 145,765 Share can of nuck -0.0059 (10.099) 173,882 Victims with no schooling before election 0.3211 (0.2870) 145,965 Share computer -0.0055 (0.0109) 173,682 Victims with no schooling before election 1.3313 (7.0974) 145,965 Share internet -0.0047 (0.0098) 173,682 Victims with complete primary schooling before election 0.8803 (6.6975) 145,965 Share internet -0.0047 (0.0098) 173,682	Age of homicide victims before election	1.0072	(0.5500)	149,037	Share car or truck	-0.0021	(0.0031)	173,082
Victims with complete primary schooling before election 0.3211 (0.670) $145,965$ Share internet -0.0033 (0.0109) $173,682$ Victims with complete primary schooling before election 0.3813 (7.0974) $145,965$ Share internet -0.0047 (0.0098) $173,682$ Victims with complete primary schooling before election 0.8803 (6.6975) $145,965$ Share internet -0.0047 (0.0098) $173,682$	Victims with no schooling before election	0 3211	(0.3803)	140,709	Share computer	-0.0039	(0.0099) (0.0100)	173,002
Victims with complete primary schooling before election 1.5515 (1.0574) $145,965$ -0.0047 (0.0096) $175,082$	Victims with incomplete primary schooling before election	1 3313	(0.2870)	145,905	Share internet	-0.0033	(0.0109)	173,002
	Victims with complete primary schooling before election	0.8803	(6.6975)	145,965		5.00-17	(0.0070)	175,002

Notes: Specifications include municipality and year fixed effects, and are estimated using OLS. The time-invariant variables—area, media coverage, and the 2010 Census variables (occupants per dwelling-share internet)—exclude municipality fixed effects. Standard errors clustered by municipality in parentheses. * denotes p < 0.1, ** denotes p < 0.05, *** denotes p < 0.01.

	Municipal pu employees pe	blic security r 1000 voters
	(1)	(2)
Local election year	0.006	-0.038
	(0.069)	(0.082)
Observations	9,655	9,655
Outcome mean	4.23	4.23
Outcome range	[.01,201.86]	[.01,201.86]
Local election year mean	0.31	0.31
Municipality-specific time trends		\checkmark

Table A8: Effect of local elections on municipal public security employment

Notes: All specifications include municipality and survey-year fixed effects, and are estimated using OLS. Municipalities with no employees are excluded. Standard errors clustered by municipality are in parentheses. * denotes p < 0.1, ** denotes p < 0.05, *** denotes p < 0.01.

	Ave	erage hominonths aft	icides in the	e two on
	(1)	(2)	(3)	(4)
Incumbent party win	4.315 (9.473)			
Change in incumbent party vote share		-7.606 (6.251)		
PAN win			0.183 (6.357)	-19.195 (25.776)
PRI win			1.124 (4.920)	1.905 (19.214)
PRD win			-8.656 (10.742)	-13.411 (52.524)
Observations Outcome mean Election outcome mean	3314 18.90 -0.05	174148 18.91 0.56	3314 18.90	1558 27.84

Table A9: Correlation between election outcomes and post-election homicides

Notes: All specifications include municipality and year fixed effects, and are estimated using OLS. Standard errors clustered by municipality are in parentheses. * denotes p < 0.1, ** denotes p < 0.05, *** denotes p < 0.01.

Table A10: Precinct-level balance over 105	variables over the	e number of local	media stations
--	--------------------	-------------------	----------------

Quitcome	Effect	f homicide	shock	Outcome	Effect of	homicide	shock
Outonie	Coef	SE	Obs	Outcome	Coef	SE	Obs
Hamiaida abaala	0.0005	(0.0004)	22.460	Mala barriaida aiatina bafana alaatina	0.0020	(0.0000)	21.705
Homicide shock	-0.0005	(0.0004)	33,460	Male homicide victims before election	-0.0028	(0.0096)	31,705
Area (km ⁻)	-0.0018	(0.0039)	22,460	Age of nomicide victims before election	-0.1094	(0.3309)	21 455
Electorate Municipal electorate	-21.3210	(13.1923) (20.214)	33,400	Victims with no schooling before election	-0.0004	(0.0000)	31,433 21,455
Floaterate density	-45.855	(39.214)	33,423	Victims with incomplete primary schooling before algorithm	-0.0023	(0.0105)	31,433 21,455
Distance from controld to municipal head (log)	0.0020	(23.6496)	22,450	Victims with complete primary schooling before election	-0.0022	(0.0094) (0.0061)	21 455
Non local media	-0.0020	(0.0050) (0.1847)	33,400	Single victims before election	-0.0013	(0.0001)	31,455
Insumbent vote share (lag)	-0.0133	(0.1647)	33,400	In related hominidae before election	-0.0004	(0.0042)	20,220
Incumbent voie share (lag)	-0.0005	(0.0003)	27 161	Organs avamined after homicide before election	-0.0008	(0.0019)	29,230
Win margin (lag)	-0.0003	(0.0004) (0.0005)	27,101	Family incident homicides before election	-0.0008	(0.0093)	30,979
Will Inargin (lag) Municipality win margin (lag)	0.0001	(0.0003)	33,400	Urban homicide victims before election	0.0001	(0.0002) (0.0084)	27 040
FNDV (lag)	0.0001	(0.0001) (0.0024)	33,400	Non Maxican homicide victims before election	-0.0071	(0.0064)	21,949
Municipality ENDV (lag)	0.0023	(0.0024)	33,460	Neighbor average homicide shock	0.0000	(0.0002)	32,703
Turnout (lag)	0.0000	(0.0005)	22 479	Flasterate weighted neigh ave hom sheek	0.0003*	(0.0002)	22,222
Municipality turnout (lag)	0.0001	(0.0000)	32,470	Average non homicide deaths (prior 3 years)	0.0004	(0.0002)	33,400
PDI incumbent	-0.0000	(0.0000)	32,319	Average non-homicide deaths (prior year)	-0.0124	(0.0130) (0.0176)	33,400
PAN incumbent	0.0001	(0.0003)	33,460	Average child mortalities (prior 3 years)	-0.0109	(0.0170)	25 250
PRD incumbent	0.0004	(0.000+) (0.0002)	33,460	Average child mortalities (prior year)	0.0019	(0.0013)	25,250
Mayor's party aligned with President	0.0002	(0.0002)	33,460	Total municipal spending	0.2580	(0.0011) (0.3446)	21,220
Mayor's party aligned with Covernor	-0.0002	(0.0003)	33,460	Police per voter	-0.2389	(0.3440)	21,524
Mayor's party aligned with President and Governor	-0.0004	(0.0003)	33,460	Occupants per dwelling	0.0004	(0.0003)	33,460
Homicides 12 months before election	0.0000	(0.0002) (0.0070)	33,460	Occupants per room	-0.0033	(0.0021) (0.0014)	33,460
Homicides 11 months before election	0.0023	(0.0070)	33,460	Share with 2 bedrooms	0.0001	(0.0014)	33,460
Homicides 10 months before election	0.0021	(0.0102)	33,460	Share 3 L bedrooms	0.0001	(0.0000)	33,460
Homicides 9 months before election	0.0010	(0.0058)	33,460	Share female	0.0010	(0.0007)	33,460
Homicides 8 months before election	0.0010	(0.0000)	33,460	Share working age	-0.0002	(0.0002)	33,460
Homicides 7 months before election	-0.0010	(0.0033)	33,460	Children per woman	-0.0003	(0.0002)	33,460
Homicides 6 months before election	-0.0011	(0.0078)	33 460	Share born out of state	0.0003	(0.0017)	33 460
Homicides 5 months before election	-0.0025	(0.0090)	33 460	Share Catholic	0.0001	(0.0005)	33 460
Homicides 4 months before election	-0.0001	(0.0050)	33,460	Share indigenous speakers	0.0001	(0.0000)	33,460
Homicides 3 months before election	0.0006	(0.0005)	33 460	Years of schooling	0.0105	(0.0002) (0.0103)	33 460
Average homicides (prior 3 years)	0.0029	(0.0034)	33 460	Female years of schooling	0.0052	(0.0103)	33 460
Average homicides (prior 9 years)	-0.00027	(0.0043)	33 460	Male years of schooling	0.0148	(0.0101)	33 460
Top homicide quartile (prior year)	0.0000	(0.0000)	33 460	Share illiterate	-0.0002	(0.0103)	33 460
Top homicide decile (prior year)	0.0000	(0.0000)	33 460	Share with no schooling	-0.0002	(0.0001)	33 460
Presence of DTO	-0.0002	(0.0002)	24 916	Share incomplete primary school	0.0002	(0.0002)	33 460
Average drug-related homicides (prior year)	0.0001	(0.0001)	33,460	Share complete primary school	0.0004	(0.0002)	33,460
Reports of inter-DTO violence before election	-0.0002	(0.0028)	24.137	Share incomplete secondary school	0.0007	(0.0007)	33,460
Reports of violent enforcement before election	0.0004	(0.0005)	24.137	Share complete secondary school	0.0010	(0.0008)	33,460
Reports of arrests before election	0.0038	(0.0026)	24.137	Share higher education	0.0011	(0.0011)	33,460
Reports of drug seizures before election	0.0050	(0.0048)	24.137	Share economically active	0.0002	(0.0003)	33,460
Reports of asset seizures before election	0.0013	(0.0010)	24.137	Share without health care	-0.0002	(0.0004)	33,460
Reports of gun seizures before election	0.0005	(0.0009)	24.137	Share state workers health care	0.0003	(0.0003)	33,460
High-risk electoral precinct	0.0011	(0.0011)	7.921	Share running water	0.0013**	(0.0006)	33,460
Gun-related homicides before election	-0.0014	(0.0091)	33,460	Share drainage	0.0006**	(0.0003)	33,460
Chemical substance-related homicides before election	0.0000	(0.0001)	33,460	Share washing machine	0.0006	(0.0009)	33,460
Hanging-related homicides before election	0.0009	(0.0008)	33,460	Share landline telephone	0.0023*	(0.0012)	33,460
Drowning-related homicides before election	0.0000	(0.0001)	33,460	Share fridge	0.0003	(0.0005)	33,460
Explosives-related homicides before election	NA	NA	33,460	Share cell phone	0.0001	(0.0005)	33,460
Smoke/fire-related homicides before election	0.0000	(0.0002)	33,460	Share car or truck	0.0006	(0.0012)	33,460
Cutting object-related homicides before election	0.0000	(0.0008)	33,460	Share computer	0.0011	(0.0013)	33,460
Blunt object-related homicides before election	0.0003	(0.0003)	33,460	Share internet	0.0020*	(0.0012)	33,460
Delayed homicide registrations before election	0.0003	(0.0103)	33,460			. /	

Notes: All specifications include neighbor group and year fixed effects, and are estimated using OLS. All observations are weighted by the number of registered voters divided by he number of comparison units within each neighbor group. There is no variation in the number of explosive-related homicides across local media in this sample. Standard errors clustered by municipality in parentheses. * denotes p < 0.1, ** denotes p < 0.05, *** denotes p < 0.01.

	Incumbent party win	Change in incumbent party vote share
	(1)	(2)
Drug-related homicide shock	-0.068	-0.020
	(0.074)	(0.017)
Observations	473	38 038
Observations	775	50,050
Outcome mean	0.53	-0.05
Homicide shock mean	0.44	0.44

Table A11: Effect of pre-election drug-related homicide shocks on municipal incumbent electoral outcomes

Notes: All specifications are estimated using OLS, and include municipality and year fixed effects. All observations are weighted by the number of registered voters. Standard errors clustered by municipality are in parentheses. * denotes p < 0.1, ** denotes p < 0.05, *** denotes p < 0.01.

with the general homicide level, a meaningful test of increased homicides around elections requires a subtler design.⁷⁷ I use the difference between the average number of homicides in the two months prior to an election and the average number of homicides over the prior electoral cycle (i.e. preceding 34 months), divided by the average number of homicides over the prior electoral cycle. This captures differences in the magnitude of pre-election shocks across elections, and thus exploits variation in the intensity of these shocks across municipalities. The identifying assumption is that municipalities experiencing different homicide deviations from cycle trends before elections otherwise follow parallel trends in incumbent support.

Table A12 reports estimates from the analog of equation (4), and demonstrates that this deviation measure of homicide shocks non-linearly decreases the incumbent party's vote share. Consistent with the idea that sanctioning is diminishing for a sufficiently large shock, column (5) includes a squared version of the deviation measure and shows negative but diminishing effects. The linear version in column (1) is also clearly negative but is not quite statistically significant in panel A. Supporting the non-linearity of the effect, column (3) shows stronger effects of magnitude comparable to Table 2 for an indicator for proportional changes above the sample median. The addition of municipality-specific trends substantially increase standard errors but generally provides similar point estimates.

⁷⁷The results of such an approach are thus similar to the small long-run estimates reported in Table 3.

	(1)	(2)	(3)	(4)	(5)	(9)
Panel A: Outcome: Incumbent party win						
Proportional deviation from average monthly homicide	-0.0178	-0.0230			-0.0334**	-0.0443
Above median proportional deviation from average monthly homicide			-0.1015* (0.0557)	-0.0427 (0.0919)		
Proportional deviation from average monthly homicide squared					0.0017** (0.0008)	0.0038 (0.0046)
Observations	3,243	3,243	3,243	3,243	3,243	3,243
Outcome mean	0.55	0.55	0.55	0.55	0.55	0.55
Homicide measure mean	0.37	0.37	0.50	0.50	0.37	0.37
Homicide measure standard deviation	1.73	1.73	0.50	0.50	1.73	1.73
Panel B: Outcome: Change in incumbent party vote share						
Proportional deviation from average monthly homicide	-0.0026	-0.0049			-0.0041	-0.0094
Above median proportional deviation from average monthly homicide	(0700.0)	(+0000)	-0.0341*** (0.0094)	-0.0355*** (0.0127)	(6700.0)	(00000)
Proportional deviation from average monthly homicide squared					0.0002	0.0008
					(0.0001)	(0.0007)
Observations	173,625	173,625	173,625	173,625	173,625	173,625
Outcome mean	0.55	0.55	0.55	0.55	0.55	0.55
Homicide measure mean	0.37	0.37	0.50	0.50	0.37	0.37
Homicide measure standard deviation	1.73	1.73	0.50	0.50	1.73	1.73
Municipality-specific time trends		>		>		>

registered voters. Standard errors clustered by municipality are in parentheses. * denotes p < 0.1, ** denotes p < 0.05, *** denotes p < 0.01.

A.5.4 Differential updating across elections

Table A13 shows the regression estimates underlying Figure 6.

A.5.5 Learning from the experiences of neighboring municipalities

To identify how comparative performance information impacts voter behavior, I compute the average homicide shock across neighboring municipalities.⁷⁸ The exogeneity of homicide shocks within municipalities facilitates a causal interpretation of the interaction between homicide shocks within a voter's municipality and shocks to their neighbors. Consistent with voters benchmarking their incumbent's performance against that of their neighbors, column (1) shows that the probability that an incumbent party is re-elected increases (albeit not quite significantly so) with the average shock experienced by neighboring municipalities. However, the negative interaction between the homicide shock and the neighbor average implies that relative performance considerations do not apply when a voter's own municipality is afflicted by a shock.⁷⁹ Plausibly consistent with increased salience, mayoral punishment instead increases with the proportion of shocks experienced by neighbors. The change is vote share follows a similar pattern but is not statistically significant, while columns (2) and (4) report similar results when an electorate-weighted neighboring municipal shock measure is used instead.

A.5.6 Differential impacts of homicide shocks across parties

I also examine differential punishment across parties. Due to the PRI's more extensive clientelistic ties (e.g. Cornelius 1996; Fox 1994; Greene 2007; Magaloni 2006), PRI voters may be less susceptible to performance information, and thus less inclined to punish PRI incumbents for homicide shocks. Table A15 provides tentative evidence consistent with this expectation. Although the interaction terms are not statistically significant, column (1) shows that the PAN and PRD are punished relatively more than the PRI. Although punishment of PRI incumbents is not statistically, columns (2)-(4) register significant punishment of PAN and PRD mayors. This is likely because the PAN and especially PRD control more urban areas where the effects of homicide shocks are larger.

Finally, who do voters turn to to reduce local violence? If reducing violence is a major concern, and voters believe that Calderón's tough stance on drug-related crime may help their municipality (Dell 2015), even PAN mayors may benefit from a homicide shock. Column (5) of panel C finds

⁷⁸The results are robust to weighting neighboring municipalities by the size of their registered electorates.

⁷⁹Unreported results separating neighboring incumbents from the same and different parties, but show no meaningful differences.

	Change in incumbent party vote share (1)
Homicide shock	-0.041**
	(0.018)
Homicide shock at last election	-0.042*
	(0.025)
Homicide shock \times Homicide shock at last election	0.064**
	(0.030)
Incumbent won previous election	-0.047**
-	(0.019)
Homicide shock \times Incumbent won previous election	0.014
-	(0.029)
Homicide shock at last election \times Incumbent won previous election	0.036
	(0.031)
Homicide shock \times Homicide shock at last election	-0.071
\times Incumbent won previous election	(0.044)
Observations	112.434
Outcome range	[-0.90,0.89]
Outcome mean	-0.06
Outcome standard deviation	0.15
Homicide shock mean	0.43
Homicide shock at last election mean	0.40
Incumbent won previous election mean	0.56

Table A13: The effect of a homicide shock on the change in incumbent party vote share, by previous homicide shocks and incumbent identity

Notes: All specifications include municipality and year fixed effects, and are estimated using OLS. All observations are weighted by the number of registered voters. Standard errors clustered by municipality in parentheses. * denotes p < 0.1, ** denotes p < 0.05, *** denotes p < 0.01.

	Incumbent party win (1)	Incumbent party win (2)	Change in incumbent party vote share (3)	Change in incumbent party vote share (4)
Homicide shock	0.007	-0.012	-0.001	-0.005
	(0.091)	(0.090)	(0.018)	(0.017)
Neighbor average homicide shock	0.161		0.026	
	(0.106)		(0.020)	
Homicide shock \times Neighbor	-0.259*		-0.043	
average homicide shock	(0.152)		(0.029)	
Electorate-weighted neighbor		0.150		0.027
average homicide shock		(0.104)		(0.019)
Homicide shock \times Electorate-weighted		-0.225		-0.036
neighbor average homicide shock		(0.152)		(0.028)
Observations	3,172	3,307	170,451	174,095
Outcome mean	0.55	0.55	-0.05	-0.05
Homicide shock mean	0.42	0.42	0.42	0.42
Interaction mean	0.47	0.46	0.47	0.46

 Table A14: Effect of pre-election homicide shocks on municipal incumbent electoral outcomes, by neighbor average homicide shock

Notes: All specifications are estimated using OLS. All observations are weighted by the number of registered voters. Standard errors clustered by municipality are in parentheses. * denotes p < 0.1, ** denotes p < 0.05, *** denotes p < 0.01.

	Change in incumbent party vote share (1)	Change in PAN incumbent vote share (2)	Change in PRD incumbent vote share (3)	Change in PRI incumbent vote share (4)	Change in PAN incumbent vote share (5)
Homicide shock	-0.008	-0.037***	-0.047*	-0.009	0.028
	(0.013)	(0.012)	(0.025)	(0.013)	(0.022)
Homicide shock \times PAN	-0.016				
	(0.023)				
Homicide shock \times PRD	-0.044				
	(0.035)				
Homicide shock × Calderón					0.044
Presidency					(0.043)
Observations	165,467	58,272	20,645	86,550	174,148
Outcome mean	-0.05	-0.06	-0.09	-0.04	-0.05
Homicide shock mean	0.42	0.43	0.44	0.41	0.42

 Table A15: Effects of pre-election homicide shocks on municipal incumbent electoral outcomes, by incumbent party

Notes: All specifications are estimated using OLS. All observations are weighted by the number of registered voters. Standard errors clustered by municipality are in parentheses. * denotes p < 0.1, ** denotes p < 0.05, *** denotes p < 0.01.

suggestive evidence for this: although a homicide shock increases the PAN's vote share by 4.4 percentage points during Calderón's presidency, this interaction is not statistically significant.